# Finite dimensional representations of symplectic reflection algebras for wreath products

by

Silvia Montarani

Laurea in Matematica Università di Roma "La Sapienza", September 2002

Submitted to the Department of Mathematics in partial fulfillment of the requirements for the degree of

Doctor of Philosophy

at the

#### MASSACHUSETTS INSTITUTE OF TECHNOLOGY

June 2008

©Silvia Montarani, MMVIII. All rights reserved.

The author hereby grants MIT permission to reproduce and to distribute publicly paper and electronic copies of this thesis document in whole or in part in any medium now known or hereafter created.

Author	
	Department of Mathematics
	April 17, 2008
Certified by	
-	Pavel Etingof
	Professor of Mathematics
	Thesis Supervisor
Accepted by	
	David Jerison
Cha	irman, Department Committee on Graduate Students

### Finite dimensional representations of symplectic reflection algebras for wreath products

by

Silvia Montarani

Submitted to the Department of Mathematics on April 17, 2008, in partial fulfillment of the requirements for the degree of Doctor of Philosophy

#### Abstract

Symplectic reflection algebras are attached to any finite group G of automorphisms of a symplectic vector space V, and are a multi-parameter deformation of the smash product  $TV \sharp G$ , where TV is the tensor algebra. Their representations have been studied in connection with different subjects, such as symplectic quotient singularities, Hilbert scheme of points in the plane and combinatorics. Let  $\Gamma \subset SL(2,\mathbb{C})$ be a finite subgroup, and let  $S_n$  be the symmetric group on n letters. We study finite dimensional representations of the wreath product symplectic reflection algebra  $H_{1,k,c}(\Gamma_n)$  of rank n, attached to the wreath product group  $\Gamma_n = S_n \ltimes \Gamma^n$ , and to the parameters (k, c), where k is a complex number (occurring only for n > 1), and c a class function on the set of nontrivial elements of  $\Gamma$ . In particular, we construct, for the first time, families of irreducible finite dimensional modules when  $\Gamma$  is not cyclic, n > 1, and (k, c) vary in some linear subspace of the space of parameters. The method is deformation theoretic and uses properties of the Hochschild cohomology of  $H_{1,k,c}(\Gamma_n)$ , and a Morita equivalence, established by Crawley-Boevey and Holland, between the rank one algebra  $H_{1,c}(\Gamma)$  and the deformed preprojective algebra  $\Pi_{\lambda}(Q)$ , where Q is the extended Dynkin quiver attached to  $\Gamma$  via the McKay correspondence. We carry out a similar construction for continuous wreath product symplectic reflection algebras, a generalization to the case when  $\Gamma \subset SL(2,\mathbb{C})$  is infinite reductive. This time the main tool is the definition of a continuous analog of the deformed preprojective algebras for the infinite affine Dynkin quivers corresponding to the infinite reductive subgroups of  $SL(2, \mathbb{C})$ .

Thesis Supervisor: Pavel Etingof Title: Professor of Mathematics

### Acknowledgments

First of all, I would like to express my deep gratitude to my advisor Pavel Etingof for supporting and guiding me in my research, with great knowledge and great humanity. I especially want to thank him for being the most enthusiastic teacher I have ever met, and for organizing endless dinner time (and after dinner time) seminars for his students, in which we learned so many things! I would also like to thank Wee Liang Gan for his willingness to answer my questions, and for patiently explaining to me the details of his work [Gan06]. Thanks to Victor Kac and Kobi Kremnizer for agreeing to be in my thesis defense committee. Special thanks go also to my undergraduate Professors in Università di Roma "La Sapienza" Corrado De Concini and Giovanni Troianiello whose beautiful lectures made me curious about mathematics, and who encouraged me to study abroad.

I am very thankful also to all my classmates and friends at MIT for talking with me mathematics and life. Many thanks to: Alejandro, Alexey, Ana Rita, Angelica, Ben, Ching-Hwa, Craig, Chris, David, Dorian, Emanuel, Fucheng, Jennifer, Jerin, Josh, Karola, Matt, Max, Miki, Nick, Olga, Peter, Ramis, Ricardo, Ting, Xiaoguang, Yulan.

Several people have been essential for my survival in this country. I am especially thankful to Silvia Sabatini, with whom I shared and I share much more than our first names. Thanks to Gabriele for cooking meals full of taste, irony, and affection. Many thanks also to Nenad and Nataša, who always welcomed me to their home whether in Boston, New York or Belgrade and whom I considered my life-long friends from the beginning. Thanks to Matjaž and Danijel for teaching me the most beautiful Slovenian songs. Thanks to Vedran for being the kindest one, in any occasion. Thanks to Amanda for giving me the opportunity to watch "The Blues Brothers" with an authentic Illinoisan. Thanks to Daniel and "Buonarroti a tutti!". Thanks to Martin for sharing with me his thoughts, and his drawing sessions. Thanks to Yi-Vonne for being my contact with the outside world. Thanks to Rekha for a trip to Cape Cod with special sound effects, and for introducing me to Indian cuisine. Thanks to Cilanne and Étienne, Natasha and Oleg for becoming my friends despite the lack of time.

My life in Boston would have not been the same without my current and past exceptional house mates, who made 26 Calvin Street in Somerville one of my favorite places. So, thanks to Martina for many dinners high in calories and understanding, thanks to Kyle for so many "monkey" reasons, thanks to Lindsey for setting the pace with the supernatural energy of a professional runner and chemist, thanks to Gabriel for not being in science, and for taking me along with him in the amazing adventures of Don Quijote.

Looking across the ocean, many thanks go also to my fellow undergraduate and Ph.D. students in Rome. Among the others, thanks to Antonio, Barbara, Caterina, Chiara, Donatella, Elisa, Fabio Massimo, Giorgio, Giulia, Laura, Luca, Nicola, and Sara for the exciting years of study and friendship we have spent together.

My thanks also go to all my friends from Italy who comforted me when I was

missing home, and still do. Thanks to Camilla Cipolla, Raffaella Greco, Natalia Mancini, Gaia Maretto, Giovanna Miele, Chiara Muccitelli, Marco Pacini, Daniela Parisi, Valeria Porcu, Valeria and Valentina Siciliano, Ilaria Venturini. I will always be eager to go back and meet them. Lorenza Salvatori deserves a special thanks for many soothing phone calls and e-mails, and also for all the reciprocal thoughts that did not become either phone calls or e-mails.

I would also like to thank all the members of my small but "enlarged" family who encouraged me in my studies: my aunt, uncle and cousin Tersilia, Guglielmo and Renato, my dear grandmother Adriana, Anna, Ennio and Davide, Rossana, Sandra and Annamaria. My gratitude goes also to all the people who helped my grandmother during these years, and made our home in Latina a truly international place. Thanks to Cecilia, Yelena and Veronica for their caring presence.

A very special thanks goes to my dear Frédéric Rochon. His patience in dealing with my Italian instability in these years has been as vast as the vast land of Canada.

Finally I would like to thank Fosco and Loredana who, during this "American adventure", as well as throughout life, always advised me to look at the world and at the people with curiosity, generosity, empathy and respect. I am very lucky that they happen to be my parents.

## Contents

1	Bas	ic defe	ormation theory	17
	1.1	Plan o	of the chapter	17
	1.2	Flat f	ormal deformations of associative algebras	17
	1.3	Hochs	schild cohomology and deformation theory	19
	1.4	Flat f	ormal deformations of modules	26
<b>2</b>	Syn	nplecti	c reflection algebras	29
	2.1	Plan o	of the chapter	29
	2.2	Defini	tion and properties	29
	2.3	The w	vreath-product construction	33
3	Fin	ite din	nensional representations for $H_{1,k,c}(\mathbf{\Gamma}_n)$	39
	3.1	Plan o	of the chapter	39
	3.2	Exten	ding irreducible $\Gamma_n$ -modules	40
		3.2.1	Irreducible representations of wreath product groups $\ . \ . \ .$	40
		3.2.2	McKay correspondence	43
		3.2.3	Representations of $S_n$ with rectangular Young diagram	45
		3.2.4	The main theorem	46
		3.2.5	Proof of Theorem 3.2.4	48
	3.3	Defor	ming irreducible $H_{1,0,c}(\mathbf{\Gamma}_n)$ -modules	59
		3.3.1	A proposition in deformation theory $\ldots \ldots \ldots \ldots \ldots$	59
		3.3.2	Deformed preprojective algebras	62
		3.3.3	Irreducible representations of $H_{1,c}(\Gamma)$	63

		3.3.4 Irreducible representations of $H_{1,0,c}(\Gamma_n)$	66
		3.3.5 The main theorem	67
		3.3.6 Proof of Theorem 3.3.10	68
4	Coi	ntinuous symplectic reflection algebras	79
	4.1	Plan of the chapter	79
	4.2	Algebraic distributions	79
	4.3	Continuous symplectic reflection algebras	82
	4.4	The wreath product case	84
	4.5	Infinitesimal Hecke algebras	88
5	Coi	ntinuous deformed preprojective algebras	91
	5.1	Plan of the chapter	91
	5.2	Infinite affine quivers and the McKay correspondence for reductive sub-	
		groups of $SL(2,\mathbb{C})$	92
	5.3	Infinite rank affine root systems	93
	5.4	Action of the Weyl group on weights	96
	5.5	Definition of the continuous deformed preprojective algebra	99
		5.5.1 The rank one case	99
		5.5.2 The higher rank case	102
	5.6	Morita equivalence	104
6	Fin	ite dimensional representations for the continuous case 1	107
	6.1	Plan of the chapter	107
	6.2	Representations in the rank one case	107
	6.3	The $SL(2,\mathbb{C})$ case	109
	6.4	Gan's reflection functors	111
	6.5	Representations in the higher rank case	117
A	Pro	oof of Theorem 5.6.1 1	123
Bi	iblio	graphy 1	135

## List of Figures

3-1	Graph $\tilde{A}_n$	. 44
3-2	Graph $\tilde{D}_n$	. 44
3-3	Graph $\tilde{E}_6$	. 44
3-4	Graph $\tilde{E}_7$	. 45
3-5	Graph $\tilde{E}_8$	. 45
5-1	Graphs associated to $GL(1,\mathbb{C})$ , $\widetilde{O}_2$ , and $SL(2,\mathbb{C})$	. 93

## Introduction

The study of symplectic reflection algebras was initiated by Etingof and Ginzburg in [EG02], although these algebras (and several of their properties) already appear in the classical work of Drinfeld [Dri86], as a special case of degenerate affine Hecke algebras for a finite group.

Symplectic reflection algebras arise from the action of a finite group of symplectomorphisms  $G \subset \operatorname{Sp}(V)$  on a symplectic vector space  $(V, \omega)$ . They form a multiparameter family of deformations  $H_{t,f}(G)$  of the skew group algebra  $SV \sharp G$ , where SV is the symmetric algebra of V. The parameter t is a complex number, and the parameter f is a conjugation invariant function on the the set S of symplectic reflections in G, i.e. elements  $s \in G$  such that  $\operatorname{rk}(\operatorname{Id} - s)|_V = 2$ . Symplectic reflections can be considered as analogs of reflections in a symplectic setting, since they fix a codimension two subspace pointwise, and act on the orthogonal complement with nontrivial complex conjugate eigenvalues of norm one. Hence the name.

Explicitly, if we denote by TV the tensor algebra of V, the symplectic reflection algebra  $H_{t,f}(G)$  is the quotient of  $TV \sharp G$  by the relations

$$x \otimes y - y \otimes x = t\omega(x, y) \cdot 1 + \sum_{s \in \mathcal{S}} f(s) \cdot \omega_s(x, y) \cdot s \quad \forall x, y \in V$$

where  $\omega_s$  denotes the (possibly degenerate) skew-symmetric form which coincides with  $\omega$  on Im(Id - s), and has ker(Id - s) as its radical.

One of the fundamental properties of  $H_{t,f}(G)$  is that it satisfies the analog of the Poincaré-Birkhoff-Witt (PBW) theorem for the universal enveloping algebra of a Lie algebra. Namely, if we consider the increasing filtration on  $H_{t,f}(G)$  obtained by assigning degree zero to the elements of G and degree one to the elements of V, we get an isomorphism for the associated graded algebra  $\operatorname{gr} H_{t,f}(G) \cong SV \sharp G$ . The PBW property assures the flatness of the family of deformations  $H_{t,f}(G)$ .

Rescaling the parameters (t, f) by a non-zero complex number does not change  $H_{t,f}(G)$  up to isomorphism. Thus we can distinguish two main cases in the study of symplectic reflection algebras: the *quasi-classical* case when t = 0, and the *quantum* case when t = 1. Algebras belonging to these two subfamilies present contrasting features that make them interesting for different reasons, both in algebraic geometry and representation theory.

On the geometric side, let us consider the orbit space V/G. This space and the corresponding commutative algebra, the ring of invariants  $SV^G$ , might not be the right objects to describe the geometric properties of the G-action, that might instead be connected with some resolutions of V/G. One can then think to approach the study of such properties by replacing  $SV^G$  with the non-commutative smash product  $SV \sharp G$ , and constructing non-commutative deformations of this algebra. In the quasiclassical case, for example, the non-commutative deformation  $H_{0,f}(G)$  of  $SV \sharp G$  has a big center which is a commutative deformation of  $SV^G$  (i.e. corresponding to an actual algebraic variety), and can be used to study symplectic resolutions of some interesting (Poisson) deformations of the orbifold V/G ([EG02]), [GS04]). A class of symplectic reflection algebras of particular interest for algebraic geometry is the one of rational Cherednik algebras. These are symplectic reflection algebras attached to an irreducible finite complex reflection group W in a vector space  $\mathfrak{h}$ , acting diagonally on  $V = \mathfrak{h} \oplus \mathfrak{h}^*$ , where  $\mathfrak{h}^*$  denotes the dual of the reflection representation. In this case, the symplectic form is given by the natural pairing. In other words, V is the cotangent bundle of  $\mathfrak{h}$  with the standard symplectic structure, and  $\mathfrak{h}$ ,  $\mathfrak{h}^*$  are W-stable irreducible Lagrangian subspaces. In the quantum case, the Cherednik algebra  $H_{1,f}(\mathfrak{h} \oplus \mathfrak{h}^*, W)$ for W of type A can be regarded in different ways as a non-commutative deformation of the Hilbert scheme of points in the plane  $\mathbb{C}^2$  ([GS06],[GS05],[KR]).

On the representation theoretic side, the most challenging case is the quantum one which is the most non-commutative one. Indeed, when t = 1 and regarding f as a formal parameter, the family  $H_{1,f}(G)$  gives a universal deformation of the smash product  $\mathcal{W} \sharp G$ , where  $\mathcal{W}$  is the Weyl algebra of the symplectic space  $(V, \omega)$  ([EG02]). The representation theory of such deformations has proved to be very rich and interesting. In particular, for rational Cherednik algebras an analog of category  $\mathcal{O}$  for Lie algebras has been defined, as well as a theory of standard modules and formal characters ([BEG03], [GGOR03], [Chm06]). Moreover, in the case of Cherednik algebras, the theory of symplectic reflection algebras is connected to the one of double affine Hecke algebras (of which they are a certain degeneration, cfr [EG02], Introduction), introduced by I. Cherednik and used by the same author to prove some important Macdonald's conjectures ([Che95]). This links the representation theory of symplectic reflection algebras to combinatorics and the study of special functions.

The main topic of this thesis is the representation theory of the symplectic reflection algebras of wreath product type  $H_{t,k,c}(\mathbf{\Gamma}_n)$ . These are symplectic reflection algebras attached to the semi-direct products  $\mathbf{\Gamma}_n := S_n \ltimes \Gamma^n \subset \operatorname{Sp}(2n, \mathbb{C})$ . Here  $S_n$ is the symmetric group on n letters,  $\Gamma$  is any finite subgroup of  $SL(2, \mathbb{C}) = \operatorname{Sp}(2, \mathbb{C})$ , and  $S_n$  acts on  $\Gamma^n$  by permuting the factors. The deformation parameter f appearing in the definition of symplectic reflection algebra can in this case be written as a pair (k, c), where k is a complex number, and c is a class function on the non-trivial elements of  $\Gamma$ . The integer n is called the rank of the algebra  $H_{t,k,c}(\mathbf{\Gamma}_n)$ . In particular, when  $\Gamma = \mathbb{Z}/m\mathbb{Z}$  the group  $S_n \ltimes (\mathbb{Z}/m\mathbb{Z})^n$  is a complex reflection group (real reflection group of type A if m = 1 and of type B if m = 2) and one recovers a subfamily of rational Cherednik algebras.

In the quasi-classical case, representations of the wreath product algebras were studied in [EG02] using a geometric approach and the main result of the authors is that, if (k, c) are generic, isomorphism classes of finite dimensional irreducible modules are parametrized by the points of the (smooth) algebraic variety corresponding to the center of  $H_{0,k,c}(\Gamma_n)$ .

In the quantum case, in contrast with the case of Cherednik algebras, a uniform approach to the representation theory does not exist yet. Nevertheless, when n = 1, there is no parameter k and  $H_{1,c}(\Gamma)$  coincides with some non-commutative deformation of the Kleinian singularity  $\mathbb{C}^2/\Gamma$  introduced by Crawley-Boevey and Holland, who classified finite dimensional irreducible modules using methods coming from the representation theory of quivers and deformed preprojective algebras, which are some special quotients of path algebras of quivers. In particular, in [CBH98] the authors established a Morita equivalence of  $H_{1,c}(\Gamma)$  with the deformed preprojective algebra  $\Pi_{\lambda}(Q)$ , where Q is the extended (ADE) Dynkin quiver attached to  $\Gamma$  via the McKay correspondence, and  $\lambda \in \mathbb{C}^I$ , where I is the set of vertices of Q, is a parameter depending on c. This allowed them to define reflection functors that give equivalences of the categories of modules for different values of the deformation parameters.

The main result of this thesis is the construction of the first (for non-cyclic  $\Gamma$ ) families of finite dimensional representations for  $H_{1,k,c}(\Gamma_n)$  when n > 1, and (k, c)vary in some linear subspaces of the space of deformation parameters. We use two methods, both arising from simple observations and corresponding natural questions.

- Irreducible representations of the group Γ<sub>n</sub> are well known. They can all be obtained in the following way. Choose any vector with positive integer coordinates n

   (n<sub>1</sub>,...,n<sub>r</sub>) such that Σ<sup>r</sup><sub>i=1</sub> n<sub>i</sub> = n. Let S<sub>ni</sub> be the subgroup of S<sub>n</sub> moving only the indices j such that n<sub>1</sub>+...+n<sub>i-1</sub> < j ≤ n<sub>1</sub>+...+n<sub>i</sub>, and consider the subgroup S<sub>n1</sub> × ... × S<sub>nr</sub> ⊂ S<sub>n</sub>. Take an irreducible representation X of the group S<sub>n</sub>: X := X<sub>1</sub> ⊗ ... ⊗ X<sub>r</sub>, where X<sub>i</sub> is an irreducible representation of S<sub>ni</sub> for any i. Choose a collection N<sub>1</sub>,..., N<sub>r</sub> of irreducible pairwise non-isomorphic representations of Γ, and form the irreducible representation N := N<sub>1</sub><sup>⊗n1</sup> ⊗ ... ⊗ N<sub>r</sub><sup>⊗nr</sup> of Γ<sup>n</sup>. Then X ⊗ N is an irreducible S<sub>n</sub> × Γ<sup>n</sup>-module (where S<sub>ni</sub> acts also on N by permuting the factors) and M := Ind<sup>Γn</sup><sub>S<sub>ni</sub>×Γ<sup>n</sup></sub>X ⊗ Y is an irreducible Γ<sub>n</sub>-module. Which such modules can be extended to an irreducible representation of the entire algebra H<sub>1,k,c</sub>(Γ<sub>n</sub>), and for which values of the parameters (k, c)?
- 2) Since  $H_{1,0,c}(\Gamma_n) = H_{1,c}(\Gamma)^{\otimes n} \sharp S_n$ , when k is zero, irreducible finite dimensional representations are known. They can all be obtained with the same procedure used in 1). We get modules  $M = \operatorname{Ind}_{H_{1,c}(\Gamma)^{\otimes n} \sharp S_{\vec{n}}}^{H_{1,0,c}(\Gamma_n)} X \otimes Y$ , where this time we take

 $Y = Y_1^{\otimes n_1} \otimes \cdots \otimes Y_r^{\otimes n_r}$ , for some  $Y_i$ s irreducible pairwise non-isomorphic  $H_{1,c}(\Gamma)$ modules. Since, by the PBW property,  $H_{1,k,c+c'}(\Gamma_n)$  is a flat formal deformation of  $H_{1,0,c}(\Gamma_n)$  a natural question is: can we formally deform a  $H_{1,0,c}$ -module Mto values of the parameters (k, c + c') with  $k \neq 0$ ?

In Theorem 3.2.4, we give an answer to question 1). Using just methods from representation theory of finite groups, we obtain a complete classification of all irreducible  $\Gamma_n$ -modules that extend to  $H_{1,k,c}(\Gamma_n)$ -modules for  $k \neq 0$ . Such modules extend if and only if the Young diagram of  $X_i$  is a rectangle for any i and  $\operatorname{Hom}_{\Gamma}(N_i, N_j \otimes \mathbb{C}^2) = 0$ for any  $i \neq j$ , where  $\mathbb{C}^2$  is the defining representation of  $\Gamma$ . Such representations have a unique extension obtained by making V act trivially, and the values of the parameters (k, c) for which they extend lie in a codimension r linear subspace, where r is the dimension of the vector  $\vec{n}$ .

In Theorem 3.3.10, using cohomological methods, we give a partial answer to the second question. We show that sufficient conditions for an irreducible representation of  $H_{1,0,c}(\mathbf{\Gamma}_n)$  to formally deform to some values of the parameter with  $k \neq 0$  are that  $X_i$  has rectangular Young diagram for any *i* and that  $\operatorname{Ext}^1_{H_{1,c}(\Gamma)}(Y_i, Y_j) = 0$ for any  $i \neq j$ . Such representations actually admit a unique deformation in the formal neighborhood of 0 of a codimension r linear subspace. We also show that in a dense open set of this linear subspace the deformation is not only formal, i.e.  $H_{1,k,c+c'}(\Gamma_n)$  admits an irreducible representation isomorphic to M as a  $\Gamma_n$ -module. We want to mention that in [GG05] Gan and Ginzburg introduced a one parameter deformation  $\mathcal{A}_{n,\nu,\lambda}(Q)$  (where  $\nu \in \mathbb{C}$ ) of the smash product  $\Pi_{\lambda}(Q)^{\otimes n} \sharp S_n$ , Morita equivalent to  $H_{1,k,c}(\Gamma_n)$  for any n, when Q is the McKay quiver of  $\Gamma$ . In ([Gan06]) Gan, using this interpretation of the wreath product symplectic reflection algebras in terms of deformed preprojective algebras, was able to generalize the reflection functors of [CBH98] to the case n > 1. This allowed him to prove the necessity of the conditions of Theorem 3.3.10, showing that our result gives an exhaustive classification of representations coming from deformations.

Symplectic reflection algebras have a generalization to reductive algebraic groups

called continuous symplectic reflection algebras ([EGG05]). In this case, the role of the group algebra is played by the ring of algebraic distributions  $\mathcal{O}(G)^*$ , the dual space of the ring of regular functions  $\mathcal{O}(G)$ . The second topic of this thesis is the study of finite dimensional representations of continuous symplectic reflection algebras of wreath product type, i.e. attached to the groups  $S_n \ltimes \Gamma^n$ , where  $\Gamma \subset SL(2, \mathbb{C})$ is an infinite reductive subgroup. This time the main tool is the definition of some "continuous" analogs of the deformed preprojective algebras for the infinite affine Dynkin quivers corresponding to the reductive subgroups of  $SL(2,\mathbb{C})$ , and of the corresponding generalization to such quivers of the Gan-Ginzburg algebra  $\mathcal{A}_{n,\nu,\lambda}(Q)$ . A Morita equivalence between these algebras and the continuous symplectic reflection algebras allows us to easily extend the methods of [CBH98] and [Gan06] to the continuous case. In particular, in Corollary 6.2.2 we give a complete classification of the finite dimensional irreducible representations for n = 1. For n > 1, in Theorem 6.5.2 and Theorem 6.5.3, we extend the results of Theorem 3.2.4 and Theorem 3.3.10 to the continuous case, giving necessary and sufficient conditions for deforming irreducible finite dimensional representations existing for k = 0 to nonzero values of k.

## Chapter 1

## **Basic deformation theory**

### 1.1 Plan of the chapter

In this chapter we first review the basic definitions of the theory of flat formal deformations for associative algebras. We then recall the fundamental role of Hochschild cohomology in this theory and the notion of *universal deformation*. Finally, we briefly discuss deformations of modules.

# 1.2 Flat formal deformations of associative algebras

Let k be a field, and let A be an associative unital algebra over k. Let U be a finite dimensional k-vector space. Denote by k[[U]] the ring of k-valued formal functions on U, and denote by  $\mathfrak{m}$  the unique maximal ideal in k[[U]].

We recall that a k[[U]]-module is called *topologically free* if it is isomorphic to V[[U]] for some k-vector space V.

**Definition 1.2.1.** A flat formal deformation of A over k[[U]] is an algebra  $A_U$  over k[[U]] which is topologically free as a k[[U]]-module, together with a fixed isomorphism of algebras  $\varphi : A_U/\mathfrak{m}A_U \longrightarrow A$ .

Thus in particular  $A_U = A[[U]]$  as a k[[U]]-module. If dim U = n, then  $A_U$  is said to be an *n*-parameter flat formal deformation of A.

Two deformations  $A_U$ ,  $A'_U$  are said to be isomorphic if there exists a k[[U]]-algebra isomorphism  $A_U \cong A'_U$  which is the identity modulo  $\mathfrak{m}$ . A deformation is said to be trivial if there exists an algebra isomorphism  $A_U \cong A[[U]]$  which is the identity modulo  $\mathfrak{m}$ , where the algebra structure on A[[U]] is given by the usual multiplication of formal power series on U with coefficients in A.

In a similar way, one can define *m*-th order deformations as deformations over the ring  $k[[U]]/\mathfrak{m}^{m+1}$ .

If  $\hbar_1, \ldots, \hbar_n$  are coordinates on U, then we can identify  $\mathbb{C}[[U]]$  with the ring of power series  $k[[\hbar_1, \ldots, \hbar_n]]$ , and the ideal  $\mathfrak{m}$  with the ideal  $(\hbar_1, \ldots, \hbar_n)$ . Using the fact that  $A_U$  is topologically free we can choose an identification  $\tilde{\varphi} : A_U \longrightarrow A[[\hbar_1, \ldots, \hbar_n]]$ as  $k[[\hbar_1, \ldots, \hbar_n]]$ -modules, coinciding with the isomorphism  $\varphi$  of Definition 1.2.1 modulo  $(\hbar_1, \ldots, \hbar_n)$ . Let us denote by  $\underline{p} = (p_1, \ldots, p_n) \in \mathbb{Z}_{\geq 0}^n$  a multi-index, and let  $\mathbf{h}^p$  be the product  $\prod_j \hbar_j^{p_j}$ . We can think of  $A_U$  as the module  $A[[\hbar_1, \ldots, \hbar_n]]$  equipped with a new  $k[[\hbar_1, \ldots, \hbar_n]]$ -linear, associative *star-product* determined by a formula

$$a * b = \sum_{\underline{p}} c_{\underline{p}}(a, b) \mathbf{h}^{\underline{p}}$$
(1.1)

where  $c_{\underline{p}} : A \times A \longrightarrow A$  are k-bilinear maps, and  $c_{0,\dots,0}(a,b) = ab$  for any  $a, b \in A$ (the product coincides with the product in A modulo  $(\hbar_1, \dots, \hbar_n)$ ).

In particular, a one parameter deformation  $A_{\hbar}$  can be thought of as  $A[[\hbar]]$  equipped with a  $k[[\hbar]]$ -linear associative product \* such that for any  $a, b \in A$ 

$$a * b = ab + c_1(a, b)\hbar + c_2(a, b)\hbar^2 + \cdots,$$
 (1.2)

where  $c_j : A \times A \longrightarrow A$  are k-bilinear maps, and  $c_0(a, b) = ab$  is just the original product in A.

One can think of  $c_1$ , and the corresponding first order deformation (deformation over the ring  $k[[\hbar]]/\hbar^2$ ), as the *infinitesimal deformation* or *differential* of the family  $A_{\hbar}$ . This leads to two natural questions. The first is finding a classifying space for infinitesimal deformations. The second is defining a convenient theoretical setting to describe the obstructions to *integrating* infinitesimal deformations, i.e. given a first order deformation  $c_1$ , lifting the associativity property of the product \* from order one to any order by choosing appropriate  $c_i$ s for i > 1. In his pioneering work ([Ger63],[Ger64]), Gerstenhaber showed how the natural language to approach these problems is the one of homological algebra, specifically the one of Hochschild cohomology that we are going to review in the next section.

## 1.3 Hochschild cohomology and deformation theory

For an A-bimodule E, consider the following complex (Hochschild complex)

$$0 \longrightarrow C^{0}(A, E) \xrightarrow{d} \cdots \xrightarrow{d} C^{m}(A, E) \xrightarrow{d} C^{m+1}(A, E) \xrightarrow{d} \cdots$$

where  $C^m(A, E) = \operatorname{Hom}_k(A^{\otimes m}, E)$  is the space of *m*-linear maps from  $A^m$  to *E* (and  $C^0(A, E) := E$ ), and the differential *d* is defined as follows:

$$(de)(a): = ae - ea \qquad \forall e \in E, a \in A$$

$$(df)(a_1, \cdots, a_{m+1}) := a_1 f(a_2, \cdots, a_{m+1}) + \sum_{i=1}^m (-1)^i f(a_1, \cdots, a_{i-1}, a_i a_{i+1}, a_{i+2}, \cdots, a_{m+1}) - (-1)^m f(a_1, \cdots, a_m) a_{m+1}.$$

**Definition 1.3.1.** The *i*-th Hochschild cohomology group  $H^i(A, E)$  of A with coefficients in the bimodule E is the *i*-th cohomology group of the Hochschild complex  $(C^{\bullet}, d)$ .

We recall that an A-bimodule structure on a k vector space space E is the same

as a left  $A \otimes A^{o}$ -module structure, where  $A^{o}$  is the opposite algebra of A, and  $A \otimes A^{o}$  is called the enveloping algebra of A. The following fact will be very useful to us in this thesis.

Proposition 1.3.2. There exists a natural isomorphism

$$H^i(A, E) \longrightarrow \operatorname{Ext}^i_{A \otimes A^o}(A, E)$$

*Proof.* Consider the A-bimodule structure on  $A^{\otimes m}$  given by

$$b(a_1 \otimes \cdots \otimes a_m)c = ba_1 \otimes \cdots \otimes a_mc$$

The so called bar resolution of the bimodule A is a projective resolution and is given by

 $\cdots \longrightarrow A^{\otimes 3} \longrightarrow A^{\otimes 2} \longrightarrow A$ 

where the differential is

$$\partial(a_1 \otimes \cdots \otimes a_m) = a_1 a_2 \otimes \cdots \otimes a_m - \cdots + (-1)^{m-1} a_1 \otimes \cdots \otimes a_{m-1} a_m.$$

It is now enough to observe that, for any  $m \ge 2$ , one has a natural isomorphism of vector spaces  $\operatorname{Hom}_{A\otimes A^o}(A^{\otimes m}, E) \cong \operatorname{Hom}_k(A^{m-2}, E) = C^{m-2}(A, E)$ , and that  $\partial$ corresponds to d under this identification.

г	_	
L		
L.		

Let us now go back to one parameter deformations and formula (1.2). Imposing the associativity condition

$$(a * b) * c = a * (b * c)$$

one gets a hierarchy of equations

$$dc_{m}(a, b, c) = \sum_{\substack{i+j=m\\i,j>0}} c_{i}(c_{j}(a, b), c) - c_{i}(a, c_{j}(b, c))$$
(1.3)

where m > 0 and d is the Hochschild differential. In particular, when m = 1 the right hand side is zero, and one gets that associativity in order one holds if and only if  $c_1$ is a Hochschild 2-cocycle with values in the bimodule A. It is easy to see that two first order deformations are isomorphic if and only if the associated 2-cocycles  $c_1$ ,  $c'_1$ are in the same cohomology class in  $H^2(A, A)$ .

**Theorem 1.3.3.** Two first order one parameter deformations of an associative algebra A are isomorphic if and only if the corresponding 2-cocycles  $c_1$ ,  $c'_1$  are in the same Hochschild cohomology class.

Proof. Suppose  $c_1$ ,  $c'_1$  are in the same cohomology class. Then  $c_1 - c'_1 = df$ , where f is an endomorphism of A as a k-vector space. Then one can check that the assignment

$$a \longrightarrow a + f(a)\hbar$$

for any  $a \in A$  defines by  $k[[\hbar]]$ -linear extension an isomorphism between the corresponding first order deformations. Vice versa suppose that two first order deformations are isomorphic. This means that there exists a  $k[[\hbar]]/\hbar^2$ -algebra isomorphism between them which is the identity modulo  $\hbar$ . Such an isomorphism is given by a map

$$a \longrightarrow a + f(a)\hbar$$

where f is a k linear endomorphism of A, and the map is compatible with the star products \*, \*' modulo  $\hbar^2$ . Imposing this compatibility condition gives exactly  $c_1 - c'_1 = df$ .

Thus the cohomology group  $H^2(A, A)$  parametrizes infinitesimal deformations up to isomorphism.

For m = 2 equation (1.3) gives

$$dc_2(a, b, c) = c_1(c_1(a, b), c) - c_1(a, c_1(b, c))$$

for any  $a, b, c \in A$ . It can be computed that, if  $c_1$  is a cocycle, then the right hand side defines a 3-cocycle  $b_1$  and its cohomology class depends only on the cohomology class of  $c_1$ . The cohomology class of  $b_1$  in  $H^3(A, A)$  is the only obstruction to lifting associativity from order one to order two.

In general, it can be shown that if equation (1.3) is satisfied for  $m = 1, \ldots, M - 1$ then for m = M the right hand side is a cocycle  $b_M$ . Thus all obstructions lie in  $H^3(A, A)$ . This time though, the cohomology class of  $b_M$  depends not only on the cohomology class  $[c_1]$  of  $c_1$  but also on the entire sequence  $c_2, \ldots, c_{M-1}$ . This is why lifting an infinitesimal deformation to a family  $A_h$  is a highly non trivial problem. Indeed, if maps  $c_1, \ldots, c_{M-1}$  satisfying (1.3) are chosen, then any two maps  $c_M, c'_M$ compatible with them must satisfy  $d(c_M - c'_M) = 0$ , i.e. they must differ by a cocycle. Moreover if two solutions differ by a coboundary they give rise to isomorphic M-th order deformations. In other words, the freedom in choosing the solution at step M lies in  $H^2(A, A)$ . The problem is that a particular choice of  $c_M$  affects the equations in the hierarchy for all m > M, and can determine an obstruction at any of the next steps. If  $H^3(A, A)$  is zero though, all obstructions vanish and any first order deformation can be lifted, i.e. it is the differential of a family of flat formal deformations.

From the previous discussion it appears that the space  $H^2(A, A)$  is a natural candidate to parametrize a *universal deformation* for the algebra A, in a sense that we are going to clarify.

Let us look at formula (1.1) for a deformation with parameters in an n dimensional vector space U. Imposing the associativity condition and arguing as in the one parameter case, one obtains that  $c_{0,...,1_j,...0}$  must be Hochschild 2-cocycles for each j. Thus any such deformation defines a natural linear map  $\phi$  from the space of parameters U to  $H^2(A, A)$ . Such a map is given by the assignment

$$\phi: \qquad U \qquad \longrightarrow \qquad H^2(A, A)$$
$$(\hbar_1, \cdots, \hbar_n) \qquad \longrightarrow \qquad \sum_j \hbar_j \left[ c_{0, \dots, 1_j, \dots 0} \right]$$

for any  $(\hbar_1, \dots, \hbar_n) \in U$ , where [C] stands for the cohomology class of a cocycle C.

**Proposition 1.3.4.** If the space  $H^2(A, A)$  is finite dimensional and  $H^3(A, A) = 0$ then there exists a flat formal deformation  $A^{\mathfrak{u}}$  over k[[U]], where  $U := H^2(A, A)$ , such that the map  $\phi$  above is the identity. This deformation is unique up to isomorphism (we allow automorphisms of k[[U]]).

Proof. Suppose dim  $H^2(A, A) = n$ . For a multi-index  $\underline{p} = (p_1, ..., p_n) \in \mathbb{Z}_{\geq 0}^n$ , let  $|\underline{p}| = p_1 + \cdots + p_n$  denote its length. For any j = 1, ..., n let  $e_j$  be the multi-index  $(0, ..., 1_j, ..., 0)$ . Fix a basis  $[c_{e_1}], \ldots, [c_{e_n}]$  for  $H^2(A, A)$  and let  $\hbar_1, \ldots, \hbar_n$  be coordinates relative to this basis. We claim that, up to isomorphism and automorphisms of k[[U]], a deformation as in the statement of the theorem must be given by a formula

$$a * b = ab + \sum_{i=1}^{n} c_{e_i}(a, b)\hbar_i + \sum_{|\underline{p}| > 1} c_{\underline{p}}(a, b)\mathbf{h}^{\underline{p}}$$
(1.4)

for some choice of representatives  $c_{e_1}, \ldots, c_{e_n}$  of the above basis and for some kbilinear maps  $c_{\underline{p}} : A \times A \to A$ . This is true because, first of all, the map  $\phi$  must be the identity. Secondly, arguing as in the proof of Theorem 1.3.3, one can see that different choices of representatives for the basis  $[c_{e_1}], \ldots, [c_{e_n}]$  do not affect the representation at order one up to isomorphism. Finally a change of basis in  $H^2(A, A)$  changes such a deformation by the corresponding induced automorphism of  $k[[U]] \cong k[[\hbar_1, ..., \hbar_n]]$ .

It is easy to compute that the condition that formula (1.4) defines a deformation over  $k[[\hbar_1, ..., \hbar_n]]$  gives, for each p with |p| > 1, an equation

$$dc_{\underline{p}} = b_{\underline{p}} \tag{1.5}$$

where  $b_{\underline{p}}$  is a 3-cocycle whose expression may involve  $c_{\underline{q}}$  only for  $|\underline{q}| < |\underline{p}|$ . Since  $H^3(A, A) = 0$  any 3-cocycle is a coboundary, thus we can recursively solve the equations above and find  $c_p$ s that give a deformation as desired.

What is left now to show is that such a deformation is unique. We discussed uniqueness at order one. It remains to prove that the deformation does not depend on our choices of  $c_{\underline{p}}$ s for  $|\underline{p}| > 1$  up to isomorphism (possibly involving automorphisms of  $k[[\hbar_1, ..., \hbar_n]]$ ). We will show this by induction on  $|\underline{p}|$ . Let now  $A^{\mathfrak{u}}$  and  $A'^{\mathfrak{u}}$  be two deformations given by two sets of maps  $\{c_{\underline{p}}\}, \{c'_{\underline{p}}\}$  respectively. Let N be the maximal number such that, up to automorphism of  $A^{\mathfrak{u}}$  and  $A'^{\mathfrak{u}}$ , it is possible to set  $c_{\underline{p}} = c'_{\underline{p}}$  for any  $\underline{p}$  such that  $|\underline{p}| < N$ . Then for any  $\underline{q}$  with  $|\underline{q}| = N$  we have  $b_{\underline{q}} = b'_{\underline{q}}$ , where  $b_{\underline{q}}$  and  $b'_{\underline{q}}$  are the cocycles on the right hand side of equation (1.5) for  $c_{\underline{q}}$  and  $c'_{\underline{q}}$  respectively. This means that  $d(c_{\underline{q}} - c'_{\underline{q}}) = 0$  i.e. the two maps differ by a cocycle. Thus we can write

$$c_{\underline{q}} = c'_{\underline{q}} + \sum_{j} \alpha_{\underline{q}j} c_{e_j}$$

with  $\alpha_{\underline{q}\underline{j}} \in \mathbb{C}$ . Consider now the automorphism  $\psi_N$  of  $k[[\hbar_1, ..., \hbar_n]]$  defined by the assignment

$$\hbar_j \longrightarrow \hbar_j + \sum_{|\underline{q}|=N} \alpha_{\underline{q}j} \mathbf{h}^{\underline{q}}.$$

It is easy to see that twisting  $A'^{u}$  with  $\psi_{N}$  we can set  $c_{\underline{q}} = c'_{\underline{q}}$  for any  $\underline{q}$  with  $|\underline{q}| = N$ , without affecting the  $c_{\underline{p}}$ s with  $|\underline{p}| < N$ . This contradicts the maximality of N.

When it exists, the deformation  $A^{\mathfrak{u}} = A^{\mathfrak{u}}_{\hbar_1,\dots,\hbar_n}$  of Theorem 1.3.4 is called the *universal deformation* of A.

The existence of the universal deformation guarantees that the moduli space of one parameter deformations is a smooth space, given by the formal neighborhood of zero in  $H^2(A, A)$ . This fact is a consequence of the following proposition stating the universal property of  $A^{u}$ .

**Proposition 1.3.5.** For every flat formal one parameter deformation  $A_{\hbar}$  there exists a unique power series  $\alpha(\hbar) = (\alpha_1(\hbar), \ldots, \alpha_n(\hbar)) \in \hbar H^2(A, A)[[\hbar]]$  such that there is an isomorphism  $A_{\hbar} \cong A^{\mathfrak{u}}_{\alpha_1(\hbar),\dots,\alpha_n(\hbar)}$  of flat formal deformations. Moreover  $\alpha'(0)$  is the cohomology class of the differential  $c_1$  of the family  $A_{\hbar}$ .

Proof. Let us denote by  $*_{\mathfrak{u}}$  the multiplication in  $A^{\mathfrak{u}}$ . Suppose dim  $H^2(A, A) = n$ . For any a, b in A, regard  $a *_{\mathfrak{u}} b = a *_{\mathfrak{u}} b(\hbar_1, \ldots, \hbar_n)$  as a formal function from  $U := H^2(A, A)$  to A. Consider now a one parameter deformation  $A_{\hbar} = (A[[\hbar]], *)$ , and regard  $a * b = a * b(\hbar)$  as a formal function from k to A. We have to show that, up to isomorphism, the algebra  $A_{\hbar}$  is given by a formula  $a * b(\hbar) = a *_{\mathfrak{u}} b(\alpha(\hbar))$  for a unique power series.

Thus we have to find  $\alpha_i(\hbar) \in \hbar k[[\hbar]]$  for i = 1, ..., n such that, for any  $a, b \in A$ , the identity

$$\sum_{j} c_j(a,b)\hbar^j = \sum_{i} c_{e_i}(a,b)\alpha_i(\hbar) + \sum_{|p|>1} c_{\underline{p}}(a,b)\alpha_1(\hbar)^{p_1} \cdots \alpha_n(\hbar)^{p_n}$$
(1.6)

is satisfied. If  $\alpha_i(\hbar) = \sum_{r>1} \alpha_{ir} \hbar^r$  then one can compute that the condition for (1.6) to be satisfied at order one is

$$\sum_{i} \alpha_{i1}[c_{e_i}] = [c_1].$$

Clearly there exist unique  $\alpha_{11}, \ldots, \alpha_{1n}$  satisfying this equation. In general one can compute that the condition that (1.6) is satisfied at order n is given by

$$\sum_{i} \alpha_{in}[c_{e_i}] = [\tilde{c}_n]$$

where  $\tilde{c}_n$  is a cocycle depending on  $\alpha_{ij}$  with j < n. It is clear that this equation has a unique solution.

We want to end this section by mentioning that the conditions of Proposition 1.3.5 are not necessary for the existence of the universal deformation. Indeed, all formal obstructions to deformations can vanish even when  $H^3(A, A)$  is nonzero, although this fact might be extremely difficult to prove.

#### **1.4** Flat formal deformations of modules

Let now M be a left A-module. Let  $A_U$  be a flat formal deformation of A over k[[U]], where U is some finite dimensional vector space, and let  $\varphi : A \longrightarrow A_U/\mathfrak{m}A_U$  be the fixed isomorphism of Definition 1.2.1. The following definition formalizes the intuitive notion of a deformation of M to an  $A_U$ -module.

**Definition 1.4.1.** A flat formal deformation of the module M is a  $A_U$ -module  $M_U$ which is topologically free as a k[[U]]-module together with a fixed isomorphism of A-modules  $\gamma : M \longrightarrow M_U/\mathfrak{m}M_U$ , where the structure of A module on  $M_U/\mathfrak{m}M_U$  is the one induced by the isomorphism  $\varphi$ .

The fundamental tool for the study of deformations of modules is again Hochschild cohomology.

Indeed, let  $\hbar_1, \ldots, \hbar_n$  be coordinates in U and let  $\underline{p} = (p_1, \ldots, p_n)$  be a multiindex in  $\mathbb{Z}_{\geq 0}^n$ . Arguing as in Section 1.2, we can think of  $M_U$  as the  $k[[\hbar_1, \ldots, \hbar_n]]$ module  $M[[\hbar_1, \ldots, \hbar_n]]$  together with a  $k[[\hbar_1, \ldots, \hbar_n]]$  algebra homomorphism  $\tilde{\rho} : A \longrightarrow$ End  $M[[\hbar_1, \ldots, \hbar_n]]$  given by a formula

$$\widetilde{\rho}(a) = \sum_{\underline{p}} \rho_{\underline{p}}(a) h^{\underline{p}}, \qquad (1.7)$$

where  $\rho_{\underline{p}} : A \longrightarrow \text{End } M$  are k-linear maps and  $\rho_{0,\dots,0}(a) = \rho(a)$ , where  $\rho$  is the homomorphism giving the representation M.

Imposing the condition that  $\tilde{\rho}$  is a homomorphism gives a hierarchy of equations

$$d\rho_{\underline{p}}(a,b) = -\sum_{\substack{\underline{q} + \underline{s} = \underline{p} \\ |\underline{q}|, |\underline{s}| > 0}} \rho_{\underline{q}}(a)\rho_{\underline{s}}(b) + \sum_{\substack{\underline{q} + \underline{s} = \underline{p} \\ |\underline{q}|, |\underline{s}| > 0}} \rho_{\underline{q}}(c_{\underline{s}}(a,b))$$
(1.8) (1.8)

where d is the Hochschild differential and the  $c_{\underline{s}}$ s are the maps defining the product in  $A_U$  as in formula (1.1).

If  $\rho_q$ s satisfy the above equation for all q with |q| < N, then for any p with |p| = N

the right hand side is a cocycle. Thus, all the obstruction to the deformation of the module M lie in  $H^2(A, \operatorname{End} M)$ . At each step, the freedom in choosing a solution for (1.8) lies in  $H^1(A, \operatorname{End} M)$ .

It is clear from this discussion that the problem of deforming modules presents similar difficulties to the one of deforming algebras, and that in general an A-module Mdoes not admit any deformation to a representation of  $A_U$ . Nevertheless, sometimes the (Hochschild-)cohomological properties of A and M are such that it is possible to find deformations, as we will see in the next chapters of this thesis.

## Chapter 2

## Symplectic reflection algebras

### 2.1 Plan of the chapter

In this chapter we recall the basics of the theory of symplectic reflection algebras. In Section 2.2 we give the general definition, and we describe the main properties of these algebras such as the PBW property. In Section 2.3 we consider more specifically symplectic reflection algebras associated to wreath product groups, that will be the object of interest of this thesis.

#### 2.2 Definition and properties

Let  $(V, \omega)$  be a symplectic vector space over  $\mathbb{C}$ , and let  $G \subset \operatorname{Sp}(V)$  be a finite group of symplectomorphisms. Denote by TV the tensor algebra of V and by  $\mathbb{C}[G]$  the group algebra. For any two vectors  $u, v \in V$  we will write uv for the tensor product  $u \otimes v$ .

**Definition 2.2.1.** The smash product algebra  $TV \ \# G$  is the vector space  $TV \otimes_{\mathbb{C}} \mathbb{C}[G]$ with the product defined by the formula

$$(u\otimes g)(v\otimes h)=u(gv)\otimes gh$$

for any  $u, v \in V$  and  $g, h \in G$ .

Note that assigning grade degree zero to the elements of  $\mathbb{C}[G]$  and grade degree one to the elements of V the algebra  $TV \sharp G$  becomes a graded algebra (with a corresponding filtration).

The main object of our interest will be a family of algebras obtained as quotients of the above smash product.

#### **Definition 2.2.2.** An element $s \in G$ is called a symplectic reflection if rk(Id-s) = 2.

Symplectic reflections can be considered as symplectic analogs of complex reflections. Indeed, any symplectic reflection s fixes a (complex) codimension two space pointwise and acts diagonally on a complement with complex conjugate eigenvalues of norm one. In other words there exists a basis such that s is a diagonal matrix

$$s = \left( \begin{array}{ccc} \lambda & & & \\ & \lambda^{-1} & & \\ & & 1 & \\ & & & \ddots & \\ & & & & 1 \end{array} \right)$$

with  $\lambda \neq 1$  and  $|\lambda| = 1$ . We will denote by S the set of symplectic reflections. By definition, this set is stable under conjugation. We will denote by C(S) the vector space  $\mathbb{C}[S]^G$  of  $\mathbb{C}$ -valued class functions on S, and we will write  $f_s = f(s)$  for any  $f \in C(S)$ .

For any  $s \in S$  consider the  $\omega$ -orthogonal decomposition  $\operatorname{Im}(\operatorname{Id} - s) \oplus \operatorname{Ker}(\operatorname{Id} - s)$ . Denote by  $\omega_s$  the skew symmetric form that coincides with  $\omega$  on  $\operatorname{Im}(\operatorname{Id} - s)$  and has  $\operatorname{Ker}(\operatorname{Id} - s)$  as its radical.

**Definition 2.2.3** ([EG02]). For any  $f \in C(S)$  and any constant  $t \in \mathbb{C}$  the symplectic reflection algebra  $H_{t,f}(G)$  is the quotient of the smash product  $TV \sharp G$  by the relations

$$uv - vu = t\,\omega(u, v) + \sum_{s \in \mathcal{S}} f_s\,\omega_s(u, v)\,s \tag{2.1}$$

for any  $u, v \in V$ .

Let now  $e = \frac{1}{|G|} \sum_{g \in G} g \in \mathbb{C}[G]$  be the averaging idempotent.

**Definition 2.2.4.** The spherical subalgebra is the algebra  $eH_{t,f}(G)e \subset H_{t,f}(G)$ .

Note that that  $eH_{t,f}(G)e$  does not contain the unit element of  $H_{t,f}(G)$ .

When both t and f are zero, one has uv - vu = 0 for any  $u, v \in V$ . Thus, if we denote by SV the symmetric algebra of V we have:

$$H_{0,0}(G) = SV \sharp G.$$

In particular the algebra  $H_{0,0}(G)$  is graded. Moreover, the isomorphism  $e(SV \sharp G)e \cong$  $SV^G$ , where  $SV^G$  is the algebra of invariant polynomials, yields an isomorphism

$$eH_{0,0}(G)e \cong SV^G$$

for the spherical subalgebra.

In general though, the defining relations (2.1) are not homogeneous and the algebra  $H_{t,f}(G)$  does not inherit the grading of  $TV \sharp G$  but only a filtration  $\mathcal{F}_{\bullet}$ . Consider the associated graded algebra

$$\operatorname{gr}(H_{t,f}(G)) = \bigoplus_{i} \mathcal{F}_{i}(H_{t,f}(G)) / \mathcal{F}_{i-1}(H_{t,f}(G)).$$

Since uv - vu lies in degree two for any  $u, v \in V$ , while any element of the group algebra lies in degree zero it is clear from the defining relations that u, v commute in  $gr(H_{t,f}(G))$ . Thus there exists a surjective homomorphism of graded algebras

$$\phi: H_{0,0}(G) = SV \sharp G \twoheadrightarrow \operatorname{gr}(H_{t,f}(G)).$$

One of the most important properties of the algebra  $H_{t,f}(G)$  is that the above homomorphism is also injective, as stated in the next theorem, called *Poincaré-Birkhoff-Witt(PBW)-Theorem* in analogy with the PBW-Theorem for the universal enveloping algebra of a Lie algebra. **Theorem 2.2.5.** ([EG02], Theorem 1.3) For any  $t \in \mathbb{C}$  and  $f \in C(S)$  the above homomorphism is an isomorphism.

It is clear that the PBW theorem also gives an isomorphism

$$\operatorname{gr}(eH_{t,f}(G)e) \cong SV^G.$$

It is not hard to see that rescaling the parameters t, f by a non-zero complex number does not change  $H_{t,f}(G)$  up to isomorphism. Thus, in particular, for any  $t \neq 0$ there is an isomorphism  $H_{t,f}(G) \cong H_{1,f/t}(G)$ . This reduces the study of symplectic reflection algebras to the two cases t = 0 (quasi-classical case) and t = 1 (quantum case) that present substantial differences. In this thesis we will be concerned with the second case.

Suppose then that t = 1. In this case, specializing the parameter f to 0 we get an isomorphism

$$H_{1,0}(G) \cong \mathcal{W} \sharp G$$

where  $\mathcal{W}$  is the Weyl algebra of the symplectic vector space  $(V, \omega)$  i.e.

$$\mathcal{W} := \frac{TV}{\langle uv - vu = \omega(u, v) \rangle_{u, v \in V}}.$$

Regard now f as a formal parameter and the family  $\{H_{1,f}(G)\}_f$  as a single algebra over  $\mathbb{C}[[C(\mathcal{S})]]$ . Thanks to the PBW property stated in Theorem 2.2.5,  $\{H_{1,f}(G)\}_f$ has no torsion as a  $\mathbb{C}[[C(\mathcal{S})]]$ -module. Thus the corresponding family is flat over  $\mathbb{C}[[C(\mathcal{S})]]$  and can be seen as a flat formal deformation of  $\mathcal{W} \sharp G$  (in this case the power series defining the star product is a degree one polynomial in f, thus it converges for any value of f and the deformation is not only formal). The following theorem is due to Etingof and Ginzburg.

**Theorem 2.2.6** ([EG02]). The family  $\{H_{1,f}(G)\}_f$  is the universal deformation of the algebra  $\mathcal{W} \sharp G$  and the family  $\{eH_{1,f}(G)e\}_f$  is the universal deformation of the algebra of invariants  $\mathcal{W}^G$ .

Proof. In [AFLS00] Alev, Farinati, Lambre and Solotar proved that the dimensions of the Hochschild cohomology groups of  $\mathcal{W} \ \ G$  are as follows

$$\dim H^{i}(\mathcal{W} \sharp G) = \begin{cases} \mathbf{n}(j) & \text{if } i = 2j \\ 0 & \text{if } j \text{ is odd} \end{cases}$$

where

$$\mathbf{n}(j) = \sharp \{ \text{conjugacy classes of elements } g \in G \text{ such that } \operatorname{rk}(\operatorname{Id} - g) = 2j \}.$$

In particular,  $H^3(\mathcal{W} \sharp G, \mathcal{W} \sharp G) = 0$  and dim  $H^2(\mathcal{W} \sharp G, \mathcal{W} \sharp G) = \dim C(\mathcal{S})$ . Thus Theorem 1.3.4 guarantees the existence of the universal deformation and the fact that it is parametrized by the space  $C(\mathcal{S})$ . The only thing left to do is proving that the family  $\{H_{1,f}(G)\}_f$  is actually the one satisfying the condition in Theorem 1.3.4. In other words, it must be verified that the cohomology classes corresponding to the family of infinitesimal (order one) deformations  $\{H_{1,f}(G)/\mathfrak{m}H_{1,f}(G)\}_f$  (here  $\mathfrak{m}$  is the maximal ideal in  $\mathbb{C}[[C(\mathcal{S})]]$ ) span the whole space  $C(\mathcal{S})$  and not a smaller one. This is done in [EG02]. The proof for  $\{H_{1,f}(G)/\mathfrak{m}H_{1,f}(G)\}_f$  is similar.

#### 2.3 The wreath-product construction

In this thesis we will be interested in the study of the symplectic reflection algebras attached to a special family of groups generated by symplectic reflections, provided by the *wreath product* construction.

Let  $\Gamma$  be a finite subgroup of  $SL(2, \mathbb{C})$ , and let  $\Gamma^n$  be the direct product  $\underbrace{\Gamma \times \cdots \times \Gamma}_{n \text{ factors}}$ . Let  $S_n$  be the symmetric group of rank n.

**Definition 2.3.1.** The wreath product  $\Gamma_n := S_n \ltimes \Gamma^n$  is the semi-direct product of  $S_n$ and  $\Gamma^n$ , where  $\Gamma^n$  is normal, and the action of  $S_n$  on  $\Gamma^n$  by conjugation is the natural one in which  $S_n$  permutes the direct factors of  $\Gamma^n$ . Let now L be a 2-dimensional complex vector space with a symplectic form  $\omega_L$ , and consider the space  $V = L^{\oplus n}$ , endowed with the induced symplectic form  $\omega_V = \omega_L^{\oplus n}$ .

Choosing a symplectic basis we can identify Sp(L) with  $SL(2, \mathbb{C})$ . Clearly, the natural (faithful) action of the wreath product group  $\Gamma_n$  on V, where each factor  $\Gamma$ in  $\Gamma^n$  acts on the corresponding summand L in V, and  $S_n$  permutes such summands, is symplectic. Thus  $\Gamma_n \subset Sp(V)$ .

In the sequel we will write  $\gamma_i \in \Gamma_n$  for any element  $\gamma \in \Gamma$  seen as an element in the *i*-th factor  $\Gamma$  of  $\Gamma^n$ .  $\Gamma_n$  acts by conjugation on the set S of its symplectic reflections. It is easy to see that there are symplectic reflections of two types in  $\Gamma_n$ :

(S) the elements  $s_{ij}\gamma_i\gamma_j^{-1}$  where  $i, j \in [1, n]$ ,  $s_{ij}$  is the transposition  $(ij) \in S_n$ , and  $\gamma \in \Gamma$ ;

( $\Gamma$ ) the elements  $\gamma_i$ , for  $i \in [1, n]$  and  $\gamma \in \Gamma \setminus \{1\}$ .

Elements of type (S) are all in the same conjugacy class, while elements of type  $(\Gamma)$  form one conjugacy class for any nontrivial conjugacy class in  $\Gamma$ . Thus functions  $f \in C(S)$  can be written as pairs (k, c), where k is a number (the value of f on elements of type (S)), and c is a conjugation invariant function on  $\Gamma \setminus \{1\}$  (encoding the values of f on the elements of type  $(\Gamma)$ ).

**Definition 2.3.2.** The wreath product symplectic reflection algebra  $H_{t,k,c}(\Gamma_n)$  is the symplectic reflection algebra attached to the vector space V, the group  $\Gamma_n$  and the parameters  $t \in \mathbb{C}$  and f = (k, c).

We will now give a more explicit presentation of the algebra  $H_{t,k,c}(\Gamma_n)$ . For any vector  $u \in L$  and any  $i \in [1, n]$  we will write  $u_i \in V$  for u placed in the *i*-th summand of V. In particular from now on we will fix a symplectic basis  $\{x, y\}$  of  $L(\omega_L(x, y) = 1)$ and we will denote by  $\{x_i, y_i\}$  the corresponding symplectic basis for V. We will also write  $c_{\gamma}$  for the value of the function c on the element  $\gamma \in \Gamma$ .

**Lemma 2.3.3.** ([GG], Lemma 3.1.1) The algebra  $H_{1,k,c}(\Gamma_n)$  is the quotient of  $TV \sharp \Gamma_n$ by the following relations: (R1) For any  $i \in [1, n]$ :

$$[x_i, y_i] = t + \frac{k}{2} \sum_{j \neq i} \sum_{\gamma \in \Gamma} s_{ij} \gamma_i \gamma_j^{-1} + \sum_{\gamma \in \Gamma \smallsetminus \{1\}} c_{\gamma} \gamma_i \,.$$

(R2) For any  $u, v \in L$  and  $i \neq j$ :

$$[u_i, v_j] = -\frac{k}{2} \sum_{\gamma \in \Gamma} \omega_L(\gamma u, v) s_{ij} \gamma_i \gamma_j^{-1}.$$

We will call the integer n the rank of the algebra  $H_{1,k,c}(\Gamma_n)$ .

We want now to have a closer look at some interesting examples.

**Example 2.3.4.** When the rank n is one, there is no parameter k (there are no symplectic reflections of type (S)). Let now  $\mathbf{c} = t + \sum_{\gamma \in \Gamma \setminus \{1\}} c_{\gamma} \gamma$  be the central element of  $\mathbb{C}[\Gamma]$  corresponding to the class function coinciding with c on  $\Gamma \setminus \{1\}$  and assuming value t on the identity element. If we identify the tensor algebra algebra TL with the ring  $\mathbb{C}\langle x, y \rangle$  of noncommutative polynomials in x, y, then the rank one wreath product symplectic reflection algebra is the quotient

$$H_{t,c}(\Gamma) := \frac{\mathbb{C}\langle x, y \rangle \sharp \Gamma}{\langle [x, y] - \mathbf{c} \rangle}$$

The algebra  $H_{t,c}(\Gamma)$  has interesting connections with the Kleinian singularity  $\mathbb{C}^2/\Gamma$ (the spherical subalgebra  $eH_{t,c}(\Gamma)e$  is a non-commutative deformation of the ring of invariants  $\mathbb{C}[x,y]^{\Gamma}$ ) and it was studied by Crawley-Boevey and Holland in [CBH98]. In particular, as we will see in the next chapter, a complete classification of the simple finite dimensional  $H_{t,c}(\Gamma)$  module is available for all values of the parameters.

**Example 2.3.5.** When k = 0 the defining relations (R1), (R2) simplify drastically and there is an isomorphism

$$H_{t,0,c}(\mathbf{\Gamma}_n) \cong H_{t,c}(\mathbf{\Gamma})^{\otimes n} \sharp S_n.$$
(2.2)

**Example 2.3.6.** Suppose  $\Gamma = \mathbb{Z}/m\mathbb{Z}$  is cyclic of some order m. In this case there is a splitting  $V = \mathfrak{h} \oplus \mathfrak{h}^*$ , where  $\mathfrak{h}$  is the reflection representation of  $S_n \ltimes (\mathbb{Z}/m\mathbb{Z})^n$ (as a complex reflection group) and  $\mathfrak{h}^*$  is its dual. The symplectic form  $\omega_V$  can be identified with the natural pairing between  $\mathfrak{h}$  and  $\mathfrak{h}^*$  that become Lagrangian subspaces. Thus the vector space V can be seen as the cotangent bundle of  $\mathfrak{h}$  endowed with its natural structure of symplectic manifold and with the diagonal (Hamiltonian) action of  $S_n \ltimes (\mathbb{Z}/m\mathbb{Z})^n$ . The algebra  $H_{t,k,c}(S_n \ltimes (\mathbb{Z}/m\mathbb{Z})^n)$  is a special case of rational Cherednik algebra. As a vector space, this algebra has a decomposition  $S\mathfrak{h} \otimes \mathbb{C}[\mathbf{\Gamma}_n] \otimes S\mathfrak{h}^*$ , analog to the triangular decomposition for the universal enveloping algebra of a semisimple Lie algebra.

As already mentioned, in this thesis we will be concerned with the representation theory of the wreath product algebra when t = 1. Although our results will be true whenever  $\Gamma$  is nontrivial, we want to recall that when  $\Gamma$  is cyclic, i.e. in the case of the rational Cherednik algebra, there exists a more general (and effective) approach to the study of representations. For rational Cherednik algebras, in fact, an analog of category  $\mathcal{O}$  for finite dimensional semisimple Lie algebras has been defined, as well as a theory of standard modules and formal characters ([BEG03], [GGOR03]).

For completeness, we want to end this section with a few words about the case t = 0. Suppose  $c \neq 0$ , the fact that makes the quasi-classical case notably different is that the algebra  $H_{0,k,c}(\Gamma_n)$  has a large center  $Z_{0,k,c}$  such that

$$\operatorname{gr}(Z_{0,k,c}) = SV^{\Gamma_n}$$

and

$$Z_{0,k,c} \cong eH_{0,k,c}(\mathbf{\Gamma}_n)e.$$

The following theorem is due to Etingof and Ginzburg.

**Theorem 2.3.7.** ([EG02], Corollary 1.14) If the parameters (k, c) are generic, all irreducible  $H_{0,k,c}(\Gamma_n)$ -modules are finite dimensional of dimension  $|\Gamma_n| = n! |\Gamma|^n$ , and are isomorphic to the regular representation of  $\Gamma_n$  as  $\Gamma_n$ -modules. Moreover,
Spec  $Z_{0,k,c}$  is a smooth algebraic variety, and irreducible modules are parametrized up to isomorphism by the points of Spec  $Z_{0,k,c}$  via the map that assigns to each module its central character.

# Chapter 3

# Finite dimensional representations for $H_{1,k,c}(\Gamma_n)$

# 3.1 Plan of the chapter

In this chapter we will present two different methods to produce examples of finite dimensional representations for the algebra  $H_{1,k,c}(\Gamma_n)$ . Both methods start from simple observations.

In the first place from Definition 2.2.3 we can see that the algebra  $H_{1,k,c}(\Gamma_n)$ contains a copy of the group algebra  $\mathbb{C}[\Gamma_n]$ . Thus, the simplest thing to do is trying to classify all irreducible  $\Gamma_n$ -modules that extend to representations of the whole algebra  $H_{1,k,c}(\Gamma_n)$ . We give a complete answer to this problem in Section 3.2, which is based on the paper [Mon07b].

Secondly, in the rank one case a complete classification of the finite dimensional representations for the wreath product algebra  $H_{1,c}(\Gamma)$  is available, thanks to the results of Crawley-Boevey and Holland ([CBH98]). Moreover, as observed in Example 2.3.5

$$H_{1,0,c}(\mathbf{\Gamma}_n) = H_{1,c}(\mathbf{\Gamma})^{\otimes n} \sharp S_n$$

i.e. when the parameter k is zero, the rank n algebra is simply the smash product of the tensor product of n copies of the rank one algebra with  $S_n$  (where  $S_n$  acts by permuting the factors). As a consequence, in this case, finite dimensional irreducible representations are also known (they can be recovered from a knowledge of the irreducible finite dimensional representations in rank one and from well known results by Macdonald about skew group algebras as explained in Section 3.3.4). We observe now that that the algebra  $H_{1,0,c}(\Gamma_n)$  has a flat formal deformation over the finite dimensional vector space C(S) given by  $H_{1,k,c+c'}$ . The fact that this deformation is flat follows from the PBW Theorem 2.2.5. Using this observation in Section 3.3 we determine some sufficient conditions for a  $H_{1,0,c}(\Gamma_n)$ -module to be deformed to values of the parameters with nonzero k. This last section is based on the papers [EM05] and [Mon07a].

We recall that in the case of cyclic (nontrivial)  $\Gamma$  some finite dimensional representation were constructed by Chmutova and Etingof in [CE03] before our work.

# 3.2 Extending irreducible $\Gamma_n$ -modules

## 3.2.1 Irreducible representations of wreath product groups

For the reader's convenience, and in order to introduce some important notation, we recall the classification of irreducible representations for a wreath product group. Everything that follows is true for any finite group  $\Gamma$  and for representations over any algebraically closed field F of characteristic 0. For simplicity we will consider  $F = \mathbb{C}$ , the field of complex numbers. For complete proofs and details the reader should refer to [JK81], Chapter 4.

A nice property of the wreath product group  $\Gamma_n$  is that the set of its irreducible representations  $\operatorname{Irr}(\Gamma_n)$  can be completely recovered from a knowledge of  $\operatorname{Irr}(\Gamma)$ , using the representation theory of the symmetric group.

Let  $\{N_1, \ldots, N_\nu\}$  denote a complete set of pairwise non-isomorphic representations of  $\Gamma$  over  $\mathbb{C}$ . Then a complete set of irreducible representations of  $\Gamma^n$  is given by  $N = N_{h_1} \otimes \cdots \otimes N_{h_n}$  where  $(h_1, \ldots, h_n)$  varies in  $[1, \nu]^n$ . If  $n_h$  denotes the number of indices i s.t.  $h_i = h$ , i.e. the number of factors of N equal to  $N_h$ , then

$$\vec{n} = (n_1, \ldots, n_\nu)$$

is called the type of N.

We will say that two representations N, N' are *conjugate* if they have the same type. This simply means that  $N = N_{h_1} \otimes \cdots \otimes N_{h_n}$  and  $N' = N_{h_{\sigma(1)}} \otimes \cdots \otimes N_{h_{\sigma(n)}}$  for some  $\sigma \in S_n$ , i.e. N' equals the representation N twisted by the outer automorphism of  $\Gamma^n$  that permutes the factors according to  $\sigma$ . It turns out that the role played by conjugate representations of  $\Gamma^n$  in recovering irreducible representations of  $\Gamma_n$  is exactly the same. This is essentially because, as one can easily argue from Definition 2.3.1, the outer automorphism induced by  $\sigma \in S_n$  on  $\Gamma^n$  is a restriction of an inner automorphism in  $\Gamma_n$  (conjugation by the element  $\sigma \in \Gamma_n$ ). So from now on we will consider only the representations of  $\Gamma^n$  that can be written as  $N = N_1^{\otimes n_1} \otimes \cdots \otimes N_{\nu}^{\otimes n_{\nu}}$ . Notice that the representations of  $\Gamma^n$ .

For any h, we denote by  $S_{n_h}$  the subgroup of  $S_n$  consisting of the permutations that move only the indices  $\{\sum_{i=1}^{h-1} n_i + 1, \dots, \sum_{i=1}^{h} n_i\}$ , corresponding to the factors of N isomorphic to  $N_h$ . We agree that  $S_{n_h} = \{1\}$  if  $n_h = 0$ . Thus we can consider the group

$$S_{\vec{n}} = S_{n_1} \times \dots \times S_{n_{\nu}} \subset S_n \subset S_n \ltimes \Gamma^n$$

called the *inertia factor* of N. Obviously any irreducible representation X of  $S_{\vec{n}}$  is obtained as  $X = X_1 \otimes \cdots \otimes X_{\nu}$ , where  $X_h$  is an irreducible representation of  $S_{n_h}$ .

The *inertia subgroup* of N, instead, is defined to be

$$(\mathbf{\Gamma}_n)_N = S_{\vec{n}} \ltimes \Gamma^n \subset S_n \ltimes \Gamma^n.$$

Let's now consider an irreducible  $\Gamma^n$ -module  $N = N_1^{\otimes n_1} \otimes \cdots \otimes N_{\nu}^{\otimes n_{\nu}}$ . There is a natural action of  $(\Gamma_n)_N$  on N in which  $\Gamma^n$  acts in the obvious way, and  $S_{\vec{n}}$  permutes the factors. This representation can be shown to be irreducible. For simplicity we will keep the notation N for this representation.

Another easy way to obtain irreducible representations of  $(\Gamma_n)_N$  is extending an irreducible representation  $X = X_1 \otimes \cdots \otimes X_\nu$  of  $S_{\vec{n}}$  by making  $\Gamma^n$  act trivially. In this case we will also keep the notation X for this extension.

Let's now consider the tensor product of X and N

$$X \otimes N = (X_1 \otimes \cdots \otimes X_{\nu}) \otimes (N_1^{\otimes n_1} \otimes \cdots \otimes N_{\nu}^{\otimes n_{\nu}}).$$

Here  $S_{\vec{n}}$  acts both on X and on N( permuting the factors), while  $\Gamma^n$  acts only on N. This is also an irreducible representation of  $(\Gamma_n)_N$  ([JK81], page 155).

We can now obtain the induced representation of  $\Gamma_n$ :

$$X \otimes N \uparrow := \operatorname{Ind}_{(\Gamma_n)_N}^{\Gamma_n} X \otimes N$$

The following theorem holds.

**Theorem 3.2.1.** The representation  $X \otimes N \uparrow$  is irreducible and runs through a complete system of pairwise non-isomorphic irreducible representations of  $\Gamma_n$  if N runs through a complete system of pairwise non-conjugate irreducible representations of  $\Gamma^n$  and, while N remains fixed, X runs through a complete system of pairwise non-isomorphic irreducible representations of  $S_{\vec{n}}$ .

In particular we have that, for a fixed X, the representation  $X \otimes N \uparrow$  depends only on the type of N. With abuse of language we will call *type* of  $X \otimes N \uparrow$  the type of N as a representation of  $\Gamma^n$ . We remark that the possible types of  $X \otimes N \uparrow$ are in bijection with the  $\nu$ -tuples  $(n_1, \ldots, n_{\nu}), n_h \ge 0, \sum_h n_h = n$  and that to any such  $\nu$ -tuple we can attach a proper partition of n, taking all the non-zero  $n_h$ s in  $(n_1, \ldots, n_{\nu})$  and ordering them in non-increasing order.

**Example 3.2.2.** Suppose all the factors of N are the same, i.e.  $N = N_h^{\otimes n}$  for some  $h \in \{1, ..., \nu\}$ . The type of N is (0, ..., 0, n, 0, ..., 0) with n in the h-th position and is associated to the partition of n of Young diagram a single row of length n, i.e. the partition corresponding to the trivial representation of  $S_n$ . For this reason we will

call these representations of "trivial type". In this case the inertia factor of N is  $S_n$ , its inertia subgroup coincides with  $S_n \ltimes \Gamma^n$ , and we need no induction. For any irreducible representation X of  $S_n$  we obtain the irreducible representation  $X \otimes N$  of  $S_n \ltimes \Gamma^n$ .

## **3.2.2** McKay correspondence

It is well known (see for example [Cox91], Chapters 6,7) that all the finite subgroups of  $SL(2, \mathbb{C})$ , or equivalently the finite groups of quaternions, can be distinguished into two infinite series

- the cyclic groups  $C_{m+1}$  for any  $m \ge 0$  ( $C_1 = \{1\}$ ), of order m + 1;
- the dicyclic groups  $\mathcal{D}_{m-2}$  for  $n \ge 4$ , of order 4(m-2);

and three exceptional groups that are the double coverings of the groups of rotations preserving regular polyhedra in  $\mathbb{R}^3$  via the homomorphism of Lie groups  $SU(2) \longrightarrow$  $SO(3,\mathbb{R})$ :

- the binary tetrahedral group  $\mathfrak{T}$ , of order 24;
- the binary octahedral group  $\mathfrak{O}$ , of order 48;
- the binary icosahedral group  $\mathfrak{I}$ , of order 120.

The terminology we used refers to the so called McKay correspondence, as we are going to explain. In ([McK81]) McKay showed that "the eigenvectors of the Cartan matrices of affine type  $\tilde{A}_m$ ,  $\tilde{D}_m$ ,  $\tilde{E}_6$ ,  $\tilde{E}_7$ ,  $\tilde{E}_8$  can be taken to be the columns of the character tables of the finite groups of quaternions". To this end he attached a graph to any finite subgroup of  $SL(2, \mathbb{C})$  in the following way. Consider the set of irreducible non-isomorphic representations of a finite group  $\Gamma \subset SL(2, \mathbb{C})$ ,  $I = \{N_1, \ldots, N_\nu\}$ , and let L be the defining representation of  $\Gamma$ , i.e. the representation of  $\Gamma$  as a subgroup of  $SL(2, \mathbb{C})$ . Notice that L is a self-dual representation. Now build the graph in which the set of vertices is I, and the number of edges between two vertices  $N_h$  and  $N_{h'}$ is the multiplicity of the irreducible representation  $N_h$  in  $N_{h'} \otimes L$  or equivalently, since L is self-dual, the multiplicity of  $N_{h'}$  in  $N_h \otimes L$ . Any such graph turns out to be an extended Dynkin graph with extending vertex corresponding to the trivial representation. If we label each vertex with the dimension of the corresponding representation the result is the following. When  $\Gamma = C_{n+1}$ ,  $n \ge 0$  is cyclic we get the extended Dynkin diagram  $\tilde{A}_n$ :



Figure 3-1: Graph  $A_n$ 

When  $\Gamma = \mathcal{D}_{n-2}, n \ge 4$  is dicyclic we get the diagram  $\tilde{D}_n$ :



Figure 3-2: Graph  $\tilde{D}_n$ 

When  $\Gamma = \mathfrak{T}$  (binary tetrahedral),  $\Gamma = \mathfrak{O}$  (binary octahedral) or  $\Gamma = \mathfrak{I}$  (binary icosahedral) we get the extended Dynkin diagrams of type  $\tilde{E}_6$ ,  $\tilde{E}_7$ ,  $\tilde{E}_8$  respectively:



Figure 3-3: Graph  $\tilde{E}_6$ 



In this setting, the adjacent vertices to a fixed vertex  $N_h$  correspond to the types of the irreducible components of the representation  $N_h \otimes L$ , while the number of connecting vertices corresponds to the multiplicities of such types. Note that the decomposition of  $N_h \otimes L$  is multiplicity free (i.e. the diagram is simply laced) except when  $\Gamma = C_2$  (i.e. for type  $\tilde{A}_1$ ). Thus, when  $\Gamma \neq C_2$ , if for any  $i = 1, \ldots, \nu$  we set  $d_i = \dim N_i$ , and we consider the vector  $\delta = \{d_i\} \in \mathbb{Z}^{\nu}$  (corresponding to the above labeling), we get the "harmonic" property

$$2d_i = \sum_{j \text{adjacent to } i} d_j$$

for the extended Dynkin diagrams above.

# **3.2.3** Representations of $S_n$ with rectangular Young diagram

In what follows we will use the following standard results from representation theory of the symmetric group. Denote by  $\mathfrak{h}$  the reflection representation of  $S_n$ . For a Young diagram  $\mu$  we denote by  $X_{\mu}$  the corresponding irreducible representation of  $S_n$  and by  $C(\mu)$  the content of  $\mu$ , i.e. the sum of signed distances of the cells from the diagonal.

**Lemma 3.2.3.** i)  $\operatorname{Hom}_{S_n}(\mathfrak{h} \otimes X_\mu, X_\mu) = \mathbb{C}^{m-1}$ , where *m* is the number of corners of the Young diagram  $\mu$ . In particular  $\operatorname{Hom}_{S_n}(\mathfrak{h} \otimes X_\mu, X_\mu) = 0$  if and only if  $\mu$ is a rectangle.

- ii) The element  $C = s_{12} + s_{13} + \dots + s_{1n}$  acts by a scalar in  $X_{\mu}$  if and only if  $\mu$  is a rectangle. In this case  $C|_{X_{\mu}} = \frac{2C(\mu)}{n}$ .
- iii) If  $\mu$  is a rectangular Young diagram of height a and width b, then  $C(\mu) = \frac{(b-a)n}{2}$ .

Proof. Let  $S_{n-1} \subset S_n$  be the subgroup of permutations fixing the index 1. It is well known that  $X_{\mu}|_{S_{n-1}} = \sum X_{\mu-j}$ , where the sum is taken over the corners of  $\mu$ and  $\mu - j$  is the Young diagram obtained from  $\mu$  by cutting off the corner j. Since  $\mathfrak{h} \oplus \mathbb{C} = \operatorname{Ind}_{S_{n-1}}^{S_n} \mathbb{C}$ , the assertion (i) follows from the Frobenius reciprocity. To prove (ii), observe that C commutes with  $S_{n-1}$ , so acts by a scalar on each  $X_{\mu-j}$ . Thus, if  $\mu$  is a rectangle, C acts as a scalar (as we have only one summand), and the "if" part of the statement is proved. To prove the "only if" part, let  $Z_n$  be the sum of all transpositions in  $S_n$ .  $Z_n$  is a central element in the group algebra, and it is known to act in  $X_{\mu}$  by the scalar  $\mathbf{c}(\mu)$ , where  $\mathbf{c}(\mu)$  is the content of  $\mu$ , i.e. the sum over all cells of the signed distances from these cells to the diagonal. Now,  $C = Z_n - Z_{n-1}$ , so it acts on  $X_{\mu-j}$  by the scalar  $\mathbf{c}(j)$ , the signed distance from the cell j to the diagonal. The numbers  $\mathbf{c}(j)$  are clearly different for all corners j, so if there are 2 or more corners, then C cannot act by a scalar. This finishes the proof of (ii). Part (iii) is a straightforward computation.

### 3.2.4 The main theorem

Our main theorem classifies the irreducible representations of  $\Gamma_n$  that extend to representations of  $H_{1,k,c}(\Gamma_n)$  for values of (k,c) with  $k \neq 0$ . For  $\Gamma = \{1\}$  it is easy to see that the algebra  $H_{1,k,c}(S_n)$  has no finite dimensional representations. In fact  $H_{1,k,c}(S_n)$  always contains a copy of the Weyl algebra (generated by the elements  $x_1 + \cdots + x_n, y_1 + \cdots + y_n$ ) that has no finite dimensional representations. We will thus consider the case  $\Gamma \neq \{1\}$ . Before stating the theorem we need to introduce some notation:

•  $\nu$  will denote the number of conjugacy classes  $\{C_1, \ldots, C_\nu\}$  of  $\Gamma$ , with  $C_1 = \{1\}$ ,

 $|C_s|$  will be the cardinality of the class  $C_s$ , and  $c_s$  the value of the class function c on  $C_s$ ;

• for any irreducible representation  $N_h$  of  $\Gamma$ ,  $\chi_{N_h}(C_s)$  will be the value of the character of  $N_h$  on the class  $C_s$ .

With this notation, the complex number  $\frac{|C_s|\chi_{N_h}(C_s)}{\dim N_h}$  is the scalar corresponding to the central element  $\sum_{\gamma \in C_s} \gamma$  in the irreducible representation  $N_h$ .

**Theorem 3.2.4.** Let  $\Gamma \neq \{1\}$ . Then:

- I) If an irreducible  $\Gamma_n$ -module M extends to a representation of  $H_{1,k,c}(\Gamma_n)$  then the generators  $x_i$ ,  $y_i$  act by zero on M for any i = 1, ..., n.
- II) For  $k \neq 0$  an irreducible representation  $M = X \otimes N \uparrow of \Gamma_n$  of type  $(n_1, \ldots, n_\nu)$ extends to a representation of some associated symplectic reflection algebra  $H_{1,k,c}(\Gamma_n)$  if and only if the following two conditions are satisfied:
  - i)  $X = X_1 \otimes \cdots \otimes X_{\nu}$ , where  $X_h$  is an irreducible representation of  $S_{n_h}$  with rectangular Young diagram of some size  $a_h \times b_h$ , for any h s.t.  $n_h \neq 0$ ;
  - ii) for any  $h \neq h'$  s.t.  $n_h, n_{h'} \neq 0$ ,  $\operatorname{Hom}_{\Gamma}(N_h \otimes L, N_{h'}) = 0$ , where L is the natural representation of  $\Gamma$ . In other words, any two non-isomorphic representations  $N_h$ ,  $N_{h'}$  of  $\Gamma$  occurring in the type of N must be nonadjacent vertices in the extended Dynkin diagram attached to  $\Gamma$ . We agree that this condition is empty when N is of trivial type (Example 3.2.2).
- III) The values of the parameter (k, c) for which  $M = X \otimes N \uparrow$  can be extended form a linear subspace of  $\mathbb{C}^{\nu}$ , which can be described as the intersection of the hyperplanes

$$\mathcal{H}_h: \qquad \dim N_h + (b_h - a_h) \frac{k}{2} |\Gamma| + \sum_{s=2}^{\nu} c_s |C_s| \chi_{N_h}(C_s) = 0 \qquad (3.1)$$

for all  $h \in \{1, ..., \nu\}$  s.t.  $n_h \neq 0$ , i.e. for any representation  $N_h$  occurring in the type of N. The space of the solutions of this system of equations has dimension  $\nu - r$  where  $r = \#\{h \text{ s.t. } n_h \neq 0\}$ .

# 3.2.5 Proof of Theorem 3.2.4

From now on we will assume  $\Gamma \neq \{1\}$ . We will divide the proof of Theorem 3.2.4 in several steps.

#### <u>STEP 1</u> Proof of Theorem 3.2.4 part I)

Without loss of generality consider the elements  $x_1, y_1 \in H_{1,k,c}(\Gamma_n)$ . From Section 2.3 we know these elements commute with the elements  $\gamma_i$  for  $i \neq 1$ , and the action of  $\gamma_1$  by conjugation on such elements corresponds to the action of  $\gamma$  on the basis vectors x, y respectively in the natural representation L of  $\Gamma$ . Thus we can view  $x_1$ ,  $y_1$  as a basis for the representation:

$$L\otimes \underbrace{\mathbb{C}\otimes \cdots \otimes \mathbb{C}}_{n-1}$$

of  $\Gamma^n$ , where  $\mathbb{C}$  is the trivial one-dimensional representation. So we have that the action of  $x_1, y_1$  on M induces maps of  $\Gamma^n$ -modules:

$$(L \otimes \mathbb{C} \otimes \cdots \otimes \mathbb{C}) \otimes M \longrightarrow M.$$

But now from Section 3.2.1 we have that, as a  $\Gamma^n$ -module, M decomposes in irreducibles as

$$\bigoplus_{\sigma} N_{h_{\sigma(1)}} \otimes \cdots \otimes N_{h_{\sigma(n)}}$$

where  $\sigma$  are permutations in  $S_n$  and factors may appear with some multiplicity. Thus composing with the  $\Gamma^n$ -module maps given by the injections and projections of the direct factors we have that  $x_1, y_1$  induce  $\Gamma^n$ -module maps:

$$\left(L\otimes N_{h_{\sigma(1)}}\right)\otimes\cdots\otimes N_{h_{\sigma(n)}}\longrightarrow N_{h_{\sigma'(1)}}\otimes\cdots\otimes N_{h_{\sigma'(n)}}$$

for any  $\sigma$ ,  $\sigma'$ . Since the  $N_h$ s are irreducible  $\Gamma$ -modules, in order for such a map to be non-zero, we must have  $N_{h_{\sigma(i)}} \cong N_{h_{\sigma'(i)}}$  for any  $i \ge 2$ . This implies  $N_{h_{\sigma(1)}} \cong N_{h_{\sigma'(1)}}$  and we get a homomorphism:

$$L \otimes N_{h_{\sigma(i)}} \longrightarrow N_{h_{\sigma(i)}}.$$

But such a homomorphism must be zero as explained in Section 3.2.2, as all extended Dynkin diagrams except  $\tilde{A}_0$  have no loop-vertices. We deduce  $x_1, y_1$  act trivially on M.

Now that we know that the generators  $x_i$ ,  $y_i$  must act trivially on M we can reduce the defining relations (*R1*), (*R2*) of  $H_{1,k,c}(\Gamma_n)$  in Lemma 2.3.3 to the simpler form:

(R1') For any  $i \in [1, n]$ :

$$0 = 1 + \frac{k}{2} \sum_{j \neq i} \sum_{\gamma \in \Gamma} s_{ij} \gamma_i \gamma_j^{-1} + \sum_{\gamma \in \Gamma \smallsetminus \{1\}} c_{\gamma} \gamma_i \, .$$

(R2') For any  $u, v \in L$  and  $i \neq j$ :

$$0 = \frac{k}{2} \sum_{\gamma \in \Gamma} \omega_L(\gamma u, v) \gamma_i \gamma_j^{-1},$$

where, with abuse of notation, we wrote  $\gamma_i$ ,  $s_{ij}$  etc... for the images of the corresponding elements of  $H_{1,k,c}(\Gamma_n)$  in the representation M.

This reduction will allow us to prove part II) and III) of Theorem 3.2.4 by using simple classical results from the representation theory of finite groups.

#### <u>STEP 2</u> The relations (R2')

It turns out that the relations (R2') have an easy interpretation in terms of the extended Dynkin diagram attached to the group  $\Gamma$  in the McKay correspondence. Let L be the natural representation of  $\Gamma$ . We have the following proposition. **Lemma 3.2.5.** If  $X \otimes Y \uparrow$  is a representation of  $\Gamma_n$  of type  $(n_1, \ldots, n_\nu)$ , then the operators of the corresponding matrix representation satisfy (R2') for  $k \neq 0$  if and only if for any pair h, h' s.t.  $n_h, n_{h'} \neq 0$ ,  $\operatorname{Hom}_{\Gamma}(L \otimes N_h, N_{h'}) = 0$  i.e. if and only if  $N_h, N_{h'}$  are not adjacent vertices of the extended Dynkin diagram associated to  $\Gamma$  in the McKay correspondence.

*Proof.* Relations (R2) are satisfied for  $k \neq 0$  if and only if

$$\sum_{\gamma \in \Gamma} \omega_L(\gamma u, v) \, \gamma_i \gamma_j^{-1} = 0 \qquad \forall \, u, v \in L, \quad \forall \, i \neq j \,.$$

We observe that, for any N, the subgroup  $\Gamma^n$  is contained in the inertia subgroup of N and is normal in  $\Gamma_n$ . For this reason the induced representation  $X \otimes N \uparrow$  can be written as:

$$\sigma_1 \cdot (X \otimes N) \oplus \dots \oplus \sigma_\ell \cdot (X \otimes N) \tag{3.2}$$

where  $\ell = \frac{n!}{n_1!\dots n_{\nu}!}$ , and  $\{\sigma_1, \dots, \sigma_\ell\}$  is a set of representatives of the left cosets of the inertia factor  $(\Gamma_n)_N$  in  $\Gamma_n$ , that can be chosen to be all in  $S_n$ . The action of an element  $g \in \Gamma_n$  on a vector  $\sigma_l \cdot v$  is defined as follows:

$$g(\sigma_l \cdot v) = \sigma_r \cdot (g'v)$$
 where  $g\sigma_l = \sigma_r g'$   $g' \in (\Gamma_n)_N$ .

By the normality of  $\Gamma^n$ , all the direct factors of (3.2) are stable under the action of  $\sum_{\gamma \in \Gamma} \omega_L(\gamma u, v) \gamma_i \gamma_j^{-1}$ , thus this operator has a block diagonal form. The *l*-th block corresponds to the operator  $A(s,t) = \sum_{\gamma \in \Gamma} \omega_L(\gamma u, v) \gamma_s \gamma_t^{-1}$ , with  $(s,t) = (\sigma_l^{-1}(i), \sigma_l^{-1}(j))$ , in the representation  $X \otimes N$  of  $(\Gamma_n)_N$ . We are reduced now to show that any such block is zero if and only if the conditions of the proposition are satisfied. Since the action of A(s,t) is trivial on X, we can suppose X to be trivial 1-dimensional, thus  $X \otimes N \cong N$ . Without loss of generality we can also suppose  $s \leq t$ . Since the bilinear form  $\omega_L$  is non degenerate and u, v vary in all L we have:

$$A(s,t) = 0 \Leftrightarrow \sum_{\gamma \in \Gamma} \gamma \otimes \gamma \otimes \gamma^{-1}|_{L \otimes N_{h_s} \otimes N_{h_t}} = 0 \Leftrightarrow \sum_{\gamma \in \Gamma} \gamma \otimes \gamma \otimes \gamma^{-1^*}|_{L \otimes N_{h_s} \otimes N_{h_t}} = 0$$

where "\*" denotes the transposition. Now, if we denote by  $N_{h_t}^*$  the dual representation of  $N_{h_t}$ , we notice that the last operator corresponds to the operator:

$$\sum_{\gamma \in \Gamma} \gamma \otimes \gamma \otimes \gamma |_{L \otimes N_{h_s} \otimes N_{h_t}^*},$$

that is a multiple of the projector on the invariants of the representation  $L \otimes N_{h_s} \otimes N_{h_t}^*$ .

Now this is zero if and only if  $\operatorname{Hom}_{\Gamma}(L \otimes N_{h_t}, N_{h_s}) = 0$ . From Section 3.2.2 we know this happens exactly when  $N_{h_s}$ ,  $N_{h_t}$  are non adjacent vertices in the Dynkin diagram attached to  $\Gamma$ .

Notice that when N is of trivial type  $(0, \ldots, n, \ldots 0)$ , i.e. when all the factors of N are the same, Lemma 3.2.5 implies that the conditions (R2') are automatically satisfied (since extended Dynkin diagrams corresponding to non-trivial finite subgroups of  $SL(2, \mathbb{C})$  have no loop vertices).

#### <u>STEP 3</u> The relations (R1')

The only thing we are left to do now is analyzing the conditions for relations (R1') to be satisfied. We will begin from the easiest case of  $M = X \otimes N \uparrow$  with N of type  $(0, \ldots, n, \ldots, 0)$ . We have the following proposition.

**Lemma 3.2.6.** For  $k \neq 0$  a representation M of  $\Gamma_n$  of trivial type  $(0, \ldots, n, \ldots, 0)$ extends to a representation of  $H_{1,k,c}(\Gamma_n)$  if and only if the following conditions are satisfied:

- i) the representation X of  $S_n$  corresponds to a rectangular Young diagram;
- ii) the parameter (k, c) satisfies the corresponding equation in part III) of Theorem 3.2.4.

Proof. As we observed in the previous subsection, Lemma 3.2.5 implies that in this case relations (R2') are satisfied. Thus we only have to consider relations (R1').

We have  $M = X \otimes N_h^{\otimes n}$ , with X and  $N_h$  irreducible modules for  $S_n$  and  $\Gamma$  respectively. We will begin with an easy clarifying example. When  $N_h$  is one dimensional, it is straightforward to check that Lemma 3.2.6 holds. In this case in fact, the permutation action of  $S_n$  on  $N_h^{\otimes n}$  is trivial and  $X \otimes N_h^{\otimes n} \cong X$  as  $S_n$ -modules. Thus the relation (*R1'*) for a fixed *i* looks like:

$$\frac{k}{2} |\Gamma| \sum_{j \neq i} s_{ij} = -1 - \sum_{\gamma \in \Gamma \setminus \{1\}} c_{\gamma} \chi_{N_h}(\gamma) = -1 - \sum_{s=2}^{\nu} c_s |C_s| \chi_{N_h}(C_s),$$

where  $\chi_{N_h}(C_s)$  is the value of the character of  $N_h$  on the conjugacy class  $C_s$ . For  $k \neq 0$  we have:

$$\sum_{j \neq i} s_{ij} = \frac{2\left(-1 - \sum_{s=2}^{\nu} c_s \left| C_s \right| \chi_{N_h}(C_s) \right)}{k \left| \Gamma \right|} \,. \tag{3.3}$$

So  $\sum_{j\neq i} s_{ij}$  must act as a scalar and Lemma 3.2.3 part ii) implies that X must have rectangular Young diagram  $\mu$  of some size  $a \times b$ . We remark that Lemma 3.2.3 part ii) iii) implies that the element  $\sum_{j\neq i} s_{ij}$  acts as the scalar  $\frac{2C(\mu)}{n} = (b-a)$  in this representation. Substituting this value in equation (3.3) we get the result in the 1dimensional case. Notice that this first consideration solves completely the case when  $\Gamma$  is cyclic.

Let's now suppose dim X = m and dim  $N_h = p > 1$ . We rewrite relations (R1') as follows:

$$-1 - \sum_{\gamma \in \Gamma \setminus \{1\}} c_{\gamma} \gamma_i = \frac{k}{2} \sum_{j \neq i} \sum_{\gamma \in \Gamma} s_{ij} \gamma_i \gamma_j^{-1}.$$
(3.4)

We observe that the left hand side of (3.4) is a central element of the group algebra  $\mathbb{C}[\Gamma^n]$  (due to the fact that c is a class function), and that, as a  $\Gamma^n$ -module,  $X \otimes N$  is isomorphic to a direct sum of dim X copies of the irreducible representation  $N_h^{\otimes n}$ . Thus the left hand side acts as a scalar in this representation. More precisely we have:

$$\sum_{\gamma \in \Gamma \smallsetminus \{1\}} c_{\gamma} \gamma_i = \sum_{s=2}^{\nu} \frac{c_s |C_s| \chi_{Y_h}(C_s)}{\dim N_h} \operatorname{Id}_{X \otimes N}.$$

So we must have that  $\frac{k}{2} \sum_{j \neq i} \sum_{\gamma \in \Gamma} s_{ij} \gamma_i \gamma_j^{-1}$  is a scalar. We will show that this operator has a block form that reduces equation (3.4) to equation (3.3). For this let's take any two bases  $\{v_1, \ldots, v_p\}$ ,  $\{x_1, \ldots, x_m\}$  for  $N_h$  and X respectively. The vectors  $\{\mathbf{v}_{\underline{i}} = v_{i_1} \otimes \cdots \otimes v_{i_n}\}$ , where the multi-index  $\underline{i} = (i_1, \ldots, i_n)$  varies in  $[1, p]^n$ , are clearly a basis for  $N = N_h^{\otimes n}$ . We can give the multi-indices I a total ordering  $\underline{i}_1, \ldots, \underline{i}_{p^n}$  using the lexicographic order. Consider now the basis of  $X \otimes N_h^{\otimes n}$  given by the vectors:

$$Z_1 = x_1 \otimes \mathbf{v}_{\underline{i}_1}, \ldots, Z_m = x_m \otimes \mathbf{v}_{\underline{i}_1}, \ldots, Z_{m(p^n-1)+1} = x_1 \otimes \mathbf{v}_{\underline{i}_{p^n}}, \ldots, Z_{mp^n} = x_m \otimes \mathbf{v}_{\underline{i}_{p^n}}.$$

Any transposition  $s_{ij} \in S_n$  induces a permutation  $\tilde{s}_{ij}$  (of order 2) on the set  $\{\underline{i}_1, \ldots, \underline{i}_{p^n}\}$ thus on the vectors of the basis  $\{Z_1, \ldots, Z_{mp^n}\}$ . Let's now denote by  $A_X(s_{ij})$  the operator (of size  $m \times m$ ) for  $s_{ij}$  in the representation X, and by  $O_m$  the 0-operator of size  $m \times m$ . It is easy to see that, using the basis  $\{Z_1, \ldots, Z_{mp^n}\}$ , we can obtain a block form for the operator  $s_{ij}$  in the representation  $X \otimes N$  from the block diagonal operator

$$\left(\begin{array}{ccccc} A_X(s_{ij}) & O_m & \cdots & O_m \\ O_m & \ddots & \cdots & O_m \\ \vdots & \vdots & \ddots & \vdots \\ O_m & \cdots & O_m & A_X(s_{ij}) \end{array}\right)$$

by simply permuting the columns according to  $\tilde{s}_{ij}$ . Using this, we can compute a block form for  $s_{ij} \sum_{\gamma \in \Gamma} \gamma_i \gamma_j^{-1}$ . We denote each block, of size  $m \times m$ , by its position  $(\underline{r}, \underline{t})$ , where  $\underline{r} = (r_1, \ldots, r_n)$ ,  $\underline{t} = (t_1, \ldots, t_n)$  are multi-indices. We have the following formulas for the blocks :

• for  $(\underline{r}, \underline{t})$  with  $\underline{r}$  differing from  $\underline{t}$  at most for the pair of indices  $(r_i, r_j)$ :

$$(\underline{r},\underline{t}) = A_X(s_{ij}) \sum_{\gamma \in \Gamma} \alpha_{r_j t_i}(\gamma) \, \alpha_{r_i t_j}(\gamma^{-1}) \,,$$

where  $\alpha_{r_i,t_i}(\gamma)$  are the matrix coefficients in the representation  $N_h$ ;

• for  $(\underline{r}, \underline{t})$  with  $\underline{r}$  differing from  $\underline{t}$  for indices different from  $r_i, r_j$ 

$$(\underline{r},\underline{t}) = O_m$$
.

Summing up over  $j \neq i$  we can now rewrite relations (R1') in block form for each  $i \in [1, n]$ :

1. for  $(\underline{r}, \underline{t})$  with  $\underline{r}$  differing from  $\underline{t}$  at most for the index  $r_i$ 

$$\frac{k}{2} \sum_{j \neq i} A_X(s_{ij}) \sum_{\gamma \in \Gamma} \alpha_{r_j t_i}(\gamma) \alpha_{r_i r_j}(\gamma^{-1}) = -\delta_{r_i t_i} I_m \left( 1 + \sum_{s=2}^{\nu} \frac{c_s |C_s| \chi_{N_h}(C_s)}{\dim N_h} \right) \quad (3.5)$$
where  $\delta_{r_i t_i} = \begin{cases} 0 & \text{if } r_i \neq t_i \\ 1 & \text{if } r_i = t_i \end{cases}$ ;

2. for  $(\underline{r}, \underline{t})$  with  $\underline{r}$  differing from  $\underline{t}$  at least for an index  $r_j, j \neq i$ , and at most for the pair of indices  $(r_i, r_j)$ 

$$\frac{k}{2} A_X(s_{ij}) \sum_{\gamma \in \Gamma} \alpha_{r_j t_i}(\gamma) \alpha_{r_i t_j}(\gamma^{-1}) = 0.$$
(3.6)

In all the other cases we only obtain trivial relations.

Now we observe that, using the orthogonality property of matrix coefficients of irreducible representations of a finite group, we get:

$$\sum_{\gamma \in \Gamma} \alpha_{r_j, t_i}(\gamma) \, \alpha_{r_i, t_j}(\gamma^{-1}) = \delta_{r_i t_i} \, \delta_{r_j t_j} \, \frac{|\Gamma|}{\dim N_h} \, .$$

Substituting these values in equation (3.6) we obtain trivial relations. From equation (3.5), instead, we obtain that  $\sum_{j \neq i} A_X(s_{ij})$  must be a scalar operator. Thus Lemma 3.2.3 implies that the Young diagram  $\mu$  attached to X is a rectangle, of some size  $a_h \times b_h$ , and that  $\sum_{j \neq i} A_X(s_{ij})$  acts on X as the scalar (b-a). Thus from equation

(3.5) we obtain the equation:

dim 
$$N_h + (b_h - a_h) \frac{k}{2} |\Gamma| + \sum_{s=2}^{\nu} c_s |C_s| \chi_{N_h}(C_s) = 0$$

which is exactly the equation (3.1) for the hyperplane  $\mathcal{H}_h$  in Theorem 3.2.4, part *III*). Notice that, in this case, we get a single equation since  $N_h$  is the only factor appearing in N.

We will now analyze the cases when the inertia factor of N is not the entire  $S_n$ and an actual induction is needed to build the representation  $X \otimes N \uparrow$ . If the type of N is  $\vec{n} = (n_1, \ldots, n_\nu)$ , then the inertia factor is  $S_{\vec{n}} = S_{n_1} \times \cdots \times S_{n_\nu}$  and we have:

$$X \otimes N \uparrow = \sigma_1 \cdot (X \otimes N) \oplus \cdots \oplus \sigma_\ell \cdot (X \otimes N)$$

where  $\ell = \frac{n!}{n_1! \cdots n_{\nu}!}$  and  $\{\sigma_1, \ldots, \sigma_{\ell}\}$  is a set of representatives for the left cosets of  $S_{\vec{n}}$  in  $S_n$ .

**Remark 3.2.7.** Let's denote by  $[\sigma]$  the left coset of  $\sigma$  with respect to  $S_{\vec{n}}$ . An easy computation shows that for any transposition  $s_{ij}$  and any permutation  $\sigma$ :

$$[s_{ij}\sigma] = [\sigma] \Leftrightarrow s_{\sigma^{-1}(i)\sigma^{-1}(j)} \in S_{\vec{n}}.$$

Moreover we observe that for any  $\sigma \in S_n$  and any i = 1, ..., n:

$$\gamma_i \sigma = \sigma \gamma_{\sigma^{-1}(i)} \,.$$

We are now ready to prove the following result.

**Lemma 3.2.8.** For a representation  $X \otimes N \uparrow$  of  $\Gamma_n$  of non-trivial type  $(n_1, \ldots, n_{\nu})$ relations (R1) are satisfied for some non-zero values of k if and only if:

- i)  $X = X_1 \otimes \cdots \otimes X_{\nu}$ , with  $X_h$  an irreducible representation of  $S_{n_h}$  with rectangular Young diagram;
- ii) the parameter (k, c) satisfies the corresponding system of equations in part III) of Theorem 3.2.4.

Proof. Let N be a representation of  $\Gamma^n$  of type  $(n_1, \ldots, n_\nu)$ . We observe that for any X, if we choose  $\{\sigma_1, \ldots, \sigma_\ell\} \subset S_n$  representatives of the left cosets of  $(\Gamma_n)_N$  in  $\Gamma_n$ :

$$X \otimes N \uparrow = \sigma_1 \cdot (X \otimes N) \oplus \dots \oplus \sigma_\ell \cdot (X \otimes N), \qquad (3.7)$$

is a  $\Gamma^n$ -stable decomposition of  $X \otimes N \uparrow$ . For any representative  $\sigma_l$ , let's denote by  $\sigma_l N$  the representation of  $\Gamma^n$  with same underlying vector space as N and with the action on N twisted by the automorphism induced by  $\sigma_l$  on  $\Gamma^n$  (the action of  $\gamma_i$  on  $\sigma_l N$  is the same as the action of  $\gamma_{\sigma^{-1}(i)}$  on N). Since  $\Gamma^n$  acts trivially on X, as a  $\Gamma^n$ -module the subspace  $\sigma_l \cdot (X \otimes N)$  is isomorphic to a direct sum of copies of the irreducible representation  $\sigma_l N$ . So, for a fixed i, the  $\Gamma^n$ -central operator  $-1 - \sum_{\gamma \in \Gamma \smallsetminus \{1\}} c_{\gamma} \gamma_i$  preserves the subspaces  $\sigma_l \cdot (X \otimes N)$  and acts as a scalar on each of them. For any vector  $\sigma_l \cdot v \in \sigma_l(X \otimes N)$  we have:

$$\begin{pmatrix} -1 - \sum_{\gamma \in \Gamma \smallsetminus \{1\}} c_{\gamma} \gamma_{i} \end{pmatrix} (\sigma_{l} \cdot v) = \sigma_{l} \cdot (-v) + \sigma_{l} \cdot \left( \left( -\sum_{\gamma \in \Gamma \smallsetminus \{1\}} c_{\gamma} \gamma_{\sigma_{l}^{-1}(i)} \right) v \right) \\ = C(\sigma_{l} \cdot v).$$

$$(3.8)$$

with  $C \in \mathbb{C}$ . The action of  $\frac{k}{2} \sum_{j \neq i} \sum_{\gamma \in \Gamma} s_{ij} \gamma_i \gamma_j^{-1}$  on such a vector is instead:

$$\left(\frac{k}{2}\sum_{j\neq i}\sum_{\gamma\in\Gamma}s_{ij}\gamma_i\gamma_j^{-1}\right)(\sigma_l\cdot v) = \sum_{j\neq i}\sigma_{r(ijl)}\cdot \left(\left(\frac{k}{2}\sum_{\gamma\in\Gamma}\tilde{\sigma}_{ijl}\gamma_{\sigma_l^{-1}(i)}\gamma_{\sigma_l^{-1}(j)}^{-1}\right)v\right)$$
(3.9)

where  $\sigma_{r(ijl)}$  is the representative in the set  $\{\sigma_1, \ldots, \sigma_\ell\}$  of the coset  $[s_{ij}\sigma_l]$  and  $\tilde{\sigma}_{ijl} \in S_{\vec{n}}$  is the unique element s.t.  $s_{ij}\sigma_l = \sigma_{r(ijl)}\tilde{\sigma}_{ijl}$ .

Relations (R1') are satisfied if and only if these two actions are the same. In particular  $\frac{k}{2} \sum_{j \neq i} \sum_{\gamma \in \Gamma} s_{ij} \gamma_i \gamma_j^{-1}$  must preserve the subspace  $\sigma_l \cdot (X \otimes N)$ . But let's now look at equation (3.9) and take  $j \neq i$  s.t.  $[s_{ij}\sigma_l] = [\sigma_r], r \neq l$  i.e.  $s_{ij}$  "moves" the subspace  $\sigma_l \cdot (X \otimes N)$  sending it to the subspace  $\sigma_r \cdot (X \otimes N)$ . Then we have:  $s_{\sigma_l^{-1}(i)\sigma_l^{-1}(j)} \notin S_{\vec{n}}$ . This means that the representations  $N_{\sigma_l^{-1}(i)}, N_{\sigma_l^{-1}(j)}$  are not isomorphic. As a consequence, arguing as in Lemma 3.2.5, we have that

$$\sum_{\gamma \in \Gamma} \gamma_{\sigma_l^{-1}(i)} \gamma_{\sigma_l^{-1}(j)}^{-1} = 0$$

in the representation  $X \otimes N$ , hence  $s_{ij}$  sends the subspace  $\sigma_l \cdot (X \otimes N)$  to 0. This means that  $\frac{k}{2} \sum_{j \neq i} \sum_{\gamma \in \Gamma} s_{ij} \gamma_i \gamma_j^{-1}$  indeed preserves the subspace  $\sigma_l \cdot (X \otimes N)$  and that relations (R1) split up into equations that can be checked on the subspaces  $\sigma_l \cdot (X \otimes N)$ . So in equation (3.9) it is enough to take the sum over the j s s.t.  $[s_{ij}\sigma_l] = [\sigma_l]$ . Moreover we know that if  $[s_{ij}\sigma_l] = [\sigma_l]$  then  $s_{ij}\sigma_l = \sigma_l s_{\sigma_l^{-1}(i)\sigma_l^{-1}(j)}$  i.e.  $\tilde{\sigma}_{ijl} = s_{\sigma_l^{-1}(i)\sigma_l^{-1}(j)}$ . Hence, for a fixed i, if  $\sigma_l^{-1}(i) = p$  the relations (R1') reduce to the following equations:

$$\frac{k}{2} \sum_{\substack{q \neq p \\ s_{pq} \in S_{n_{h_p}}}} s_{pq} \sum_{\gamma \in \Gamma} \gamma_p \gamma_q^{-1} = -1 - \sum_{\gamma \in \Gamma \smallsetminus \{1\}} c_\gamma \gamma_p$$
(3.10)

where the identity must be considered in the representation  $X \otimes N$  of  $S_{\vec{n}} \ltimes \Gamma^n$  and  $p \in \{\sigma_1^{-1}(i), \ldots, \sigma_\ell^{-1}(i)\}$ . For any p, equation (3.10) is exactly the p-th equation of relations (R1') for the extension of the representation of trivial type  $X_{h_p} \otimes N_{h_p}^{\otimes n_{h_p}}$  of  $S_{n_{h_p}} \ltimes \Gamma^{n_{h_p}}$  to the algebra  $H_{1,k,c}(S_{n_{h_p}} \ltimes \Gamma^{n_{h_p}})$ . It is easily checked that, letting i and  $\sigma_l$  vary, we obtain all the relations for the extension of the representations  $X_h \otimes N_h^{\otimes n_h}$  of  $S_{n_h} \otimes \Gamma^{n_h}$  for any  $n_h \neq 0$ . Using Lemma 3.2.6 we get the result.

#### **<u>STEP 4</u>** The conditions on the parameter (k, c)

Now that we classified the representations of  $\Gamma_n$  that can potentially be extended to representations of  $H_{1,k,c}(\Gamma_n)$  for  $k \neq 0$ . We would now like to show that such extensions exist for a non-empty set of values of (k, c). This amounts to proving that the system of equations in Theorem 3.2.4 part *III*) admits solutions. Fix a representation  $M = X \otimes N \uparrow$  of  $\Gamma_n$  of type  $(n_1, \ldots, n_\nu)$  satisfying conditions *i*) *ii*) of Theorem 3.2.4 part *II*). We have the following lemma.

**Lemma 3.2.9.** If  $r = \#\{h \text{ s.t. } n_h \neq 0\}$ , then the space of the solutions for the system of equations in part II) of Theorem 3.2.4 has dimension  $\nu - r$ .

*Proof.* By condition ii) we have that:

 $r = \#\{h \text{ s.t. } n_h \neq 0\} < \nu = \#\{\text{vertices in the extended Dynkin diagram of } \Gamma\}.$ 

Without loss of generality we can suppose  $n_1, \ldots, n_r \neq 0$ ,  $n_h = 0$  for h > r. So in matrix form the system has size  $r \times \nu$ , with  $r \leq \nu - 1$ :

$$\begin{pmatrix} \frac{(b_1-a_1)|\Gamma|}{2} & |C_2|\chi_{N_1}(C_2) & \dots & |C_\nu|\chi_{N_1}(C_\nu) \\ \frac{(b_2-a_2)|\Gamma|}{2} & |C_2|\chi_{N_2}(C_2) & \dots & |C_\nu|\chi_{N_2}(C_\nu) \\ \vdots & \vdots & \vdots & \vdots \\ \frac{(b_r-a_r)|\Gamma|}{2} & |C_2|\chi_{N_r}(C_2) & \dots & |C_\nu|\chi_{N_r}(C_\nu) \end{pmatrix} \begin{pmatrix} k \\ c_2 \\ \vdots \\ c_\nu \end{pmatrix} = \begin{pmatrix} -dimN_1 \\ -dimN_2 \\ \vdots \\ -dimN_r \end{pmatrix}.$$

But now we have:

$$\operatorname{rk}\left(\begin{array}{ccccc} |C_{2}|\chi_{N_{1}}(C_{2}) & \dots & |C_{\nu}|\chi_{N_{1}}(C_{\nu}) \\ |C_{2}|\chi_{N_{2}}(C_{2}) & \dots & |C_{\nu}|\chi_{N_{2}}(C_{\nu}) \\ \vdots & \vdots & \vdots \\ |C_{2}|\chi_{N_{r}}(C_{2}) & \dots & |C_{\nu}|\chi_{N_{r}}(C_{\nu}) \end{array}\right) = \operatorname{rk}\left(\begin{array}{cccc} \chi_{N_{1}}(C_{2}) & \dots & \chi_{N_{1}}(C_{\nu}) \\ \chi_{N_{2}}(C_{2}) & \dots & \chi_{N_{2}}(C_{\nu}) \\ \vdots & \vdots & \vdots \\ \chi_{N_{r}}(C_{2}) & \dots & \chi_{N_{r}}(C_{\nu}) \end{array}\right) = r.$$

In fact the rows  $R_1, \ldots, R_r$  on the RHS are rows of the character table for  $\Gamma$  from which we have erased the entries  $\chi_{N_h}(1) = \dim N_h$ . If a non-trivial linear combination  $\sum_{h=1}^r a_h R_h$  of these rows is zero then the class function  $\chi = \sum_{h=1}^r a_h \chi_{N_h}$  satisfies the equation:

$$\chi(\gamma) = 0, \quad \forall \gamma \in \Gamma \smallsetminus \{1\}.$$

This is possible only if  $\chi = m\rho$ , where  $m \in \mathbb{C}$  and  $\rho$  is the character of the regular representation. Now we must have  $m \neq 0$  since characters of non-isomorphic irreducible representations are linearly independent. But  $m \neq 0$  is also impossible since, by condition *ii*),  $N_1, \ldots N_r$  are not a complete set of irreducible representations of  $\Gamma$ while, on the other hand, any irreducible representation of  $\Gamma$  occurs in the regular representation with non-zero multiplicity. So the matrix for the system in part *III*) of Theorem 3.2.4 has maximal rank and the space of solutions has dimension  $\nu - r$ .

We are finally ready to prove part II) and III) of Theorem 3.2.4.

#### STEP 5 Proof of Theorem 3.2.4 parts II) and III)

Just combine the results of Lemma 3.2.5, Lemma 3.2.6 and Lemma 3.2.9.

# **3.3** Deforming irreducible $H_{1,0,c}(\Gamma_n)$ -modules

## 3.3.1 A proposition in deformation theory

Let A be an associative unital algebra over  $\mathbb{C}$  and let  $A_U$  be a flat formal deformation over  $\mathbb{C}[[U]]$ , where U is some finite dimensional vector space. Let M be a left Amodule.

There exists a natural map  $\eta : U \to H^2(A, \operatorname{End} M)$ . The map  $\eta$  is the composition of the map  $\phi : U \longrightarrow H^2(A, A)$  of Section 1.3 with the natural map  $\psi : H^2(A, A) \longrightarrow H^2(A, \operatorname{End} M)$  induced by functoriality by the homomorphism  $\rho : A \longrightarrow \operatorname{End} M$  giving the representation. The next proposition will be our main tool in investigating the possibility of obtaining  $H_{1,k,c+c'}(\Gamma_n)$ -modules for  $k \neq 0$  as deformations of  $H_{1,0,c}(\Gamma_n)$ -modules.

**Proposition 3.3.1.** Assume that  $\eta$  is surjective with kernel K, and moreover that  $H^1(A, \operatorname{End} M) = 0$ . Then:

(i) There exists a unique smooth formal subscheme S of the formal neighborhood of the origin in U, with tangent space K at the origin, such that M deforms to a representation of the algebra  $A_S := A_U \hat{\otimes}_{\mathbb{C}[[U]]} \mathbb{C}[S]$  (where  $\hat{\otimes}$  is the completed tensor product).

(ii) The deformation of M over S is unique.

Proof. Let us realize  $A_U$  explicitly as  $A[[\hbar_1, ..., \hbar_n]]$  equipped with a product \* as in (1.1). We may assume that K is the space of all vectors  $(\hbar_1, ..., \hbar_n)$  such that  $\hbar_{m+1} = ... = \hbar_n = 0.$ 

Let *D* be the formal neighborhood of the origin in *K*, with coordinates  $t_1 = \hbar_1, ..., t_m = \hbar_m$ . Let  $\theta : D \to U$  be a map given by the formula  $\theta(t_1, ..., t_m) = (\hbar_1, ..., \hbar_n)$ , where  $\hbar_i = t_i$  for  $i \leq m$ , and

$$\hbar_k = \sum_{k, p_1, \dots, p_m} \hbar_{k, p_1, \dots, p_m} t_1^{p_1} \dots t_m^{p_m}, \ k > m,$$

where  $\hbar_{k,p_1,...,p_m} \in \mathbb{C}$ . More briefly, we can write  $\hbar_k = \sum_{\underline{p}} \hbar_{k\underline{p}} t^{\underline{p}}$ , where  $\underline{p}$  is a multiindex. For brevity we also let  $e_j$  to be the multi-index  $(0, ..., 1_j, ..., 0)$ .

We claim that there exist unique formal functions  $\hbar_k = \hbar_k(t)$ , k > m, for which we can deform M over D. Indeed, such a deformation would be defined by a series

$$\widetilde{\rho}(a) = \sum_{\underline{p}} \rho_{\underline{p}}(a) t^{\underline{p}},$$

where  $\rho_0(a) = \rho(a)$ , and  $\rho$  is the homomorphism giving the representation M. The condition that  $\tilde{\rho}$  is a representation gives for each p

$$d \rho_{\underline{p}} = \sum_{j} \hbar_{j\underline{p}} \rho(c_{e_j}) + B_{\underline{p}}, \qquad (3.11)$$

where for  $j \leq m$ ,  $\hbar_{j\underline{p}} = 1$  if  $\underline{p} = e_j$  and zero otherwise, and  $B_{\underline{p}}$  is a 2-cocycle whose expression may involve  $\rho_{\underline{q}}$  and  $\hbar_{\underline{kq}}$  only with  $|\underline{q}| < |\underline{p}|$ . Since the map  $\eta$  is surjective, there are (unique)  $\hbar_{m+1,\underline{p}}, ..., \hbar_{\underline{np}}$  for which the right hand side is a coboundary. For such  $\hbar_{m+1,p}, ..., \hbar_{\underline{np}}$  (and only for them), we can solve (3.11) for  $\rho_p$ . This shows the existence of the functions  $\hbar_j(t)$ , j > m, such that the deformation of M over D is possible. To show the uniqueness of these functions, let  $\hbar_j$  and  $\hbar'_j$ be two sets of functions for which the deformation exists. Let  $\rho_{\underline{p}}, \rho'_{\underline{p}}$  be the coefficients of the corresponding representations  $\tilde{\rho}, \tilde{\rho}'$ . Let N be the maximal number such that  $\hbar_{j\underline{p}} = \hbar'_{j\underline{p}}$  for  $|\underline{p}| < N$ . Since  $H^1(A, \operatorname{End} M) = 0$ , the solution  $\rho_{\underline{p}}$  of (3.11) is unique up to adding a coboundary. Thus we can use changes of basis in M to modify  $\tilde{\rho}$  so that  $\rho_{\underline{p}} = \rho'_{\underline{p}}$  for  $|\underline{p}| < N$  (note that this does not affect  $\hbar_j$ ). Then for any  $\underline{q}$  with  $|\underline{q}| = N$ ,  $B_{\underline{q}}(\tilde{\rho}) = B_{\underline{q}}(\tilde{\rho}')$ , and hence  $\hbar_{j\underline{q}} = \hbar'_{j\underline{q}}$ . This contradicts the maximality of N.

Thus, we have shown that the functions  $\hbar_j$  exist and are unique; they define a parametrization of the desired subscheme S by D. Our proof also implies that the deformation of M over S is unique, so we are done.

		٦

We end this section by recalling the following fact from algebraic geometry that will guarantee that the representations we will find in Theorem 3.3.10 are actually irreducible.

Let X be an affine irreducible algebraic variety over  $\mathbb{C}$ ,  $R = \mathbb{C}[X]$ . Let A be an algebra over R and M an A-module, such that A and M are free as R-modules and M is of finite rank. For  $x \in X$ , let  $A_x$ ,  $M_x$  be the fibers of A, M at x; so  $A_x$  is a  $\mathbb{C}$ -algebra and  $M_x$  a finite dimensional module over  $A_x$ .

#### **Proposition 3.3.2.** The set of x for which $M_x$ is irreducible is open in X.

Proof. Let x be a point of X where  $M_x$  is irreducible. We have that the map  $f_x : A \to \operatorname{End} M_x$  is surjective. This means that there exist elements  $a_1, \ldots, a_{N^2}$  in A,  $N = \dim_R M$ , such that  $f_x(a_i)$  is a basis of  $\operatorname{End} M_x$ . The set U of points z of X such that  $f_z(a_i)$  are a basis of  $\operatorname{End} M_z$  is open and contains x. We found a neighborhood U of x such that, for all z in U,  $M_z$  is an irreducible  $A_z$ -module, as desired.

## 3.3.2 Deformed preprojective algebras

In order to apply Proposition 3.3.1 to our case, we need to have a closer look at the finite dimensional irreducible representations of  $H_{1,0,c}(\Gamma_n) = H_{1,c}(\Gamma)^{\otimes n} \sharp S_n$ . As we already mentioned, in analogy with the case of the wreath product groups, such representations can be recovered from a knowledge of the irreducible representations of the rank one algebra  $H_{1,c}(\Gamma)$ , using the representation theory of  $S_n$ . The complete classification of the finite dimensional irreducible representations of  $H_{1,c}(\Gamma)$  was obtained by Crawley-Boevey and Holland in [CBH98] using the representation theory of quivers and deformed preprojective algebras.

In this section we recall the Crawley-Boevey and Holland definition of the deformed preprojective algebra and how it is related with  $H_{1,c}(\Gamma)$ .

Let Q be a finite quiver (finite oriented graph) and let I be the set of its vertices. If two vertices  $i, j \in I$  are connected by an arrow a in such a way that

$$a: i \longrightarrow j$$

we will denote by h(a) = j the head of the arrow a and by t(a) = i its tail. We will denote by  $\overline{Q}$  be the double quiver of Q, obtained by adding a reverse arrow

$$a^*: j \longrightarrow i$$

for any arrow  $a: i \longrightarrow j$  of Q.

Let  $B := \bigoplus_{i \in I} \mathbb{C}e_i$  be the semisimple finite dimensional algebra spanned by orthogonal idempotents  $e_i$  corresponding to the vertices. Let let E be the vector space with basis the set of arrows of  $\overline{Q}$ . We have that E is a B-bimodule and  $E = \sum_{i,j \in I} E_{ij}$ , where  $E_{ij}$  is spanned by all the arrows a with h(a) = i, t(a) = j. We can form the path algebra of  $\overline{Q}$  defined as  $\mathbb{C}\overline{Q} := T_B E = \bigoplus_{n\geq 0} T_B^n E$ , where  $T_B^n E$  is the n-fold tensor product of E over B. Each idempotent  $e_i$  corresponds to the trivial path that does not move from the vertex i, and arrows compose as paths on the oriented graph when it is possible (otherwise their composition gives 0). For any  $\lambda \in B$  we write  $\lambda = \sum_{i \in I} \lambda_i e_i$ . For any  $i \in I$  define the element  $R_i$  of  $\mathbb{C}\overline{Q}$  as follows:

$$R_i := \sum_{\{a \in Q \mid h(a) = i\}} aa^* - \sum_{\{a \in Q \mid t(a) = i\}} a^*a.$$
(3.12)

**Definition 3.3.3.** For any  $\lambda \in B$ , the deformed preprojective algebra  $\Pi_{\lambda}(Q)$  is the quotient

$$\frac{\mathbb{C}\overline{Q}}{\langle R_i - \lambda_i e_i \rangle_{i \in I}}$$

where  $\langle \dots \rangle$  denotes the two-sided ideal generated by the indicated elements.

By [CBH98] Lemma 2.2, the algebra  $\Pi_{\lambda}(Q)$  does not depend on the orientation of Q and it is unchanged up to isomorphism by multiplying  $\lambda$  by a nonzero scalar.

Consider now any quiver Q obtained by assigning any orientation to the extended Dynkin diagram attached to  $\Gamma$  via the McKay correspondence. In this case the set of vertices I is in bijection with the set of isomorphism classes of irreducible  $\Gamma$ -modules  $\{N_i\}_{i\in I}$ . Consider a parameter  $\lambda \in \mathbb{C}^I = \mathbb{C}^{\nu}$  as above related to the parameter c of the family  $H_{1,c}(\Gamma)$  in the following way. If  $\mathbf{c} = 1 + \sum_{\gamma \in \Gamma \smallsetminus \{1\}} c_{\gamma}\gamma$  is the central element of  $\mathbb{C}[\Gamma]$  that appears in the definition of  $H_{1,c}(\Gamma)$  (see Example 2.3.4) then

$$\lambda_i = \operatorname{tr}_{N_i} \mathbf{c} \,. \tag{3.13}$$

We have the following theorem.

**Theorem 3.3.4.** For Q and  $\lambda$  as above the algebra  $H_{1,c}(\Gamma)$  is Morita equivalent to the algebra  $\Pi_{\lambda}(Q)$ .

# **3.3.3** Irreducible representations of $H_{1,c}(\Gamma)$

From the previous section we know that classifying the finite dimensional irreducible representations of  $H_{1,c}(\Gamma)$  is equivalent to solving the same problem for the algebra  $\Pi_{\lambda}(Q)$ . If we denote by  $\cdot$  the standard scalar product in  $\mathbb{C}^{\nu}$ , by  $\mathcal{R}$  the regular representation of  $\Gamma$ , and by  $\delta \in \mathbb{Z}^{\nu}$  the vector at the end of Section 3.2.2 then

$$\lambda \cdot \delta = \mathrm{tr}_{\mathcal{R}} \mathbf{c}$$

and the condition  $1 = \operatorname{tr}_{\mathcal{R}} \mathbf{c}$  corresponds to the condition  $\lambda \cdot \delta = 1$ . Thus, in particular it is enough to look at parameters  $\lambda$  with  $\lambda \cdot \delta \neq 0$ .

Any finite dimensional representation Y of  $\Pi_{\lambda}(Q)$  is also a representations of the path algebra of  $\overline{Q}$ . Thus we can attach to Y a *dimension vector*  $\alpha \in \mathbb{Z}^{\nu}$  such that  $\alpha_i = \dim e_i Y$ .

We recall now that one can associate an affine root system to the McKay graph Q. The roots of such system can be distinguished into real roots and imaginary roots. The real roots are divided into positive and negative roots, and are the images of the coordinate vectors of  $\mathbb{Z}^{\nu}$  under sequences of some suitably defined reflections, generating the Weyl group W attached to Q. The imaginary roots, instead, are all the non-zero integer multiples of the vector  $\delta$ . When  $\lambda \cdot \delta \neq 0$ , let us denote by  $R_{\lambda}$  the set of real roots  $\alpha$  such that  $\lambda \cdot \alpha = 0$  (this is a finite set), and by  $\Sigma_{\lambda}$  the unique basis of  $R_{\lambda}$  consisting of positive roots. The following theorem gives a classification of the isomorphism classes of irreducible finite dimensional representations for  $H_{1,c}(\Gamma)$ .

**Theorem 3.3.5** ([CBH98], Theorem 7.4). If  $\lambda \cdot \delta \neq 0$ , then  $\Pi_{\lambda}(Q)$  has only finitely many finite-dimensional simple modules up to isomorphisms, and they are in one-toone correspondence with the set  $\Sigma_{\lambda}$ . The correspondence is the one assigning to each module its dimension vector.

We end this section with three lemmas that will be useful in the sequel.

**Lemma 3.3.6.** The characters of the irreducible finite dimensional  $H_{1,c}(\Gamma)$ -modules are linearly independent.

Proof. Let  $\{Y_h\}_{h=1,\dots,p}$ , where  $p = |\Sigma_\lambda|$ , be a complete collection of finite dimensional, irreducible, pairwise non-isomorphic representations of  $H_{1,c}(\Gamma)$ . According to Theorem 3.3.5, we write  $\Sigma_{\lambda} \ni \alpha_{(h)} = \{\alpha_{(h),j}\}_{j=1,\dots,\nu}$  for the dimension vector corresponding to  $Y_h$  under the Morita equivalence of Section 3.3.2, where  $\lambda$  corresponds

to c as in (3.13). This means we have a decomposition  $Y_h = \bigoplus_{j=1}^{\nu} N_j^{\oplus \alpha_{(h),j}}$  as a  $\Gamma$ -module (see [CBH98], §3). Thus for the character of  $Y_h$  as a  $\Gamma$ -module we get  $(\chi_{Y_h})|_{\Gamma} = \sum_{j=1}^{\nu} \alpha_{(h),j} \chi_{N_j}$ . By Theorem 3.3.5 the  $\alpha_h$ 's are a basis of the vector space  $\operatorname{Span}_{\mathbb{C}}(R_{\lambda}) \subset \mathbb{C}^{\nu}$ , thus they are linearly independent vectors in  $\mathbb{C}^{\nu}$ . The result now follows from the fact that the  $\chi_{N_j}$ 's for  $j = 1, \ldots, \nu$  are linearly independent functions on  $\Gamma$ .

We recall that the symmetrized Ringel form attached to the quiver Q is the bilinear form (-, -) on  $\mathbb{Z}^I$  defined as follows. For any  $\alpha = \{\alpha_i\}$  and  $\beta = \{\beta_i\}$  in  $\mathbb{Z}^I$  set

$$\langle \alpha, \beta \rangle = \sum_{i \in I} \alpha_i \beta_i - \sum_{a \in Q} \alpha_{t(a)} \beta_{h(a)}.$$
 (3.14)

Then

$$(\alpha, \beta) = \langle \alpha, \beta \rangle + \langle \beta, \alpha \rangle. \tag{3.15}$$

This bilinear form is W-invariant, where W is the Weyl goup of Q. Moreover, when the underlying graph of Q is extended Dynkin, this form is positive semidefinite. The radical is generated by the vector  $\delta$  ([Kac90] Proposition 4.7, Theorem 4.8).

**Lemma 3.3.7.** Let  $Y_h$ ,  $Y_{h'}$  be two irreducible, finite dimensional, non isomorphic representations of  $H_{1,c}(\Gamma)$  and let  $\alpha_{(h)}$ ,  $\alpha_{(h')}$  be the two distinct roots corresponding to them under the Morita equivalence. The following are equivalent

- *i*)  $(\alpha_{(h)}, \alpha_{(h')}) = 0$
- *ii)*  $\operatorname{Ext}^{1}_{H_{1,c}(\Gamma)}(Y_{h}, Y_{h'}) = 0$

Proof. We can of course prove the result for the corresponding  $\Pi_{\lambda}(Q)$ -modules that we will still denote by  $Y_h$ ,  $Y_{h'}$ . The result then can be easily deduced from [CBH98], § 7. The two conditions are in fact equivalent when  $\lambda$  is a *dominant* parameter (see [CBH98] § 7 for a definition of dominance). In this case in fact, by [CBH98] Lemma 7.1, the dimension vectors  $\alpha_{(h)}, \alpha_{(h')}$  are simple roots corresponding to two distinct vertices. Thus  $\operatorname{Ext}_{\Pi_{\lambda}}^{1}(Y_{h}, Y_{h'}) \neq 0$  if and only if  $\alpha_{(h)}, \alpha_{(h')}$  correspond to adjacent vertices i.e. if and only if  $(\alpha_{(h)}, \alpha_{(h')}) \neq 0$  (see formulas (3.14), (3.15)). When  $\lambda$  is not dominant but  $\lambda \cdot \delta = 1$ , there exists an element w of the Weyl group W attached to the quiver Q such that  $w\lambda = \lambda^{+}$  is dominant. In this case the algebra  $\Pi_{\lambda}$  is Morita equivalent to the deformed preprojective algebra  $\Pi_{\lambda^{+}}$ , and this Morita equivalence acts on the dimension vectors as the element  $w \in W$ , i.e. in particular preserving the Ringel form (see [CBH98] Corollary 5.2, Lemma 7.2, and Theorem 7.4). Thus we can reduce ourselves to the case when  $\lambda$  is dominant.

**Lemma 3.3.8.** Let Y be an irreducible  $H_{1,c}(\Gamma)$ -module. Then  $\operatorname{Ext}^{1}_{H_{1,c}(\Gamma)}(Y,Y) = 0$ .

Proof. We will prove the result for the corresponding  $\Pi_{\lambda}(Q)$ -module that we will also denote by Y. But it is known ([CBH98], Corollary 7.6) that  $\Pi_{\lambda}(Q)$  contains only one minimal ideal J among all the nonzero ideals, and  $\operatorname{Ext}_{\Pi_{\lambda}(Q)/J}^{1}(Y',Y') = 0$  for any irreducible module Y' over the (finite dimensional) quotient algebra  $\Pi_{\lambda}(Q)/J$ . Since any finite dimensional  $\Pi_{\lambda}(Q)$ -module must factor through  $\Pi_{\lambda}(Q)/J$ , we get  $\operatorname{Ext}_{\Pi_{\lambda}(Q)}^{1}(Y,Y) = 0$ , as desired.

# **3.3.4** Irreducible representations of $H_{1,0,c}(\Gamma_n)$

We recall that  $H_{1,0,c}(\Gamma_n) = H_{1,c}(\Gamma)^{\otimes n} \sharp S_n$ . Let  $\{Y_h\}_{h=1,\dots,p}$ , where  $p = |\Sigma_\lambda|$ , be a complete set of pairwise non-isomorphic finite dimensional  $H_{1,c}(\Gamma)$ -modules. In a similar fashion as in Section 3.2.1, let  $\mathbf{Y} = Y_1^{\otimes n_1} \otimes \cdots \otimes Y_p^{\otimes n_p}$ , with  $\sum_i n_i = n$ , be a representation of  $H_{1,c}(\Gamma)^{\otimes n}$  of type  $\vec{n} = (n_1, \dots, n_p)$ . Set  $S_{\vec{n}} = S_{n_1} \times \cdots \times S_{n_p} \subset S_n$ , and let  $X = X_1 \otimes \cdots \otimes X_p$  be an irreducible representation of  $S_{\vec{n}}$ . As in Section 3.2.1, we will say that a representation  $\mathbf{Y}' = Y_{h_1} \otimes \cdots \otimes Y_{h_n}$  is conjugate to  $\mathbf{Y}$  (or has the same type as  $\mathbf{Y}$ ) if it has the same factors up to a permutation. The tensor product  $X \otimes \mathbf{Y}$  is an irreducible representation of the subalgebra  $H_{1,c}(\Gamma)^{\otimes n} \sharp S_{\vec{n}} \subset H_{1,0,c}(\Gamma_n)$ , where  $S_{\vec{n}}$  acts both on X and on Y (permuting the factors of Y). We can form the induced  $H_{1,0,c}(\Gamma_n)$ -module:

$$X \otimes \mathbf{Y} \uparrow := \operatorname{Ind}_{H_{1,c}(\Gamma)^{\otimes n} \sharp S_{\vec{n}}}^{H_{1,0,c}(\Gamma_n)} X \otimes Y.$$

By [Mac80] (paragraph after (A.5)), we have the following theorem.

**Theorem 3.3.9.** The representation  $X \otimes \mathbf{Y} \uparrow$  is irreducible and runs through a complete system of pairwise non-isomorphic irreducible representations of  $H_{1,0,c}(\mathbf{\Gamma}_n)$  if  $\mathbf{Y}$ runs through a complete system of pairwise non-conjugate irreducible representations of  $H_{1,c}(\Gamma)$  and, while  $\mathbf{Y}$  remains fixed, X runs through a complete system of pairwise non-isomorphic irreducible representations of  $S_{\vec{n}}$ .

# 3.3.5 The main theorem

Denote by  $M_c = X \otimes \mathbf{Y} \uparrow$  an  $H_{1,0,c}(\mathbf{\Gamma}_n)$ -module of some type  $(n_1, \ldots, n_p)$  for some c. Notice that, by Theorem 3.3.5, such representations exist only for special values of c. Denote by M the underlying vector space of  $M_c$ . Let  $\mathcal{M} = \mathcal{M}(\mathbf{Y}, X)$  be the moduli space of irreducible representations of  $H_{1,k,c}(\mathbf{\Gamma}_n)$  isomorphic to M as  $\mathbf{\Gamma}_n$ -modules (where (k, c) are allowed to vary). This is a quasi-affine algebraic variety: it is the quotient of the quasi-affine variety  $\widetilde{\mathcal{M}}(\mathbf{Y}, X)$  of extensions of the  $\mathbf{\Gamma}_n$ -module M to an irreducible  $H_{1,k,c}(\mathbf{\Gamma}_n)$ -module by a free action of the reductive group G of basis changes in M compatible with  $\mathbf{\Gamma}_n$  modulo scalars. Let  $F : \mathcal{M} \to C(\mathcal{S})$  be the morphism which sends a representation to the corresponding values of (k, c).

Let  $\chi_{Y_h}$  be the character of the representation  $Y_h$  of  $H_{1,c_0}(\Gamma)$  and let  $\alpha_{(h)}$  be the positive real root attached to  $Y_h$  under the Morita equivalence. With the same notation as in Theorem 3.2.4 we have the following result.

**Theorem 3.3.10.** Suppose  $X_h$  has rectangular Young diagram of some size  $a_h \times b_h$  for any h such that  $n_h \neq 0$ , and that moreover  $(\alpha_{(h)}, \alpha_{(h')}) = 0$  for any  $h \neq h'$  such that  $n_h, n_{h'} \neq 0$  (where we agree that the last condition is empty if  $\vec{n} = (0, \ldots, n, \ldots, 0)$ corresponds to the trivial partition). Then: (i) For any  $c_0$  the representation  $M_{c_0}$  of  $H_{1,0,c_0}(\Gamma_n)$  can be formally deformed to a representation of  $H_{1,k,c}(\Gamma_n)$  along the intersection of the hyperplanes

$$\mathcal{H}_{Y_h, a_h, b_h} : \dim Y_i + \frac{k}{2} |\Gamma|(b_i - a_i) + \sum_{s=2}^{\nu} c_s |C_s| \chi_{Y_i}(C_s) = 0$$

for all  $h \in \{1, ..., p\}$  such that  $n_h \neq 0$ , but not in other directions. This deformation is unique. The linear space described by the above equations has dimension  $\nu - r$ where  $r = \#\{h \text{ s.t. } n_h \neq 0\}$ .

(ii) The morphism F maps  $\mathcal{M}$  to  $\bigcap_{\substack{h|n_h\neq 0}} \mathcal{H}_{Y_h,a_h,b_h}$  and is étale at  $M_{c_0}$  for all  $c_0$ . Its restriction to the formal neighborhood of  $M_{c_0}$  is the deformation from (i).

(iii) There exists a nonempty Zariski open subset  $\mathcal{U}$  of  $\bigcap_{\substack{h|n_h\neq 0}} \mathcal{H}_{Y_h,a_h,b_h}$  such that for  $(k,c) \in \mathcal{U}$ , the algebra  $H_{1,k,c}(\Gamma_n)$  admits a finite dimensional irreducible representation isomorphic to M as a  $\Gamma_n$ -module.

# 3.3.6 Proof of Theorem 3.3.10

#### <u>STEP 1</u> Homological properties of $H_{1,c}(\Gamma)$

We recall the following definition (see [vdB98, vdB02, EO06]):

**Definition 3.3.11.** An algebra B is defined to be in the class VB(d) if it is of finite Hochschild dimension (i.e. there exists  $n \in \mathbb{N}$  s.t.  $H^i(B, E) = 0$  for any i > n and any B-bimodule E) and  $H^*(B, B \otimes B^o)$  is concentrated in degree d, where it equals B as a B-bimodule.

The meaning of this definition is clarified by the following result by Van den Bergh ([vdB98, vdB02]).

**Theorem 3.3.12.** If  $B \in VB(d)$  then for any B-bimodule E, the Hochschild homology  $H_i(B, E)$  is naturally isomorphic to the Hochschild cohomology  $H^{d-i}(B, E)$ .

#### **Proposition 3.3.13.** The algebra $H_{1,c}(\Gamma)$ belongs to the class VB(2).

Proof. Let us pose  $B := H_{1,c}(\Gamma)$ . If  $\Gamma = \{1\}$ , the statement is well known ([vdB98, vdB02]; see also [EO06]). Let us consider the case  $\Gamma \neq \{1\}$ . We have to show that B has finite Hochschild dimension and that:

$$H^i(B, B \otimes B^o) = 0$$
 for  $i \neq 2$ 

$$H^2(B, B \otimes B^o) \cong B$$
 as  $B$  – bimodules.

The algebra  $\mathbb{C} \langle x, y \rangle \sharp \Gamma$  has a natural increasing filtration obtained by putting x, y in degree 1 and the elements of  $\Gamma$  in degree 0. This filtration clearly induces a filtration on  $B: B = \bigcup_{n \ge 0} F^n B$ , and the associated graded algebra is  $B_0 = grB = \mathbb{C} [x, y] \sharp \Gamma$  (by the PBW theorem), which has Hochschild dimension 2. So by a deformation argument we have that B has finite Hochschild dimension (equal to 2) and  $H^i(B, B \otimes B^o) = 0$ for  $i \ne 2$ , as this is true for  $B_0$  (which is easily checked since  $B_0$  is a semidirect product of a finite group with a polynomial algebra).

It remains to show the *B*-bimodule  $E := H^2(B, B \otimes B^o)$  is isomorphic to *B*. Using again a deformation argument (cf. [vdB98]), we can see that *E* is invertible and free as a right and left *B*-module, because this is true for  $B_0$ . So  $E = B\phi$  where  $\phi$  is an automorphism of *B* such that  $gr\phi = 1$ . We will now show that  $\phi = 1$ , which will conclude the proof.

Define a linear map  $\xi : B_0 \to B_0$  as follows: if  $z \in B_0$  is a homogeneous element of degree n, and  $\tilde{z}$  is its lifting to B, then  $\xi(z)$  is defined to be the projection of the element  $\phi(\tilde{z}) - \tilde{z}$  (which has filtration degree n - 1) to  $(grB)_{n-1}$ . It is easy to check that  $\xi$  is well defined (i.e., independent on the choice of the lifting), and is a derivation of  $B_0$  of degree -1.

Our job is to show that  $\xi = 0$ . This would imply that  $\phi = 1$ , since B is generated by  $F^1B$ .

It is clear that any homogeneous inner derivation of  $B_0$  has nonnegative degree. Hence, it suffices to show that the degree -1 part of  $H^1(B_0, B_0)$  is zero. But it is easy to compute using Koszul complexes that  $H^1(B_0, B_0) = \operatorname{Vect}(L)^{\Gamma}$ , the space of  $\Gamma$ -invariant vector fields on L. In particular, vector fields of degree -1 are those with constant coefficients. But such a vector field cannot be  $\Gamma$ -invariant unless it is zero, since the space L has no nonzero vectors fixed by  $\Gamma$ . Thus,  $\xi = 0$  and we are done.

Corollary 3.3.14.  $H^2(H_{1,c}(\Gamma), \operatorname{End} Y) = H_0(H_{1,c}(\Gamma), \operatorname{End} Y) = \mathbb{C}.$ 

Proof. Posing again  $B := H_{1,c}(\Gamma)$ , we can apply Theorem 3.3.12 to obtain the first identity. Furthermore,  $H_0(B, \operatorname{End} Y) = \operatorname{End} Y/[B, \operatorname{End} Y] = \mathbb{C}$  as Y is irreducible, so the second identity follows.

# <u>STEP 2</u> Homological properties of $H_{1,0,c}(\Gamma_n)$

We would now like to apply Proposition 3.3.1 to the algebra  $H_{1,0,c_0}(\Gamma_n)$ , its flat formal deformation  $H_{1,k,c_0+c'}(\Gamma_n)$  over the finite dimensional vector space  $U = C(\mathcal{S})$ , and any module M satisfying the conditions of Theorem 3.3.10. Our job is to compute the cohomology groups  $H^2(H_{1,0,c}(\Gamma_n), \operatorname{End} M)$ ,  $H^1(H_{1,0,c}(\Gamma_n), \operatorname{End} M)$  and to show the surjectivity of the map  $\eta$ .

**Proposition 3.3.15.** If M is as in Theorem 3.3.10 and  $r \in \mathbb{N}$  is as in part (i) of the same theorem then

$$H^2(H_{1,0,c}(\Gamma_n), \operatorname{End} M) = \bigoplus_{h|n_h \neq 0} H^2(H_{1,c}(\Gamma), \operatorname{End} Y_h) = \mathbb{C}^r$$

Proof. The second equality follows from Corollary 3.3.14. Let us prove the first equality. For simplicity, let B denote the algebra  $H_{1,c}(\Gamma)$  as above, and let A be the

algebra  $H_{1,0,c}(\Gamma_n)$ . Thus we can write  $A = S_n \sharp B^{\otimes n}$ . We have:

$$H^*(A, \operatorname{End} M) = \operatorname{Ext}_{A \otimes A^o}^*(A, \operatorname{End} M)$$
$$= \operatorname{Ext}_{B^{\otimes n} \sharp S_n \otimes B^o \otimes^n \sharp S_n}^*(B^{\otimes n} \sharp S_n, \operatorname{End} M)$$
$$= \operatorname{Ext}_{(B^{\otimes n} \otimes B^o \otimes^n) \sharp S_n \times S_n}^*(B^{\otimes n} \sharp S_n, \operatorname{End} M)$$

Now, the  $(B^{\otimes n} \otimes B^{o^{\otimes n}}) \sharp S_n \times S_n$ -module  $B^{\otimes n} \sharp S_n$  is induced from the module  $B^{\otimes n}$ over the subalgebra  $(B^{\otimes n} \otimes B^{o^{\otimes n}}) \sharp S_n$ , in which  $S_n$  acts simultaneously permuting the factors of  $B^{\otimes n}$  and  $B^{o^{\otimes n}}$  (note that  $(B^{\otimes n} \otimes B^{o^{\otimes n}}) \sharp S_n$  is indeed a subalgebra of  $(B^{\otimes n} \otimes B^{o^{\otimes n}}) \sharp S_n \times S_n$  as it can be identified with the subalgebra  $(B^{\otimes n} \otimes B^{o^{\otimes n}}) \sharp D$ where  $D = \{(\sigma, \sigma), \sigma \in S_n\} \subset S_n \times S_n$ ). Applying the Shapiro Lemma we get:

$$\operatorname{Ext}_{(B^{\otimes n} \otimes B^{o \otimes n}) \sharp S_{n} \times S_{n}}^{*}(B^{\otimes n} \sharp S_{n}, \operatorname{End} M) = \operatorname{Ext}_{(B^{\otimes n} \otimes B^{o \otimes n}) \sharp S_{n}}^{*}(B^{\otimes n}, \operatorname{End} M)$$
$$= \left(\operatorname{Ext}_{B^{\otimes n} \otimes B^{o \otimes n}}^{*}(B^{\otimes n}, \operatorname{End} M)\right)^{S_{n}}.$$

We observe now that the subalgebra  $B^{\otimes n}$  is stable under the inner automorphisms induced by the elements  $\sigma \in S_n \subset A$ . Thus setting  $M' = X \otimes \mathbf{Y}$  the induced Amodule M can be written as:

$$M = \sigma_1 M' \oplus \sigma_2 M' \oplus \dots \oplus \sigma_\ell M' \tag{3.16}$$

where  $\ell = \frac{n!}{n_1!...n_r!}$  and  $\{\sigma_1, \ldots, \sigma_\ell\}$  is a set of representatives for the left cosets of  $S_{\vec{n}}$  in  $S_n$ . The action of an element  $\sigma(b_1 \otimes \cdots \otimes b_n)$  on a vector  $\sigma_l \cdot m' \in \sigma_l M'$  is the following:

$$\sigma(b_1 \otimes \cdots \otimes b_n) (\sigma_l \cdot m') = \sigma_s \cdot (\sigma'(b_{\sigma_l(1)} \otimes \cdots \otimes b_{\sigma_l(n)})m') \in \sigma_s M'$$
(3.17)

where  $\sigma' \in S_{\vec{n}}, \sigma_s \in \{\sigma_1, \ldots, \sigma_\ell\}$  are the only elements satisfying  $\sigma\sigma_l = \sigma_h \sigma'$ . In particular as a  $B^{\otimes n}$ -module, each summand  $\sigma_l M' = \sigma_l X \otimes \sigma_l Y$  equals:

$$X \otimes Y_{h_{\sigma_l^{-1}(1)}} \otimes \dots \otimes Y_{h_{\sigma_l^{-1}(n)}}$$

with trivial action of  $B^{\otimes n}$  on X.

Thus we have a chain of  $S_n$ -equivariant isomorphisms:

$$\operatorname{Ext}_{B^{\otimes n} \otimes B^{o \otimes n}}^{*}(B^{\otimes n}, \operatorname{End} M)$$

$$= \operatorname{Ext}_{B^{\otimes n} \otimes B^{o \otimes n}}^{*}\left(B^{\otimes n}, \operatorname{End}\left(\bigoplus_{l}\sigma_{l}M'\right)\right)$$

$$= \operatorname{Ext}_{B^{\otimes n} \otimes B^{o \otimes n}}^{*}\left(B^{\otimes n}, \bigoplus_{l,s}\operatorname{Hom}(\sigma_{l}M', \sigma_{s}M')\right)$$

$$= \bigoplus_{l,s}\operatorname{Ext}_{B^{\otimes n} \otimes B^{o \otimes n}}^{*}\left(B^{\otimes n}, \operatorname{Hom}(\sigma_{l}M', \sigma_{s}M')\right)$$

$$= \bigoplus_{l,s}\operatorname{Ext}_{B^{\otimes n} \otimes B^{o \otimes n}}^{*}\left(B^{\otimes n}, \operatorname{Hom}(\sigma_{l}X, \sigma_{s}X) \otimes \operatorname{Hom}(\sigma_{l}\mathbf{Y}, \sigma_{s}\mathbf{Y})\right)$$

$$= \bigoplus_{l,s}\operatorname{Hom}(\sigma_{l}X, \sigma_{s}X) \otimes \operatorname{Ext}_{B^{\otimes n} \otimes B^{o \otimes n}}^{*}\left(B^{\otimes n}, \operatorname{Hom}(\sigma_{l}\mathbf{Y}, \sigma_{s}\mathbf{Y})\right)$$

$$= \bigoplus_{l,s}\operatorname{Hom}(\sigma_{l}X, \sigma_{s}X) \otimes \operatorname{Ext}_{B^{\otimes n} \otimes B^{o \otimes n}}^{*}\left(B^{\otimes n}, \operatorname{Hom}(\sigma_{l}\mathbf{Y}, \sigma_{s}\mathbf{Y})\right)$$

$$(3.18)$$

where the third identity holds since the action of  $B^{\otimes n} \otimes B^{o \otimes n}$  does not permute the direct factors in  $\bigoplus_{l} \sigma_{l} M'$  and is trivial on X in the module M'. Since all the identities pass to invariants, all is left to do is computing the degree 2 component of the invariance of (3.18).

To this end, we apply the Künneth formula in degree 2. Lemma 3.3.7, Lemma 3.3.8 and our conditions on the  $Y_h$ 's guarantee  $Ext_B^1(Y_h, Y_{h'}) = 0$  for any  $h \neq h'$  with  $n_h, n'_h \neq 0$ . Moreover, for any  $h \neq h'$ ,  $Y_h, Y_{h'}$  are non-isomorphic irreducible representations of B, thus  $Ext_B^0(Y_h, Y_{h'}) = Hom_B(Y_h, Y_{h'}) = 0$ . As a consequence we get

$$\left(\bigoplus_{\substack{l,s\\(\sigma_l^{-1}(i)\sigma_s^{-1}(i))\in S_{\vec{n}}\,\forall i}}\operatorname{Hom}(\sigma_l X, \sigma_s X)\otimes \bigoplus_{i=1}^{n}\operatorname{Ext}_{B\otimes B^o}^2\left(B, \operatorname{Hom}(Y_{h_{\sigma_l^{-1}(i)}}, Y_{h_{\sigma_s^{-1}(i)}})\right)\right)^{S_n}$$
where  $(\sigma_l^{-1}(i)\sigma_s^{-1}(i))$  denotes the transposition moving the corresponding indices. Now we have  $(\sigma_l^{-1}(i)\sigma_h^{-1}(i)) \in S_{\vec{n}}, \forall i \text{ if and only if } \sigma_l = \sigma_s \sigma \text{ with } \sigma \in S_{\vec{n}}.$  But  $\sigma_l, \sigma_s$  belong to different left cosets of  $S_{\vec{n}}$ . Thus we can rewrite the last term as:

$$\left(\bigoplus_{l} \operatorname{End} \sigma_{l} X \otimes \bigoplus_{i} \operatorname{Ext}_{B \otimes B^{o}}^{2}(B, \operatorname{End} Y_{h_{\sigma_{l}^{-1}(i)}})\right)^{S_{n}}$$
$$= \left(\bigoplus_{h|n_{h}\neq 0} \operatorname{End} X \otimes \left(\operatorname{Ext}_{B \otimes B^{o}}^{2}(B, \operatorname{End} Y_{h})\right)^{\oplus n_{h}}\right)^{S_{\vec{n}}}$$
$$= \left(\bigoplus_{h|n_{h}\neq 0} \left(\bigotimes_{h'|n_{h'}\neq 0} \operatorname{End} X_{h'}\right) \otimes \left(\operatorname{Ext}_{B \otimes B^{o}}^{2}(B, \operatorname{End} Y_{h})\right)^{\oplus n_{h}}\right)^{S_{\vec{n}}}$$

Now we have that as an  $S_{n_h}$ -module:

$$\left(\operatorname{Ext}_{B\otimes B^{o}}^{2}\left(B,\operatorname{End}Y_{h}\right)\right)^{\oplus n_{h}}=\operatorname{Ext}_{B\otimes B^{o}}^{2}\left(B,\operatorname{End}Y_{h}\right)\otimes\mathbb{C}^{n_{h}}$$

with  $S_{n_h}$  acting only on  $\mathbb{C}^{n_h}$  by permuting the factors, and  $\mathbb{C}^{n_h} = \mathbb{C} \oplus \mathfrak{h}_h$ , where  $\mathbb{C}$  the trivial representation, and  $\mathfrak{h}_h$  is the reflection representation of  $S_{n_h}$ . So we have:

$$\begin{split} &\bigoplus_{h|n_h\neq 0} \operatorname{Ext}_{B\otimes B^o}^2 (B, \operatorname{End} Y_h) \otimes (\operatorname{End} X_1 \otimes \cdots \otimes (\mathbb{C}^{n_h} \otimes \operatorname{End} X_h) \otimes \cdots \otimes \operatorname{End} X_p)^{S_n} \\ &= \bigoplus_{h|n_h\neq 0} \operatorname{Ext}_{B\otimes B^o}^2 (B, \operatorname{End} Y_h) \otimes \\ &\otimes \left( \operatorname{End}_{S_{n_1}} X_1 \otimes \cdots \otimes (\mathbb{C} \otimes \operatorname{End} X_h \oplus \mathfrak{h}_h \otimes \operatorname{End} X_h)^{S_{n_h}} \otimes \cdots \otimes \operatorname{End}_{S_{n_p}} X_p \right) \\ &= \bigoplus_{h|n_h\neq 0} \operatorname{Ext}_{B\otimes B^o}^2 (B, \operatorname{End} Y_h) \otimes \left( \operatorname{End}_{S_{n_h}} X_h \oplus \operatorname{Hom}_{S_{n_h}} (\mathfrak{h}_h \otimes X_h, X_h) \right) \\ &= \bigoplus_{h|n_h\neq 0} \operatorname{Ext}_{B\otimes B^o}^2 (B, \operatorname{End} Y_h) = \mathbb{C}^r \end{split}$$

since, by Lemma 3.2.3, the fact that  $X_h$  has rectangular Young diagram for any h such that  $n_h \neq 0$  guarantees  $\operatorname{Hom}_{S_{n_h}}(\mathfrak{h}_h \otimes X_h, X_h) = 0$  for all such hs.

Corollary 3.3.16. If M is as in Proposition 3.3.15, then map

$$\eta: U \longrightarrow H^2(H_{1,0,c}(\mathbf{\Gamma}_n), \operatorname{End} M)$$

is surjective.

Proof. As in the previous proofs, let  $B := H_{1,c}(\Gamma)$  and  $A := H_{1,0,c}(\Gamma_n)$ . Let  $U_0 \subset U$ be the subspace of vectors (0, c'). It is sufficient to show that the restriction of  $\eta$  to  $U_0$  is surjective. But this restriction is a composition of three natural maps:

$$U_0 \to H^2(B, B) \to H^2(A, A) \to H^2(A, \operatorname{End} M).$$

Here the first map  $\eta_0 : U_0 \to H^2(B, B)$  is induced by the deformation of B along  $U_0$ , the second map  $\xi : H^2(B, B) \to H^2(A, A)$  comes from the Künneth formula, and the third map  $\psi : H^2(A, A) \to H^2(A, \operatorname{End} M)$  is induced by the homomorphism  $A \to \operatorname{End} M$ .

Now, by Proposition 3.3.15, the map  $\psi \circ \xi$  coincides with the map  $\psi_0 : H^2(B, B) \to \bigoplus_{h|n_h \neq 0} H^2(B, \operatorname{End} Y_h)$  induced by the homomorphism  $\phi : B \to \bigoplus_{h|n_h \neq 0} \operatorname{End} Y_h$ . Let  $K_0 = \operatorname{Ker}(\psi_0)$  and  $U'_0 = \eta_0^{-1}(K_0)$ . We have to show that  $\operatorname{codim} U'_0 \ge r = \sharp \{h \text{ s.t. } n_h \neq 0\}$ . By the results of **STEP 1**, Proposition 3.3.1 can be applied to the algebra B and the representation  $Y_h$  for any h such that  $n_h \neq 0$ . Thus, for any such h, the representation  $Y_h$  admits a first order deformation along  $U'_0$ . So, by the defining relations for the rank one case (see Example 2.3.4), on  $U'_0$  we must have  $\operatorname{tr}_{Y_h}(\mathbf{c}) = 0$ , where  $\mathbf{c}$  is as in Section 3.3.2. Since the representations  $Y_h$  are non-isomorphic their characters are linearly independent by Lemma 3.3.6, hence so are these linear equations. Thus the  $\operatorname{codim}(U'_0) \ge r$ , and  $\eta|_{U_0}$  is surjective, as desired.

#### **Proposition 3.3.17.** $H^1(H_{1,0,c}(\Gamma_n), \text{End}M) = 0.$

Proof. Arguing as in the proof of Proposition 3.3.15, and using the same notation, we get that  $H^1(A, \operatorname{End} M) = \bigoplus_{h|n_h \neq 0} H^1(B, \operatorname{End} Y_h)$ , which is zero by Lemma 3.3.8. This proves the proposition. We have proved the following result.

**Proposition 3.3.18.** If the conditions of Theorem 3.3.10 are satisfied then there exists a unique smooth codimension r formal subscheme S of the formal neighborhood of the origin in U such that the representation of  $H_{1,0,c_0}(\Gamma_n)$  on the vector space Mdeforms to a representation of  $H_{1,k,c_0+c'}(\Gamma_n)$  along S (i.e., abusing the language, for  $(k,c') \in S$ ). Furthermore, such a deformation over S is unique.

Proof. Corollary 3.3.16 and Proposition 3.3.17 show that our case satisfies all the hypotheses of Proposition 3.3.1. Moreover, from  $H^2(A, \operatorname{End} M) = \mathbb{C}^r$  we deduce dim Ker  $\eta = \dim U - r$ , and the Proposition follows.

#### **<u>STEP 3</u>** The subscheme S and the proof of part (i)

Now we would like to find the subscheme S of Proposition 3.3.18. We will do this computing some appropriate trace conditions for the deformation of the module M. Let  $x_i, y_i, \gamma_i, s_{ij}$  be the elements of  $H_{1,k,c}(\Gamma_n)$  that appear in Lemma 2.3.3.

Let again r be as in Theorem 3.3.10 part (*ii*). For simplicity let us write  $\mathbf{Y} = Y_{h_1}^{\otimes n_{h_1}} \otimes \cdots \otimes Y_{h_r}^{\otimes h_r}$  where  $h_s \in \{1, \ldots, p\}$  are all distinct, and  $X = X_{h_1} \otimes \cdots \otimes X_{h_r}$ . Since the  $Y_{h_s}$  are finite dimensional, irreducible, non-isomorphic representations of  $H_{1,c}(\Gamma)$ , Lemma 3.3.6 ensures that for any choice of complex numbers  $z_1, \ldots, z_r$  there exists a central element Z of  $\mathbb{C}[\Gamma]$  such that  $\chi_{Y_{h_s}}(Z) = \operatorname{tr}_{Y_{h_s}}(Z) = z_s$ . Fix now Z(1) such that  $\operatorname{tr}_{Y_{h_s}}(Z(1)) = 1 - \delta_{1s}$  and consider the element:

$$P_1 = \underbrace{1 \otimes \cdots \otimes 1}_{n_{h_1}} \otimes \underbrace{Z(1) \otimes \cdots \otimes Z(1)}_{n-n_{h_1}} \in H_{1,k,c}(\Gamma_n).$$

Such element commutes with  $x_1$ ,  $y_1$ . Consider now the relation (R1) in Lemma 2.3.3 for i = 1 (and t = 1), and multiply it by  $P_1$  on the right. The left hand side becomes  $[x_1, y_1P_1]$ , thus it has trace zero. To compute the trace of the right hand side operator it is convenient to use again the decomposition of the induced module M given in the previous section (cfr. (3.16), (3.17)). The trace of the right hand side reduces to the sum of the three terms:

$$\operatorname{tr}_{M} P_{1} = m \left( \dim Y_{h_{1}} \right)^{n_{h_{1}}} \prod_{s} \dim X_{h_{s}}$$
(3.19)

$$\frac{k}{2} \operatorname{tr}_{M} \left( \sum_{j=2}^{n_{h_{1}}} \sum_{\gamma \in \Gamma} s_{1j} \gamma_{1} \gamma_{j}^{-1} \right) P_{1} = \frac{k}{2} m \operatorname{tr}_{X_{h_{1}} \otimes Y_{h_{1}}^{\otimes n_{h_{1}}}} \left( \sum_{j=2}^{n_{h_{1}}} \sum_{\gamma \in \Gamma} s_{1j} \gamma_{1} \gamma_{j}^{-1} \right) \prod_{s \neq 1} \dim X_{h_{s}} \quad (3.20)$$
$$\operatorname{tr}_{M} \left( \sum_{\gamma \in \Gamma \smallsetminus \{1\}} c_{\gamma} \gamma_{1} \right) P_{1} = m \left( \sum_{\gamma \in \Gamma \smallsetminus \{1\}} c_{\gamma} \operatorname{tr}_{Y_{h_{1}}}(\gamma) \right) \left( \dim Y_{h_{1}} \right)^{n_{h_{1}}-1} \prod_{s} \dim X_{h_{s}} \quad (3.21)$$

where  $m = \frac{(n-n_{h_1})!}{n_{h_2}!...n_{h_r}!}$  is the number of elements  $\sigma_l$  in the set of representatives  $\{\sigma_1, \ldots, \sigma_\ell\}$  such that  $\sigma_l^{-1}(\{1, 2, \ldots, n_{h_1}\}) = \{1, 2, \ldots, n_{h_1}\}$ . Now we claim that:

$$\operatorname{tr}_{X_{h_1} \otimes Y_{h_1}^{\otimes n_{h_1}}}(s_{1j} \gamma_1 \gamma_j^{-1}) = \operatorname{tr}_{X_{h_1}}(s_{1j}) \dim Y_{h_1}^{n_{h_1}-1} \qquad \forall j \le n_{h_1}.$$
(3.22)

To obtain (3.22), we observe that, for  $j \leq n_{h_1}$ ,  $s_{1j} \gamma_1 \gamma_j^{-1}$  is conjugate in the subgroup  $\Gamma_{n_{h_1}} = S_{n_{h_1}} \ltimes \Gamma^{n_{h_1}}$  to  $s_{1j}$  and that the character of  $S_{n_{h_1}}$  on  $X_{h_1} \otimes Y_{h_1}^{\otimes n_{h_1}}$  is simply the product of the characters on  $X_{h_1}$  and  $Y_{h_1}^{\otimes n_{h_1}}$ . An easy computation gives  $\operatorname{tr}_{Y_{h_1}^{\otimes n_{h_1}}} s_{ij} = \dim Y_{h_1}^{n_{h_1}-1}$ , hence the formula.

From the proof of Lemma 3.2.3, we know that the central element  $\sum_{i < j \le n_{h_1}} s_{ij}$  in  $S_{n_{h_1}}$  acts on  $X_{h_1}$  as the scalar  $\mathbf{c}(\mu)$ , where  $\mathbf{c}(\mu)$  is the content of the Young diagram  $\mu$  attached to  $X_{h_1}$ . We can deduce that the trace of each transposition  $s_{1j}$  is  $\operatorname{tr}_{Y_{h_1}}(s_{1j}) = \frac{\dim Y_{h_1}}{n_{h_1}(n_{h_1}-1)/2} \mathbf{c}(\mu)$ . Thus if the Young diagram of  $X_{h_1}$  is of size  $a_{h_1} \times b_{h_1}$  we get:

$$\operatorname{tr}_{X_{h_1} \otimes Y_{h_1}^{\otimes n_{h_1}}}(s_{1j}\gamma_1\gamma_j^{-1}) = \frac{(b_{h_1} - a_{h_1})\dim X_{h_1}}{n_{h_1} - 1}\dim Y_{h_1}^{n_{h_1} - 1}.$$
 (3.23)

where we used Lemma 3.2.3 part (*iii*) to evaluate  $\mathbf{c}(\mu) = (b_{h_1} - a_{h_1})n_{h_1}/2$ . Substitut-

ing in (3.20), summing up (3.19), (3.20), (3.21) and simplifying we get the equation:

$$\dim Y_{h_1} + \frac{k}{2} |\Gamma| (b_{h_1} - a_{h_1}) + \sum_{\gamma \in \Gamma \smallsetminus \{1\}} c_{\gamma} \chi_{Y_{h_1}}(\gamma) = 0$$
(3.24)

that (grouping the elements  $\gamma \in \Gamma \setminus \{1\}$  according their conjugacy class) is exactly the equation for the hyperplane  $\mathcal{H}_{Y_{h_1},a_{h_1},b_{h_1}}$ . Analogously we can define Z(s),  $P_s$  and obtain the equations of the hyperplane  $\mathcal{H}_{Y_{h_s},a_{h_s},b_{h_s}}$  for  $s = 2, \ldots, r$ . We get exactly rindependent necessary linear conditions.

This shows that  $(0, c_0) + S \subset \bigcap_{s=1}^r \mathcal{H}_{Y_{h_s}, a_{h_s}, b_{h_s}}$ . But since the two subschemes have the same dimension we have that S is the formal neighborhood of zero in  $\bigcap_{i=1}^r \mathcal{H}_{Y_{h_s}, a_{h_s}, b_{h_s}} - (0, c_0)$  and Theorem 3.3.10, (*i*) is proved.

#### **<u>STEP 4</u>** Proof of part (ii) and (iii)

Let  $\mathcal{M}'$  be the formal neighborhood of  $M_{c_0}$  in  $\mathcal{M}$ . We have shown that the morphism  $F : \mathcal{M} \to U$  lands in  $\bigcap_{h|n_h\neq 0} \mathcal{H}_{Y_h,a_h,b_h}$ , and that  $F|_{\mathcal{M}'} : \mathcal{M}' \to (0,c_0) + S$  is an isomorphism. This implies that the map  $F : \mathcal{M} \to \bigcap_{h|n_h\neq 0} \mathcal{H}_{Y_h,a_h,b_h}$  is étale at  $M_{c_0}$ . This proves part (ii) of Theorem 3.3.10, and, together with Proposition 3.3.2, also implies (iii), since a map which is étale at one point is dominant.

## Chapter 4

## Continuous symplectic reflection algebras

## 4.1 Plan of the chapter

Continuous symplectic reflection algebras were introduced by Etingof, Gan, and Ginzburg in [EGG05] as a special case of continuous Hecke algebras. They are a generalization of symplectic reflection algebras to reductive algebraic groups, where the role of the group algebra is played by the ring of algebraic distributions  $\mathcal{O}(G)^*$ . In this chapter, after recalling some generalities about algebraic distributions in Section 4.2, we define continuous symplectic reflection algebras in Section 4.3 and describe some of their properties. In Section 4.4 we consider more specifically continuous symplectic reflection algebras for wreath product groups. Finally, in Section 4.5 we define infinitesimal Hecke algebras, another special case of continuous Hecke algebras we will be interested in.

## 4.2 Algebraic distributions

Let G be a reductive algebraic group over  $\mathbb{C}$ , and denote by  $\mathcal{O}(G)$  the algebra of regular functions on G. The algebraic distributions on G are the elements of the dual space  $\mathcal{O}(G)^*$ . **Example 4.2.1.** For any  $g \in G$  there is a unique distribution  $\delta_g$  such that  $\langle \delta_g, f \rangle = f(g)$  for any  $f \in \mathcal{O}(G)$ . This distribution is called the delta distribution concentrated at the point g.

The algebra  $\mathcal{O}(G)^*$  is naturally equipped with the *weak* or *initial* topology, as we are going to explain. Let  $\langle , \rangle : \mathcal{O}(G)^* \times \mathcal{O}(G) \longrightarrow \mathbb{C}$  be the natural pairing, and for any  $f \in \mathcal{O}(G)$  consider the map

$$T_f: \mathcal{O}(G)^* \longrightarrow \mathbb{C}$$
$$\mu \longrightarrow \langle \mu, f \rangle$$

The weak topology on  $\mathcal{O}(G)^*$  is the coarsest topology making all the functions  $T_f$  for  $f \in \mathcal{O}(G)$  continuous, where  $\mathbb{C}$  is given the discrete topology. To understand how the weak topology looks like, let us realize  $\mathcal{O}(G)^*$  as the projective limit

$$\mathcal{O}(G)^* = \lim U^*$$

where  $U \subset \mathcal{O}(G)$  ranges over all finite dimensional sub-vector spaces. Then, the weak topology coincides with the *inverse limit* topology on  $\mathcal{O}(G)^*$ , i.e. the coarsest topology making all the projections  $\mathcal{O}(G)^* \to U$  continuous (where each U is given the discrete topology). In this topology a system of neighborhoods of  $0 \in \mathcal{O}(G)^*$  is given by all the sub-vector spaces of  $\mathcal{O}(G)^*$  of finite codimension.

The coalgebra structure on  $\mathcal{O}(G)$  induces an algebra structure on  $\mathcal{O}(G)^*$  given by the convolution product. To ease notation we will write simply  $\mu\mu'$  for the convolution of any two distributions  $\mu, \mu' \in \mathcal{O}(G)^*$ . If  $\Delta : \mathcal{O}(G) \longrightarrow \mathcal{O}(G) \otimes \mathcal{O}(G)$  denotes the coproduct for  $\mathcal{O}(G)$ , then the convolution product  $\mu\mu'$  is the unique distribution on G such that:

$$\langle \mu \mu', f \rangle = \langle \mu \otimes \mu', \Delta(f) \rangle \quad \forall f \in \mathcal{O}(G).$$
 (4.1)

**Example 4.2.2.** If G is a finite group, the assignment  $g \longrightarrow \delta_g$  defines an algebra isomorphism  $\mathbb{C}[G] \longrightarrow \mathcal{O}(G)^*$ .

The algebra  $\mathcal{O}(G)^*$  is equipped with a natural action of the algebra  $\mathcal{O}(G)$ . For

any  $f \in \mathcal{O}(G)$  and  $\mu \in \mathcal{O}(G)^*$  define the element  $f\mu = \mu f$  as the unique distribution such that:

$$\langle f\mu, h \rangle = \langle \mu, fh \rangle \qquad \forall h \in \mathcal{O}(G).$$
 (4.2)

Let now Z be a closed subscheme of G and let I(Z) be the defining ideal of Z. An algebraic distribution  $\mu$  is said to be supported on Z if  $\mu$  annihilates all functions in I(Z). Using formula (4.2), this is equivalent to say that  $I(Z) \subset \operatorname{Ann}_{\mathcal{O}(G)}(\mu)$ . It is clear that the space of algebraic distributions on G supported on Z is isomorphic to the space of algebraic distributions on Z.

We recall that there are two left actions of G on  $\mathcal{O}(G)$  given by the left and right translation respectively. More specifically, if we denote by  $g \longrightarrow \lambda_g \in \operatorname{End}\mathcal{O}(G)$  the homomorphism giving the left translation action, and by  $g \longrightarrow \rho_g \in \operatorname{End}\mathcal{O}(G)$  the homomorphism giving the right translation we have

$$\lambda_g(f)(h) = f(g^{-1}h)$$
$$\rho_g(f)(h) = f(hg)$$

for any  $f \in \mathcal{O}(G)$  and any  $g, h \in G$ . Both these actions induce left actions on  $\mathcal{O}(G)^*$ in the obvious way. Keeping the same notation for such actions, we have:

$$\langle \rho_g(\mu), f \rangle = \langle \mu, \rho_{g^{-1}}(f) \rangle, \ \forall g \in G, \ \mu \in \mathcal{O}(G)^*, \ f \in \mathcal{O}(G)$$

(and similarly for the left translation  $\lambda_g$ ). These actions commute and thus make  $\mathcal{O}(G)^*$  into a  $G \times G$ -module. Note that as  $G \times G$ -modules  $\mathcal{O}(G)$  and  $\mathcal{O}(G)^*$  have the following decompositions:

$$\mathcal{O}(G) \cong \bigoplus_{i \in I} N_i \otimes N_i^* \cong \bigoplus_{i \in I} \operatorname{End}(N_i), \qquad \mathcal{O}(G)^* \cong \prod_{i \in I} N_i \otimes N_i^* \cong \prod_{i \in I} \operatorname{End}(N_i)$$
(4.3)

where  $N_i$  ranges over all (isomorphism classes of) irreducible finite dimensional representations of G, and  $N_i^*$  denotes the dual representation.

Example 4.2.3. For any reductive group G there exists a unique right and left

translation-invariant algebraic distribution  $\int_G(\cdot) dg : \mathcal{O}(G) \longrightarrow \mathbb{C}$  such that  $\int_G 1 dg =$ 1. If we consider the decomposition of  $\mathcal{O}(G)$  of formula (4.3) and we take  $N_0 \cong \mathbb{C}$  to be the trivial representation, then  $\int_G(\cdot) dg$  can be described as the projection on the one dimensional subspace  $N_0 \otimes N_0^*$ .

Similarly the action of G on itself by conjugation induces actions on  $\mathcal{O}(G)$  and  $\mathcal{O}(G)^*$ . These actions can be seen as the restrictions to the diagonal copy of G in  $G \times G$  of the  $G \times G$  actions on  $\mathcal{O}(G)$  and  $\mathcal{O}(G)^*$  that we described above. Thus, for any closed  $\operatorname{Ad}(G)$ -invariant subscheme Z of G, we have an induced action of G on the space  $\mathcal{O}(Z)^*$  and a natural identification  $(\mathcal{O}(Z)^*)^G = \mathcal{O}(Z/G)^*$ . We will denote by C(Z) this last space. We will identify C(Z) with the space of G-invariant distributions supported on Z.

## 4.3 Continuous symplectic reflection algebras

Let  $(V, \omega)$  be as in Section 2.2. Let  $\{v_i\}$  be any basis of V and let  $\{v_i^*\}$  be its dual basis. Let G be a reductive algebraic group with an algebraic representation  $\rho: G \longrightarrow \operatorname{Sp}(V)$ .

Let TV be the tensor algebra of V.

**Definition 4.3.1.** The semidirect product  $TV \sharp \mathcal{O}(G)^*$  is the algebra generated by  $u \in V$  and  $\mu \in \mathcal{O}(G)^*$  with the relations

$$\mu \cdot u = \sum_{i} v_i \cdot (v_i^*, gu) \mu$$

where  $(v_i^*, gu)\mu$  denotes the action of the regular function  $(v_i^*, gu)$  on  $\mu$ .

Notice that, since  $V^{\otimes N}$  is finite dimensional for all  $N \in \mathbb{Z}_{\geq 0}$ , we have

$$TV \sharp \mathcal{O}(G)^* = \bigoplus_{N \in \mathbb{Z}_{\geq 0}} V^{\otimes N} \otimes \mathcal{O}(G)^* = \bigoplus_{N \in \mathbb{Z}_{\geq 0}} ((V^*)^{\otimes N} \otimes \mathcal{O}(G))^*.$$

We can then give each summand  $((V^*)^{\otimes N} \otimes \mathcal{O}(G))^*$  the weak topology and equip

 $TV \ddagger \mathcal{O}(G)^*$  with the *direct sum topology*, i.e. the finest topology such that all the canonical injections are continuous.

Let now  $p : \bigwedge^3 V \longrightarrow V$  be the projection obtained by contracting the first two components using  $\omega$ , and let  $\Sigma$  be the closed subscheme of G defined by the equation  $p \circ \bigwedge^3 (1 - g|_V) = 0$ . This subscheme is clearly  $\operatorname{Ad}(G)$ -invariant.

**Definition 4.3.2.** For any  $t \in (\mathcal{O}(\operatorname{Ker} \rho)^*)^G$  and any  $\phi \in C(\Sigma)$  the continuous symplectic reflection algebra  $\mathcal{H}_{t,\phi}(G)$  is the quotient of the smash product  $TV \sharp \mathcal{O}(G)^*$  by the relations

$$uv - vu = \omega(u, v)t + \omega((1 - g)u, (1 - g)v)\phi$$

$$(4.4)$$

for any  $u, v \in V$ .

Moreover,  $\mathcal{H}_{0,0}(G) = SV \sharp \mathcal{O}(G)^*$ , and assigning grade degree zero to  $\mathcal{O}(G)^*$  and grade degree one to V, we obtain a filtration on  $\mathcal{H}_{t,\phi}(G)$  and a well defined surjective algebra map

$$\varphi: \mathcal{H}_{0,0}(G) = SV \sharp \mathcal{O}(G)^* \twoheadrightarrow \operatorname{gr}(\mathcal{H}_{t,\phi}(G)).$$

By [EGG05] Theorem 3.1 the map  $\varphi$  is an isomorphism, that is to say  $\mathcal{H}_{t,\phi}(G)$  satisfies the PBW property as in the finite case.

Finally, let us consider the following analog of symplectic reflections for the continuous case.

**Definition 4.3.3.** The set S of symplectic reflections is the set of elements  $s \in G$  such that  $rk(1-s)|_V \leq 2$ .

It can be seen that S is contained in the set of closed points  $\Sigma(\mathbb{C})$  of  $\Sigma$  and that any semisimple element  $g \in \Sigma(\mathbb{C})$  is in S. Using this it is easy to see, looking at the defining relations, that when G is finite and  $\rho$  is a faithful representation, the algebra  $\mathcal{H}_{t,\phi}(G)$  is the same as the symplectic reflection algebra defined in Section 2.2 (see [EGG05], § 3.1).

## 4.4 The wreath product case

Let  $L = (\mathbb{C}^2, \omega_L)$  and  $(V = L^{\oplus n}, \omega_V)$  be as in Section 2.3. Let  $\Gamma \subset SL(2, \mathbb{C}) = \operatorname{Sp}(L)$ an infinite reductive subgroup. It is well known that, up to conjugation, there exist only three such groups:

- 1)  $SL(2, \mathbb{C});$
- 2)  $GL(1,\mathbb{C}) = \mathbb{C}^*$ , identified with the maximal torus of diagonal matrices;
- 3)  $\widetilde{O}_2$ , identified with the normalizer of the maximal torus.

There is a natural symplectic action of the wreath product  $\Gamma_n = S_n \ltimes \Gamma^n$  on V. Thus we can write  $\Gamma_n \subset Sp(V)$ .

Let us now to consider the ring of algebraic distributions over  $\Gamma_n$ . This is the ring

$$\mathcal{O}(\Gamma_n)^* \cong (\mathcal{O}(\Gamma^n) \sharp \mathcal{O}(S_n))^* \cong (\mathcal{O}(\Gamma)^{\otimes n} \sharp \mathcal{O}(S_n))^* = (\mathcal{O}(\Gamma)^{\otimes n})^* \sharp \mathcal{O}(S_n)^*$$

where the last identity holds because  $\mathcal{O}(S_n) \cong \mathbb{C}[S_n]$  is finite dimensional. Let  $\{N_i\}_{i \in I}$ be a complete collection of pairwise non-isomorphic irreducible finite dimensional  $\Gamma$ modules, where  $I = \mathbb{Z}$  if  $\Gamma = GL(1, \mathbb{C})$  and  $I = \mathbb{Z}_{\geq 0}$  otherwise. Using decomposition (4.3), we can consider the filtration by ideals of  $\mathcal{O}(\Gamma)^*$  given by

$$\mathcal{O}(\Gamma)^* = \mathcal{O}(\Gamma)_0^* \supset \mathcal{O}(\Gamma)_1^* \supset \mathcal{O}(\Gamma)_2^* \supset \cdots$$

where  $\mathcal{O}(\Gamma)_N^* := \prod_{|i|>N} \operatorname{Mat}(d_i)$  and  $d_i = \dim N_i$ .

For any l = 1, ..., n and any  $N \in \mathbb{Z}_{\geq 0}$  consider the ideals in  $\mathcal{O}(\Gamma)^{*\otimes n}$ 

$$\mathcal{O}(\Gamma^{n})_{N,l}^{*} = \mathcal{O}(\Gamma)^{*\otimes l-1} \otimes \mathcal{O}(\Gamma)_{N}^{*} \otimes \mathcal{O}(\Gamma)^{*\otimes n-l} \text{ and } \mathcal{O}(\Gamma^{n})_{N}^{*} = \mathcal{O}(\Gamma^{n})_{N,1}^{*} + \dots + \mathcal{O}(\Gamma^{n})_{N,n}^{*}.$$

We denote by  $\mathcal{O}(\Gamma)^{*\hat{\otimes}n}$  the completion of the algebra  $\mathcal{O}(\Gamma)^{*\otimes n}$  with respect to the filtration by ideals

$$\mathcal{O}(\Gamma)^{*\otimes n} = \mathcal{O}(\Gamma^n)_0^* \supset \mathcal{O}(\Gamma^n)_1^* \supset \mathcal{O}(\Gamma^n)_2^* \supset \cdots$$

that is to say the projective limit

$$\mathcal{O}(\Gamma)^{*\hat{\otimes}n} := \varprojlim \mathcal{O}(\Gamma)^{*\otimes n} / \mathcal{O}(\Gamma^n)_N^*.$$

We have  $\mathcal{O}(\Gamma)^{*\hat{\otimes}n} \cong (\mathcal{O}(\Gamma)^{\otimes n})^*$ , and since from Example 4.2.2 we know  $\mathcal{O}(S_n)^* = \mathbb{C}[S_n]$ , we get:

$$\mathcal{O}(\Gamma_n)^* \cong \mathcal{O}(\Gamma)^{*\otimes n} \sharp S_n.$$

Note also that all the quotients  $\mathcal{O}(\Gamma)^{*\otimes n}/\mathcal{O}(\Gamma^n)_N^*$  are finite dimensional vector spaces. Thus, equipping these vector spaces with the discrete topology, the inverse limit topology on  $\mathcal{O}(\Gamma)^{*\otimes n} \cong \mathcal{O}(\Gamma^n)^*$  coincides also in this case with the weak topology.

We want now to give a description of the subscheme  $\Sigma$  defined in the previous section. Let  $\gamma_i, s_{ij} \in \Gamma_n$  be as in Section 2.3. Let  $S_0 = S \cap \Gamma^n$  be the set of symplectic reflections in  $\Gamma^n$  and let  $\operatorname{Ad}(\Gamma_n)s_{ij} = \{s_{lm}\gamma_l\gamma_m^{-1}|l \neq m, \gamma \in \Gamma\}$  be the conjugacy class of any transposition. Then it is easy to see that  $S = S_0 \cup \operatorname{Ad}(\Gamma_n)s_{ij}$ . The group  $\Gamma_n$  acts by conjugation on S preserving this decomposition. Moreover we have  $\Sigma = \Sigma_0 \cup \operatorname{Ad}(\Gamma_n)s_{ij}$ , where  $\Sigma_0 = \Sigma \cap \Gamma^n$ . It can be proved (see [EGG05], Proposition 6.4) that for  $\Gamma$  as in case 1),2),3) above, the set of orbits  $S/\Gamma_n$  is a scheme isomorphic to  $\Sigma/\Gamma_n$  (see [EGG05], proof of Proposition 6.4). It follows that  $C(S) = C(\Sigma)$ , and  $C(S) = C(S_0) \oplus \mathbb{C}\Delta$ , where  $\Delta$  is the integration over  $\operatorname{Ad}(\Gamma_n)s_{ij}$ , and we have a natural identification  $C(S_0) = C(\Gamma)$ . Thus for any  $\phi \in C(S)$  we can write  $\phi = (k, c), c \in C(\Gamma), k \in \mathbb{C}$ .

Since in our case  $\rho$  is the defining representation of  $\Gamma_n$  as a subgroup of Sp(V), we have Ker $\rho = \{1\}$ . Thus the parameter t can be identified with a complex number corresponding to a scalar multiple  $t\delta_1$  of the delta distribution supported at the identity element. If  $\Gamma \subset SL(2, \mathbb{C})$  is infinite reductive, though, the identity element 1 is in the closure of  $S \setminus \{1\}$ , and the parameter t can be absorbed in c.<sup>1</sup>

<sup>&</sup>lt;sup>1</sup>Note that all what we said here is not true for continuous symplectic reflection algebras in general (see for example the case of the continuous Cherednik algebra attached to the group  $O_n$ , [EGG05] § 3.3.2) and the extra parameter t becomes essential for a definition including all cases ([EGG05], § 3.1).

We can thus consider the continuous symplectic reflection algebra  $\mathcal{H}_{k,c}(\Gamma_n)$  attached to  $\Gamma_n$  and to the parameters (k, c).

We want now to give an analog of Lemma 2.3.3 for the continuous case. Let  $f \in \mathcal{O}(\Gamma_n) \cong \mathcal{O}(\Gamma)^{\otimes n} \sharp \mathcal{O}(S_n)$  be a "decomposable" function, i.e.  $f = \tilde{f}(f_1 \otimes \cdots \otimes f_n)$ , with  $\tilde{f} \in \mathcal{O}(S_n)$  and  $f_i \in \mathcal{O}(\Gamma)$  for any *i*. Then we can write for the distribution  $\Delta$ :

$$\langle \Delta, f \rangle = \sum_{i < j} \langle \delta_{s_{ij}}, \tilde{f} \rangle \left( \int_{\Gamma} f_i(\gamma) f_j(\gamma^{-1}) \, d\gamma \prod_{l \neq i, j} \langle \delta_1, f_l \rangle \right)$$

$$= \sum_{i < j} \langle \delta_{s_{ij}}, \tilde{f} \rangle \langle \Delta_{i,j}, f_1 \otimes \cdots \otimes f_n \rangle$$

$$(4.5)$$

where  $\Delta_{ij}$  is the distribution on  $\Gamma^n$  acting as shown above. Thus  $\Delta = \sum_{i,j|i < j} \delta_{s_{ij}} \Delta_{ij}$ . We denote by  $\omega_L(\gamma u, v) \Delta_{ij}$  the distribution on  $\Gamma^n$  such that

$$\langle \omega_L(\gamma u, v) \Delta_{ij}, f_1 \otimes \cdots \otimes f_n \rangle = \int_{\Gamma} \omega_L(\gamma u, v) f_i(\gamma) f_j(\gamma^{-1}) \, d\gamma \, \prod_{l \neq i, j} \langle \delta_1, f_l \rangle$$

Finally for  $c \in C(\mathcal{S})$  we will denote by  $c_i$  the algebraic distribution on  $\Gamma^n$  given by  $\delta_1^{\otimes l-1} \otimes c \otimes \delta_1^{\otimes n-l}$ .

Let  $x_i, y_i, u_i, v_i$  be as in Lemma 2.3.3.

**Lemma 4.4.1.** The algebra  $\mathcal{H}_{k,c}(\Gamma_n)$  is the quotient of  $TV \sharp \mathcal{O}(\Gamma_n)^*$  by the following relations:

(R1) For any  $i \in [1, n]$ :

$$[x_i, y_i] = c_i + 2k \sum_{j \mid j \neq i} \delta_{s_{ij}} \Delta_{ij} \,.$$

(R2) For any  $u, v \in L$  and  $i \neq j$ :

$$[u_i, v_j] = -2k\delta_{s_{ij}} (\omega_L(\gamma u, v)\Delta_{ij})$$

*Proof.* For the sake of clarity let us first look at the rank one example. In this case, according to definition 4.3.2, and absorbing the parameter t in c, the only defining

relation should be:

$$[x,y] = \omega((1-\gamma)x, (1-\gamma)y)c \tag{4.6}$$

where x, y is a symplectic basis. From this we get the expression

$$[x, y] = (2 - \operatorname{tr}_L(\gamma))c \tag{4.7}$$

where  $\operatorname{tr}_L$  denotes the trace in the defining representation of  $\Gamma$  on L. Now it is enough to show that the invariant function  $2 - \operatorname{tr}_L(\gamma)$  is not a zero divisor in  $\mathcal{O}(\Gamma)^{\Gamma}$ . Indeed, if this is true then the multiplication by  $2 - \operatorname{tr}_L(\gamma)$  is an injective linear endomorphism of  $\mathcal{O}(\Gamma)^{\Gamma}$ , thus the induced linear endomorphism of  $\mathcal{O}(\Gamma)^{*\Gamma}$  is surjective, and any invariant distribution c' can be written as  $c' = (2 - \operatorname{tr}_L(\gamma))c$ , for some  $c \in \mathcal{O}(\Gamma)^{*\Gamma} =$  $C(\Gamma)$ . But now for  $\Gamma = \mathbb{C}^*$  and  $\Gamma = SL(2,\mathbb{C})$  the ring  $O(\Gamma)^{\Gamma}$  is clearly a domain. When  $\Gamma = \widetilde{O}_2$ , the cover of the group  $O_2$ , we have  $\mathcal{O}(\Gamma)^{\Gamma} = \mathbb{C}[z, z^{-1}] \oplus \mathbb{C}$ , where the two summands come from two connected components (so it has zero divisors), but the function  $2 - \operatorname{tr}(\gamma)$ , which is clearly not identically 0 on the first summand, maps to 2 in the second summand (since  $\operatorname{tr}(\gamma) = 0$  for  $\gamma$  from the conjugacy class of complex orthogonal reflections), so it is not a zero divisor.

To pass to the higher rank case, we observe first of all that  $\omega(u_i, v_j) = 0$  if  $i \neq j$ and  $\omega(x_i, y_j) = \delta_{ij}$ . Moreover, since the distribution  $\Delta$  is supported on the conjugacy class  $\operatorname{Ad}(\Gamma_n)s_{ij} = \{s_{lm}\gamma_l\gamma_m^{-1}|l\neq m, \gamma\in\Gamma\}$  and for all  $i\neq j$  the orbit of  $s_{ij}$  under the action of  $\Gamma^n \subset \Gamma_n$  is  $\operatorname{Ad}(\Gamma^n)s_{ij} = \{s_{ij}\gamma_i\gamma_j^{-1}|\gamma\in\Gamma\}$ , we have:

$$\omega((1-g)u_i, (1-g)v_j)\Delta = (1-\delta_{ij})\left(-\omega(u_i, (\gamma^{-1}v)_i) - \omega((\gamma u)_j, v_j)\right)\delta_{s_{ij}}\Delta_{ij} \\
+ \delta_{ij}\sum_{l\neq i}\left(\omega(u_i, v_i) + \omega((\gamma u)_l, (\gamma v)_l)\right)\delta_{s_{il}}\Delta_{il} \\
= -2(1-\delta_{ij})\delta_{s_{ij}}\left(\omega_L(\gamma u, v)\Delta_{ij}\right) + 2\delta_{ij}\omega_L(u, v)\sum_{l\neq j}\delta_{s_{il}}\Delta_{il}$$
(4.8)

It's now trivial to deduce (R1), (R2) from (4.8) and the above observations.

As in Section 2.3 we have the following interesting examples.

Example 4.4.2. If n = 1 then

$$\mathcal{H}_c(\Gamma) = \frac{\mathbb{C}\langle x, y \rangle \sharp \mathcal{O}(\Gamma)^*}{\langle [x, y] - c \rangle}$$

is a continuous version of the Crawley-Boevey and Holland algebra of Example 2.3.4.

**Example 4.4.3.** When k = 0 we have

$$\mathcal{H}_{0,c}(\Gamma_n) = \mathcal{H}_c(\Gamma)^{\hat{\otimes}n} \sharp S_n.$$

**Example 4.4.4.** When  $\Gamma = GL(1, \mathbb{C})$  is a maximal torus there is a decomposition  $V = \mathfrak{h} \oplus \mathfrak{h}^*$ , where  $\mathfrak{h}$  is the irreducible representation of  $S_n \ltimes GL(1, \mathbb{C})^n$  on  $\mathbb{C}^n = \mathbb{C} \oplus \cdots \oplus \mathbb{C}$  where  $(\alpha_1, \ldots, \alpha_n) \in GL(1, \mathbb{C})^n$  acts as the matrix  $\operatorname{diag}(\alpha_1, \ldots, \alpha_n)$  and  $S_n$  permutes the vectors of the standard basis. The symplectic form  $\omega_V$  can be identified with the natural pairing between  $\mathfrak{h}$ ,  $\mathfrak{h}^*$  and this algebra is called continuous Cherednik algebra.

## 4.5 Infinitesimal Hecke algebras

The rank 1 algebra  $\mathcal{H}_c(\Gamma)$  has an interesting *infinitesimal* counterpart called the *infinitesimal Hecke algebra* (cfr. [EGG05], Section 4). In this section we recall the definition of such algebra.

For  $\Gamma = SL(2, \mathbb{C}), GL(1, \mathbb{C}), \tilde{O}_2$ , let  $\mathfrak{g}$  be the Lie algebra of  $\Gamma$ . Then the enveloping algebra  $\mathcal{U}\mathfrak{g}$  is naturally isomorphic to the subalgebra of  $\mathcal{O}(\Gamma)^*$  of all algebraic distributions set-theoretically supported at the identity element  $1 \in \Gamma$  (cf. [DG80], II, § 6). More precisely, if we identify any element  $D \in \mathcal{U}\mathfrak{g}$  with the corresponding left invariant differential operator on  $\Gamma$ , then the above mentioned isomorphism sends Dto the distribution  $\tilde{D}$  such that:

$$\langle \tilde{D}, f \rangle := (Df)(1) \quad \forall f \in \mathcal{O}(\Gamma)$$

$$(4.9)$$

where by Df we just mean D applied to f as a differential operator.

In particular,  $\operatorname{Ad}(\Gamma)$ -invariant distributions supported at the origin can be identified with elements of the center  $\mathcal{Z}(\mathcal{U}\mathfrak{g})$  of the enveloping algebra. If the distribution c belongs to the subalgebra  $\mathcal{U}\mathfrak{g} \subset \mathcal{O}(\Gamma)^*$  we define the infinitesimal Hecke algebra  $\mathcal{H}_c(\mathfrak{g})$  as the quotient:

$$\frac{TV \sharp \mathcal{U}\mathfrak{g}}{\langle [x,y]-c\rangle}.$$

When  $\mathfrak{g} = \mathfrak{sl}_2$ , representations of the algebra  $\mathcal{H}_c(\mathfrak{g})$ , called *deformed symplectic* oscillator algebra of rank 1, were studied by Khare in [Kha05]. We will compare his results with our results about finite dimensional representations of the algebra  $\mathcal{H}_c(SL(2,\mathbb{C}))$  in Section 6.3.

## Chapter 5

# Continuous deformed preprojective algebras

## 5.1 Plan of the chapter

In Section 3.3.2 and Section 3.3.3 we illustrated the fundamental role played by the theory of deformed preprojective algebras in the study of the representations of the rank one algebra  $H_{1,c}(\Gamma)$ . In the higher rank case, a remarkable development in the representation theory of wreath product symplectic reflection algebras has been the introduction, by Gan and Ginzburg ([GG05]), of the higher rank deformed preprojective algebra  $\mathcal{A}_{n,\nu,\lambda}(Q)$  for any quiver Q. The algebra  $\mathcal{A}_{n,\nu,\lambda}(Q)$  is a one-parameter deformation of the smash product  $\Pi_{\lambda}(Q)^{\otimes n} \sharp S_n$ . In the case when the underlying graph of Q is affine Dynkin, this deformation is Morita equivalent to the higher rank symplectic reflection algebra of wreath product type. Recently, following this interpretation of wreath product symplectic reflection algebras of higher rank in terms of deformed path algebras, Gan defined a version of the reflection functors for the higher rank case ([Gan06]). This allowed him to give a more elegant and transparent formulation and proof of Theorem 3.2.4 and Theorem 3.3.10, as well as a proof of the necessity of the conditions in Theorem 3.3.10.

In the light of these results, and of the extended McKay correspondence between infinite reductive subgroups of  $SL(2, \mathbb{C})$  and infinite affine Dynkin diagrams of type  $A_{\infty}, A_{+\infty}, D_{\infty}$ , in this chapter we define a continuous version of the deformed preprojective algebra  $\mathcal{A}_{n,\nu,\lambda}(Q)$  for any quiver Q with such underlying graph. For appropriate values of the parameters, we establish a Morita equivalence between the continuous symplectic reflection algebra  $\mathcal{H}_{k,c}(\Gamma_n)$  and the algebra  $\mathcal{A}_{n,\nu,\lambda}(Q)$ , where Qis any quiver with underlying graph corresponding to  $\Gamma$ . This result provides us the link we were looking for between the representation theory of quivers and deformed preprojective algebras and the representation theory of the continuous symplectic reflection algebra. As we will see in Chapter 6 it will allow us to extend Gan's methods to the continuous case.

The structure of the chapter is as follows. In Sections 5.2, and Section 5.3, we recall some generalities about infinite affine Dynkin quivers, the definitions of their root systems and of their Weyl groups. In Section 5.4 we examine some properties of the action of the Weyl group on the space of weights that will be fundamental for the application of reflection functors to the continuous case. All these facts are known, and can be easily deduced from [Kac90], but we explicitly present them here in the form and with the level of detail which is convenient for our purposes. In Section 5.5 we give the formal definition of the continuous deformed preprojective algebra  $\mathcal{A}_{n,\nu,\lambda}(Q)$ . Finally, in Section 5.6, we establish the above mentioned Morita equivalence.

## 5.2 Infinite affine quivers and the McKay correspondence for reductive subgroups of $SL(2, \mathbb{C})$

Let  $\Gamma \subset SL(2, \mathbb{C})$  be one of the three groups in Section 4.4. With the exact same procedure as in Section 3.2.2 one can associate a graph to  $\Gamma$ . In particular, if  $\{N_i\}_{i\in I}$ (where we take  $I = \mathbb{Z}$  for  $\Gamma = GL(1, \mathbb{C})$  and  $I = \mathbb{Z}_{\geq 0}$  for  $\Gamma = \widetilde{O}_2$ ,  $SL(2, \mathbb{C})$ , and  $N_0$ denotes the trivial representation) is a complete collection of finite dimensional irreducible pairwise non-isomorphic representations of  $\Gamma$ , the set of vertices of the graph attached to  $\Gamma$  is in bijection with I and is thus infinite. It is a classical result that the graphs associated to  $GL(1,\mathbb{C})$ ,  $\widetilde{O}_2$ ,  $SL(2,\mathbb{C})$  are the infinite Dynkin diagrams  $A_{\infty}$ ,  $D_{\infty}$ ,  $A_{+\infty}$ , respectively. This can be seen as an extension of the McKay correspondence to the reductive case. We will use the notation  $\Gamma$  for both the group and the corresponding graph.



Figure 5-1: Graphs associated to  $GL(1, \mathbb{C})$ ,  $\widetilde{O}_2$ , and  $SL(2, \mathbb{C})$ .

We recall that the graphs  $A_{\infty}$ ,  $D_{\infty}$ ,  $A_{+\infty}$ , together with the analog graphs  $B_{\infty}$ ,  $C_{\infty}$ , form the complete list of connected Dynkin diagrams of infinite affine Cartan matrices, i.e. generalized Cartan matrices of infinite order, such that any principal minor of finite order is positive ([Kac90], § 4.10). In particular we get the matrices:

$$A_{\infty} = \begin{pmatrix} \cdots & \cdots & \vdots & \vdots & \vdots & \cdots \\ \cdots & -1 & 2 & -1 & \vdots & \cdots \\ \cdots & \vdots & -1 & 2 & -1 & \cdots \\ \cdots & \vdots & \vdots & \vdots & \ddots & \cdots \end{pmatrix} \qquad D_{\infty} = \begin{pmatrix} 2 & 0 & -1 & \cdots & \cdots \\ 0 & 2 & -1 & \cdots & \cdots \\ -1 & -1 & 2 & -1 & \vdots & \cdots \\ 0 & 0 & -1 & 2 & -1 & \cdots \\ \vdots & \vdots & \vdots & \vdots & \vdots & \cdots \end{pmatrix}$$

$$A_{+\infty} = \begin{pmatrix} 2 & -1 & \cdots & \cdots \\ -1 & 2 & -1 & \cdots \\ \vdots & \vdots & \vdots & \cdots \end{pmatrix}$$

In the sequel we will denote by  $A = A(\Gamma)$  any such matrix and by  $\mathfrak{g}'(A)$  the corresponding Kac-Moody algebra ([Kac90], § 1, 2).

### 5.3 Infinite rank affine root systems

Here we want to give a description of the root system attached to the graph  $\Gamma$  or, equivalently, to the matrix A.

Consider the space  $(\mathbb{Z}^I)_0$  (where  $I = \mathbb{Z}$  for  $A = A_\infty$  and  $I = \mathbb{Z}_+$  for  $A = A_{+\infty}$ ,  $D_\infty$ ) of sequences  $\{\alpha_i\}_{i\in I}$  of integer numbers which have only finitely many nonzero entries. This space has a  $\mathbb{Z}$ -basis  $\{\epsilon_i\}_{i\in I}$ , where  $\epsilon_i$  is the sequence with 1 in the *i*-th position and 0 elsewhere. We will write  $\alpha = \sum_{i\in I} \alpha_i \epsilon_i$ , where  $\alpha_i = 0$  for all but finitely many indices. Denote by  $Q = Q(\Gamma)$  the quiver obtained by assigning any orientation to the graph  $\Gamma$ . Since any of the graphs  $\Gamma$  is locally finite (i.e. any vertex has finite valency), formulas (3.14), (3.15) of Section 3.3.3 make sense for any such Q and any two vectors  $\alpha, \beta \in (\mathbb{Z}^I)_0$ . Thus the symmetrized Ringel form can still be defined for any such Q as a bilinear form on  $(\mathbb{Z}^I)_0$ . Moreover, the matrix representing the Ringel form in the basis  $\{\epsilon_i\}_{i\in I}$  is exactly the Cartan matrix A.

We are now ready to define the root system for A. Our construction works more generally whenever A is the matrix of the symmetrized Ringel form for any locally finite quiver Q, in particular when the quiver is finite it coincides with the usual definition of root system for a quiver ([CBH98], § 6). Moreover our description coincides with the one given in [Kac90] (§ 7.11) for the root system of an infinite rank affine Kac-Moody algebra  $\mathfrak{g}'(A)$ .

We will say that  $\epsilon_i$  are the simple roots for A (or for Q or  $\Gamma$ ), and we will denote the set of simple roots by  $\Pi = \Pi(A)$ . Note that, since all the finite order principal minors of A are nondegenerate, the form (, ) has radical equal to  $\{0\}$  on  $(\mathbb{Z}^I)_0$ .

For any  $i \in I$  we will now define the simple reflection  $s_i : (\mathbb{Z}^I)_0 \to (\mathbb{Z}^I)_0$  by:

$$s_i(\alpha) = \alpha - (\alpha, \epsilon_i)\epsilon_i.$$

The Weyl group W attached to A (equivalently to  $\Gamma$ ) is the group of linear automorphisms of  $(\mathbb{Z}^I)_0$  generated by the simple reflections  $s_i, \forall i \in I$ .

The real roots of A, and in general for a locally finite quiver Q, are defined to be the union of the orbits of the simple roots  $\epsilon_i$  under the action of W, we will denote the set of real roots by  $\Delta^{re} = \Delta^{re}(A)$ . So we have, by definition,  $\Delta^{re} = \bigcup_{w \in W} w \Pi$ . It is standard that any such root is positive or negative (i.e. is a sum of simple roots with all non-negative, respectively non-positive, integer coefficients) and that  $\Delta^{re}_{-} = -\Delta^{re}_{+}$ . The imaginary roots of A, or in general for a locally finite quiver Q, are instead the elements of  $(\mathbb{Z}^I)_0$  that are of the form  $\pm w\beta$ , for  $\beta \in F$ , where F is the fundamental region:

$$F = \{\beta \in (\mathbb{Z}_{>0}^{I})_{0} | \beta \neq 0, \text{ support of } \beta \text{ connected, and } (\beta, \epsilon^{i}) \leq 0 \ \forall i \in I \}.$$

We denote such vectors by  $\Delta^{im} = \Delta^{im}(A)$ . The root system for A is the union of real and imaginary roots and we will denote it by  $\Delta = \Delta(A) = \Delta^{re} \cup \Delta^{im}$ . In our case, when A is an infinite rank affine Cartan matrix, we have  $\Delta = \Delta^{re}$ , and there are no imaginary roots ([Kac90], § 7.11). This is because any infinite rank affine matrix A(as well as its graph  $\Gamma$  and its root system  $\Delta$ ), can be seen as the limit of a sequence of positive finite rank Cartan matrices A(n), all of the same type, (with their Dynkin diagrams  $\Gamma(n)$  and root systems  $\Delta(n)$ ), and for such matrices there are no imaginary roots ([Kac90], § 5.2, Proposition 5.2 c).

All this is in contrast with the theory for finite affine Dynkin quivers, for which the set of imaginary roots is infinite and is given by the nonzero integer multiples of  $\delta$ . For finite affine Dynkin quivers the vector  $\delta$  is also the minimal vector with nonnegative integer coordinates among the vectors generating the one dimensional kernel of the corresponding affine Cartan matrix or, equivalently, the radical of the corresponding symmetrized Ringel form, which is positive semi-definite in this case ([Kac90], Theorem 5.6, b). We can recover the analogy with the finite case if we observe that the matrix A makes sense as an endomorphism on the space  $\mathbb{Z}^I$  of all sequences  $\{\alpha_i\}_{i\in I}$  with integer entries. This is because any of its rows has only finitely many non-zero entries. The kernel of this endomorphism of  $\mathbb{Z}^I$  is a rank one  $\mathbb{Z}$ -module generated by the vector  $\delta = \{d_i\}$ , where  $d_i = \dim N_i$ . We want to remark that this vector is *not* a root for the Kac-Moody algebra  $\mathfrak{g}'(A)$  but it can be seen as a root for a central extension of a completion of  $\mathfrak{g}'(A)$  (see [Kac90], § 7.12).

## 5.4 Action of the Weyl group on weights

Consider the space  $(\mathbb{C}^{I})_{0}$  of sequences  $\{u_{i}\}_{i\in I}$  of complex numbers such that  $u_{i} = 0$ for all but finitely many indices. Clearly the space  $(\mathbb{Z}^{I})_{0}$  above can be embedded as a lattice in  $(\mathbb{C}^{I})_{0}$  and the  $\mathbb{Z}$ -basis  $\{\epsilon_{i}\}_{i\in I}$  of  $(\mathbb{Z}^{I})_{0}$  gives a basis of  $(\mathbb{C}^{I})_{0}$  as a complex vector space via this embedding. The dual space of  $(\mathbb{C}^{I})_{0}$  is called the weight space and it is isomorphic to the space  $\mathbb{C}^{I}$ . We will denote by  $\cdot$  the standard pairing between  $(\mathbb{C}^{I})_{0}$  and  $\mathbb{C}^{I}$ , and by  $\{\epsilon_{i}^{*}\}$  the "dual basis" (spanning  $\mathbb{C}^{I}$  topologically) of  $\{\epsilon_{i}\}$  with respect to this pairing, that is to say  $\epsilon_{i}^{*} \cdot \epsilon_{j} = \delta_{ij}$ . For any  $\lambda \in \mathbb{C}^{I}$  we will write  $\lambda = \{\lambda_{i}\}$ , where  $\lambda = \sum_{i} \lambda_{i} \epsilon_{i}^{*}$  (where the sum is now possibly infinite).

We will consider  $(\mathbb{C}^I)_0$  as embedded in  $\mathbb{C}^I$  via the map

$$\begin{split} \phi &: (\mathbb{C}^I)_0 &\longrightarrow & \mathbb{C}^I \\ \epsilon_i &\longrightarrow & \sum_j a_{ji} \epsilon_j^* \end{split}$$

where  $a_{ji} = (\epsilon_j, \epsilon_i)$ . In the basis  $\{\epsilon_i\}, \{\epsilon_i^*\}$  the map  $\phi$  is given by the Cartan matrix  $A(\Gamma)$ . Moreover, for any vector  $\alpha \in \mathbb{C}^I$  and any  $i \in I$ , we have:

$$(\alpha, \epsilon_i) = \phi(\alpha) \cdot \epsilon_i = \phi(\alpha)_i \,. \tag{5.1}$$

For any reflection  $s_i, i \in I$ , we can now consider its dual reflection  $r_i : \mathbb{C}^I \to \mathbb{C}^I$ which is uniquely determined by the property

$$r_i \lambda \cdot \alpha = \lambda \cdot s_i \alpha \qquad \forall \lambda \in \mathbb{C}^I, \quad \alpha \in (\mathbb{Z}^I)_0.$$

In other words, we have  $(r_i\lambda)_j = \lambda_j - (\epsilon_i, \epsilon_j)\lambda_i$  for any j, which is equivalent to:

$$r_i \lambda = \lambda - \lambda_i \phi(\epsilon_i) \,. \tag{5.2}$$

Thus we can define an action of W on  $\mathbb{C}^{I}$  by the condition

$$\lambda \cdot (w\alpha) = w^{-1}\lambda \cdot \alpha \qquad \forall \lambda \in \mathbb{C}^I, \quad \alpha \in (\mathbb{Z}^I)_0.$$

Let now " $\prec$ " be a total ordering on  $\mathbb{C}$  satisfying the following properties ([CBH98], § 7):

- 1. If  $a \prec b$ , then  $a + c \prec b + c$ , for any  $c \in \mathbb{C}$ .
- 2. On integers  $\prec$  coincides with the usual order.
- 3. For any  $a \in \mathbb{C}$  there is  $m \in \mathbb{Z}$  with  $a \prec m$ .

An example of such an order is the lexicographic order with respect to the  $\mathbb{R}$ -basis  $\{1, \sqrt{-1}\}$  of  $\mathbb{C}$ .

We say that a weight  $\lambda$  is dominant if  $\lambda_i \succeq 0$  for all  $i \in I$ . Let  $J \subset I$  be a finite set of indices. For any weight  $\lambda$  let  $\lambda_J$  be the weight such that  $(\lambda_J)_i = \lambda_i$ , if  $i \in J$ ,  $(\lambda_J)_i = 0$  otherwise. Then we say that  $\lambda$  is J-dominant if  $\lambda_J$  is dominant.

Let now J be a finite subset of indices corresponding to some full connected subquiver  $Q_J$ . Let  $R_{\lambda,J}$  be the set of roots  $\alpha$  of Q such that the support of  $\alpha$  is contained in J, and  $\lambda \cdot \alpha = 0$ . Let  $\Sigma_{\lambda,J}$  be the set of minimal positive elements of  $R_{\lambda,J}$ . Denote by  $W_J$  the subgroup of W generated by the reflections  $s_j$ , for all  $j \in J$ . For any  $\alpha \in R_{\lambda,J}$ , let  $s_{\alpha}$  be the automorphism of  $(\mathbb{Z}^I)_0$  given by  $\beta \to \beta - (\alpha, \beta)\alpha$ and let  $W_{\lambda,J}$  be the subgroup of  $W_J$  generated by these automorphisms. We have the following lemma.

#### Lemma 5.4.1. For any $\lambda \in \mathbb{C}^{I}$

- R<sub>λ,J</sub> is the set of roots of a reduced root system in the (finite dimensional) vector space it generates. The group W<sub>λ,J</sub> acts faithfully on R<sub>λ,J</sub> and identifies with its Weyl group.
- 2) Any  $\lambda$  is  $W_J$ -conjugate to a unique J-dominant weight  $\lambda^+$ , and there exists a unique  $w^+ \in W_J$  of minimal length (in  $W_J$ ) such that  $w^+\lambda = \lambda^+$ .
- 3)  $\Sigma_{\lambda,J}$  is the unique basis of  $R_{\lambda,J}$  consisting of positive roots and  $w^+\Sigma_{\lambda,J} = \Sigma_{\lambda^+,J}$ .
- 4) If  $\lambda$  is J-dominant then  $\Sigma_{\lambda,J} = \{\epsilon_i | \lambda_i = 0\}.$

Proof. Let us start from part 1). We observe that  $Q_J$  is Dynkin. Moreover the roots of Q supported in J are in bijection with the roots of (the finite quiver)  $Q_J$ and the bijection is simply the restriction  $\alpha \to \alpha|_J$ . It is also clear that, using this identification, the group  $W_J$  acts on these roots as the Weyl group  $W(Q_J)$  of the quiver  $Q_J$ . Thus part 1) can be reduced to the analog statement for the quiver  $Q_J$ ([CBH98] Lemma 7.2 part (3) ).

Let us now prove part 2). Let  $J' = J \cup \partial J$ , where  $\partial J$  is the set of adjacent vertices for the subquiver  $Q_J$  ( the vertices that are not in  $Q_J$  but are joined to  $Q_J$  by a path of length one). Let  $U_{J'} \subset \mathbb{C}^I$  be the vector space of weights  $\mu$  satisfying  $\mu_i = 0$  for  $i \notin J'$  (i.e. the span of  $\epsilon_j^*$ ,  $j \in J'$ ). As above, let  $\lambda_{J'}$  be the weight  $(\lambda_{J'})_j = \lambda_j$  if  $j \in J'$ ,  $(\lambda_{J'})_j = 0$  otherwise. Clearly  $\lambda_{J'} \in U_{J'}$ . Write  $\lambda = \lambda_{J'} + (\lambda - \lambda_{J'})$ . Then we have that  $W_J$  fixes  $(\lambda - \lambda_{J'})$  and preserves  $U_{J'}$ . Identifying  $U_{J'}$  with a finite dimensional vector space of dimension |J'|, the weight space for the finite Dynkin quiver  $Q_{J'}$ , we see that  $W_J$  acts on  $U_{J'}$  as the parabolic subgroup  $W(Q_J) \subset W(Q_{J'})$ . The result now follows from the ordinary theory of Dynkin quivers. Indeed, write  $\lambda_{J'} = \sum_{j \in J'} b_j \nu(\epsilon_j)$ (this is clearly possible since any principal minor of the Cartan matrix  $A(\Gamma)$  is nondegenerate) and define the *height* of  $\lambda$  as  $ht(\lambda) := \sum_{j \in J'} b_j$ . Consider now a vector of the form  $w\lambda_{J'}$ , for  $w \in W(Q_J)$ , of maximal height with respect to " $\prec$ " (this exists since  $W(Q_J)$  is finite). If  $(w\lambda_{J'})_j \prec 0$  for some  $j \in J$  then, from formula (5.2), we get

$$\operatorname{ht}(w\lambda_{J'}) - \operatorname{ht}(r_j w\lambda_{J'}) = \operatorname{ht}(w\lambda_{J'} - r_j w\lambda_{J'}) = \operatorname{ht}((w\lambda_{J'})_j \nu(\epsilon_j)) = (w\lambda_{J'})_j \prec 0$$

Thus  $\operatorname{ht}(w\lambda_{J'})$  is not maximal: a contradiction. So we must have  $(w\lambda_{J'})_J = (w\lambda)_J$ dominant, and  $\lambda^+ := w\lambda$  is *J*-dominant.

For the uniqueness it is enough to prove that two J-dominant weights cannot be  $W_J$  conjugate. Suppose now  $\lambda$  is J-dominant and  $w \in W_J$ . We have to prove that if  $w\lambda$  is J-dominant then  $w\lambda = \lambda$ . Thanks to the identification with  $W(Q_J)$  the group  $W_J$  is endowed with a length function. We will prove the result by induction on the length of w. If w is of length one then  $w = r_j$  for some  $j \in J$ . If  $r_j\lambda$  is J-dominant

then  $(r_j\lambda)_i = r_j\lambda \cdot \epsilon_i \succeq 0$  for any  $i \in J$ . But

$$r_j \lambda \cdot \epsilon_j = \lambda \cdot s_j \epsilon_j = \lambda \cdot (-\epsilon_j) = -(\lambda \cdot \epsilon_j) = -\lambda_j \preceq 0$$

since  $\lambda$  is *J*-dominant. Thus it must be  $\lambda_j = \lambda \cdot \epsilon_j = 0$  and  $r_j \lambda = \lambda$  by equation (5.2). Now suppose  $w = r_{i_1} \cdots r_{i_s}$  is a reduced expression for w. Since  $i_s \in J$  we must have  $\lambda \cdot \epsilon_{i_s} \succeq 0$  and therefore  $w\lambda \cdot w\epsilon_{i_s} \succeq 0$ . But, by Lemma 3.11 of [Kac90],  $w\epsilon_{i_s}$  is a negative root (for  $Q_J$ ) and since  $w\lambda$  is *J*-dominant we must also have  $w\lambda \cdot w\epsilon_{i_s} \preceq 0$ . Thus  $0 = w\lambda \cdot w\epsilon_{i_s} = \lambda \cdot \epsilon_{i_s}$ . This implies  $r_{i_s}\lambda = \lambda$  and  $w\lambda = wr_{i_s}\lambda$ , with  $wr_{i_s}$  shorter than w and the result follows by induction. It is standard to prove that there exists a unique  $w^+ \in W_J$  of minimal length (as an element of  $W(Q_J)$ ) with the property  $w\lambda = \lambda^+$ .

The proof of part 3) is now straightforward if we use the results of 1) and 2). Part 4) is just a trivial observation.

## 5.5 Definition of the continuous deformed preprojective algebra

### 5.5.1 The rank one case

We recall that in general ([Eis95], § 7), given an algebra A and a descending filtration by ideals

$$A = \mathfrak{m}_0 \supset \mathfrak{m}_1 \supset \cdots$$

the completion of A with respect to this filtration is defined as the projective limit

$$\widehat{A} := \varprojlim A/\mathfrak{m}_i.$$

If M is an A-module (A-bimodule) the completion of M is the  $\widehat{A}$ -module ( $\widehat{A}$ -bimodule) defined as

$$\widehat{M} := \varprojlim M/\mathfrak{m}_i M \qquad (\widehat{M} := \varprojlim M/\mathfrak{m}_i M + M\mathfrak{m}_i).$$

Let now Q be a quiver with underlying graph an infinite affine Dynkin diagram of type  $A_{\infty}$ ,  $D_{\infty}$ ,  $A_{+\infty}$  with set of vertices I.

With the same notation as in Section 3.3.2, let  $B := \bigoplus_{i \in I} \mathbb{C} e_i \cong (\mathbb{C}^I)_0$  be the algebra spanned by the idempotents  $e_i$  for all  $i \in I$ . Let  $\widehat{B} := \prod_{i \in I} \mathbb{C} e_i \cong \mathbb{C}^I$  be the algebra over  $\mathbb{C}$  topologically spanned by the same idempotents. Note that  $\widehat{B} = B^*$  as vector spaces, and that  $\widehat{B}$  is the completion of B with respect to the filtration:

$$B = B_0 \supset B_1 \supset B_2 \supset \cdots$$

where  $B_N = \bigoplus_{|i| \ge N} \mathbb{C}e_i$ .

Let E be the vector space spanned by the edges of the double quiver  $\overline{Q}$ , and let  $\widehat{E}$  be the vector space with topological basis formed by the same edges. Thus  $\widehat{E}$  is a  $\widehat{B}$ -bimodule, and as such it decomposes as  $\widehat{E} = \prod_{i,j \in I} E_{ij}$ , where  $E_{ij}$  is spanned by all edges  $a \in \overline{Q}$  such that h(a) = i, t(a) = j. Note that the  $\widehat{B}$ -bimodule  $\widehat{E}$  is the completion of the *B*-bimodule *E*, i.e. it is the completion of *E* with respect to the filtration by *B*-sub-bimodules

$$E = E_0 \supset E_1 \supset E_2 \supset \cdots$$

with  $E_N := \bigoplus_{|i| \ge N} E_{ij} + \bigoplus_{|j| \ge N} E_{ij} = B_N E + EB_N$  (thus  $E/E_N = \bigoplus_{|i|,|j| < N} E_{ij}$ ). Moreover, since Q ( hence  $\overline{Q}$ ) is locally finite, all the spaces  $E_{ij}$  are finite dimensional. Thus  $E_{ij}^* \cong E_{ij}$  and we have an identification  $\widehat{E} = E^*$  as vector spaces.

For any  $k \in \mathbb{Z}_{\geq 0}$ , let us now consider the *B*-bimodule

$$T_B^k E := \underbrace{E \otimes_B \cdots \otimes_B E}_{k \text{ factors}}$$

As we know, this module is identified with the vector spaced spanned by all the paths of length k on the double quiver  $\overline{Q}$ , with its natural B-bimodule structure. Thus, the completion  $\widehat{T_B^k E}$  of this module can be identified with the vector space topologically spanned by the paths of length k, endowed with its natural  $\widehat{B}$ -bimodule structure. We will write

$$\widehat{T_B^k E} = \underbrace{\widehat{E} \hat{\otimes}_{\widehat{B}} \cdots \hat{\otimes}_{\widehat{B}} \widehat{E}}_{k \text{ factors}}$$

To justify this notation we observe that there is an isomorphism of  $\widehat{B}$ -bimodules

$$\varprojlim T_B^k E / B_N T_B^k E + T_B^k E B_N \cong \varprojlim \widehat{E} \otimes_{\widehat{B}} \cdots \otimes_{\widehat{B}} \widehat{E} / \widehat{B}_N (\widehat{E} \otimes_{\widehat{B}} \cdots \otimes_{\widehat{B}} \widehat{E}) + (\widehat{E} \otimes_{\widehat{B}} \cdots \otimes_{\widehat{B}} \widehat{E}) \widehat{B}_N$$

where  $\widehat{B}_N = \prod_{|i| \ge N} \mathbb{C}e_i$ .

Note that as vector spaces  $\widehat{T_B^k E} = (T_B^k E)^*$ .

**Definition 5.5.1.** The continuous path algebra of  $\overline{Q}$  is defined as

$$\widehat{\mathbb{C}\,\overline{Q}} := \widehat{T_B E} = \bigoplus_{k \ge 0} \widehat{T_B^k E}$$

Observe that, as vector spaces, we have

$$\widehat{T_B E} = \bigoplus_{k \ge 0} \widehat{T_B^k E} = \bigoplus_{k \ge 0} (T_B^k E)^*$$

Thus we can give to each summand the weak topology, and we can equip  $\widehat{\mathbb{C}Q}$  with the direct sum topology (the finest topology such that all the canonical injections are continuous).

We want to remark how this is a reasonable definition of continuous path algebra for the case of infinite locally finite quivers. Indeed, for any locally finite quiver, there are only a finite number of paths of length k (i.e. belonging to the space  $\widehat{T_B^k E}$ ) passing through a fixed vertex. Any element of  $\widehat{\mathbb{C}Q}$  is a, possibly infinite, linear combination of paths such that the length of the summands is bounded, and thus each vertex is contained in at most a finite number of summands. This means that we are avoiding "pathologic cases" of infinite linear combinations of paths passing through the same vertex infinitely many times. Moreover, we want to stress the fact that the algebra of Definition 5.5.1 is unital, with unit  $\prod_{i \in I} e_i$ , while the usual path algebra  $\mathbb{C} \overline{Q}$  (where we take  $B = \bigoplus_{i \in I} \mathbb{C} e_i$  and  $E = \bigoplus E_{ij}$  and usual tensor products) is not unital when the quiver is infinite.

For any  $\lambda \in \widehat{B}$  we write  $\lambda = \sum_{i \in I} \lambda_i e_i$ .

Note that, since the quiver Q is locally finite, for any  $i \in I$  the element  $R_i$  described by formula (3.12) in Section 3.3.2 is a well defined element of  $\widehat{\mathbb{C}Q}$ .

**Definition 5.5.2.** The continuous deformed preprojective algebra  $\hat{\Pi}_{\lambda}(Q)$  attached to the infinite affine quiver Q and to the parameter  $\lambda \in \widehat{B} = \mathbb{C}^{I}$  is the quotient:

$$\widehat{\Pi}_{\lambda}(Q) = \frac{\widehat{\mathbb{C}}\overline{\overline{Q}}}{\langle\langle R_i - \lambda_i e_i \rangle\rangle}_{i \in I}$$

where  $\langle \langle \dots \rangle \rangle$  is the closed ideal generated by the indicated elements in the completed path algebra  $\widehat{\mathbb{C}Q}$ .

Later in this thesis, we will need to consider a global version of the deformed preprojective algebra. So if R is a commutative unital  $\mathbb{C}$ -algebra we define  $\widehat{RQ}$  by substituting  $\widehat{B}$  with  $\widehat{B}_R := \widehat{B} \otimes R \cong R^I$  and  $\widehat{E}$  with  $\widehat{E}_R = \widehat{E} \otimes R$ . For any  $\lambda_i \in R$  for any  $i \in I$ , we define  $\widehat{\Pi}_{R,\lambda}(Q)$  to be  $\widehat{RQ}/\langle\langle R_i - \lambda_i e_i \rangle\rangle_{i \in I}$ .

#### 5.5.2 The higher rank case

The definition of higher rank continuous deformed preprojective algebra given in this section is just a generalization to the continuous case of the one given by Gan and Ginzburg in [GG05], § 1.2. Namely, let us fix a positive integer n > 1 and consider the algebra  $\mathcal{B} := B^{\otimes n}$ . For any  $l = 1, \ldots, n$  consider the ideals in  $\mathcal{B}$ 

$$\mathcal{B}_{N,l} = B^{\otimes l-1} \otimes B_N \otimes B^{\otimes n-l}$$
 and  $\mathcal{B}_N = \mathcal{B}_{N,1} + \cdots + \mathcal{B}_{N,n}$ .

where  $B_N$  is as in the previous section.

We denote by  $\widehat{\mathcal{B}} = \widehat{B}^{\otimes n}$  the completion of  $\mathcal{B}$  with respect to the chain of ideals:

$$\mathcal{B}=\mathcal{B}_0\supset\mathcal{B}_1\supset\mathcal{B}_2\supset\cdots.$$

For any  $l \in [1, n]$  consider the  $\mathcal{B}$ -bimodules:

$$\mathcal{E}_l = B^{\otimes (l-1)} \otimes E \otimes B^{\otimes (n-l)} \text{ and } \mathcal{E} = \bigoplus_{1 \le l \le n} \mathcal{E}_l$$

and the completed  $\widehat{\mathcal{B}}$ -bimodules

$$\widehat{\mathcal{E}}_l = \widehat{B}^{\hat{\otimes}(l-1)} \hat{\otimes} \widehat{E} \hat{\otimes} \widehat{B}^{\hat{\otimes}(n-l)} \text{ and } \widehat{\mathcal{E}} = \bigoplus_{1 \le l \le n} \widehat{\mathcal{E}}_l.$$

Here  $\widehat{\mathcal{E}}_l$  is the completion of the  $\mathcal{B}$ -bimodule  $\mathcal{E}_l := B^{\otimes (l-1)} \otimes E \otimes B^{(n-l)}$  with respect to the chain of sub-bimodules

$$\mathcal{E}_l = \mathcal{E}_{l,0} \supset \mathcal{E}_{l,1} \supset \mathcal{E}_{l,2} \supset \cdots$$

with  $\mathcal{E}_{l,N} = \mathcal{B}_N \mathcal{E}_l + \mathcal{E}_l \mathcal{B}_N.$ 

In a similar fashion as in the previous section we can consider the  $\mathcal{B}$ -bimodule  $T^k_{\mathcal{B}}\mathcal{E} := \underbrace{\mathcal{E} \otimes_{\mathcal{B}} \cdots \otimes_{\mathcal{B}} \mathcal{E}}_{k \text{ factors}}$  and the completed  $\widehat{\mathcal{B}}$ -bimodule  $\widehat{T^k_{\mathcal{B}}\mathcal{E}} := \underbrace{\widehat{\mathcal{E}} \otimes_{\widehat{\mathcal{B}}} \cdots \otimes_{\widehat{\mathcal{B}}} \widehat{\mathcal{E}}}_{k \text{ factors}}$ . We define

 $\widehat{T_{\mathcal{B}}\mathcal{E}} := \bigoplus_{k \ge 0} \widehat{T_{\mathcal{B}}^k\mathcal{E}}.$ 

Note that as vector spaces  $\widehat{T_{\mathcal{B}}^k \mathcal{E}} = (T_{\mathcal{B}}^k \mathcal{E})^*$ . Thus we can equip  $\widehat{T_{\mathcal{B}}^k \mathcal{E}}$  with the weak topology and  $\widehat{T_{\mathcal{B}} \mathcal{E}}$  with the direct sum topology.

Note that  $S_n$  acts naturally on  $\widehat{\mathcal{E}}$ , thus on  $\widehat{T_{\mathcal{B}}\mathcal{E}}$ . Now for any  $l \in [1, n]$ , any path  $a \in \widehat{\mathbb{C}Q}$  and any  $\underline{i} = (i_1, \ldots, i_n) \in I^n$  we consider the elements

$$|_{\underline{i}} := e_{i_1} \otimes \cdots \otimes e_{i_n} \in \widehat{\mathcal{B}}$$

and

$$a_l|_{\underline{i}} := e_{i_1} \otimes \cdots \otimes a e_{i_l} \otimes \cdots \otimes e_{i_n} \in \widehat{T_{\mathcal{B}} \mathcal{E}_l}$$

For an arrow  $a \in \overline{Q}$ , if  $i_l = t(a)$  we define

$$a_l(\underline{i}) := (i'_1, \dots, i'_n) \in I^n, \text{ where } i'_m = \begin{cases} i_m & \text{if } m \neq l \\ h(a) & \text{if } m = l \end{cases}$$

**Definition 5.5.3.** For any  $\lambda \in \widehat{B}$  and  $\nu \in \mathbb{C}$ , define the  $\widehat{\mathcal{B}}$ -algebra  $\mathcal{A}_{n,\nu,\lambda}(Q)$  to be the quotient of  $\widehat{T_{\mathcal{B}}\mathcal{E}} \sharp S_n$  by the following relations:

(I) For any  $l \in [1, n]$  and  $\underline{i} = (i_1, \ldots, i_n)$ :

$$(R_{i_l} - \lambda_{i_l})_l|_{\underline{i}} = \nu \sum_{\{m \neq l \mid i_m = i_l\}} s_{ml}|_{\underline{i}};$$

(II) For any  $l, m \in [1, n]$ ,  $l \neq m$ ,  $a, b \in \overline{Q}$  and  $\underline{i} = (i_1, \dots, i_n)$  with  $i_l = t(a)$ ,  $i_m = t(b)$ :

$$a_{l}|_{b_{m}(\underline{i})}b_{m}|_{\underline{i}} - b_{m}|_{a_{l}(\underline{i})}a_{l}|_{\underline{i}} = \begin{cases} \nu s_{lm}|_{\underline{i}} & \text{if } b \in Q \text{ and } a = b^{*} \\ -\nu s_{lm}|_{\underline{i}} & \text{if } a \in Q \text{ and } b = a^{*} \\ 0 & \text{else} \end{cases}$$

For n = 1 there is no parameter  $\nu$  and  $\mathcal{A}_{1,\lambda}(Q) = \hat{\Pi}_{\lambda}(Q)$ , while for n > 1 and  $\nu = 0$  we have  $\mathcal{A}_{n,\lambda,0}(Q) = \hat{\Pi}_{\lambda}(Q)^{\hat{\otimes}n} \sharp S_n$ .

As in the previous section, if R is a commutative unital  $\mathbb{C}$ -algebra, we can define  $\widehat{\mathcal{B}}_R = \underbrace{\widehat{B}_R \otimes_R \cdots \otimes_R \widehat{B}_R}_{n \text{ factors}}$  and similarly  $\widehat{E}_R$ . Taking  $\nu \in R$ ,  $\lambda_i \in R$  for any  $i \in I$  we can then define  $\mathcal{A}_{R,n,\nu,\lambda}$ .

## 5.6 Morita equivalence

Let Q be a McKay quiver for  $\Gamma$  and let  $\chi_{N_i}$  be the character of the irreducible representation  $N_i$ . The following theorem is the analog of Theorem 3.5.2 of [GG05].

**Theorem 5.6.1.** The algebra  $\mathcal{H}_{k,c}(\Gamma_n)$  is Morita equivalent to the algebra  $\mathcal{A}_{n,\nu,\lambda}(Q)$ for  $\nu = 2k$  and  $\lambda = \{\lambda_i\}$  where  $\lambda_i = \langle c, \chi_{N_i} \rangle$ .

Our proof of this theorem follows very closely the one of [GG05]. We report the proof in detail in Appendix A since proving this Morita equivalence is the key result from which all the results about the representation theory of  $\mathcal{H}_{k,c}(\Gamma_n)$  will follow. Also these computations can help the interested reader to become familiar with the language of algebraic distributions.

Let now  $J \subset I$  be a finite subset of indices, and consider the finite dimensional subspace

$$B_J := \bigoplus_{i \in J} \mathbb{C}e_i \subset \prod_{i \in I} \mathbb{C}e_i = \widehat{B}$$

i.e.  $B_J = \left\{ \lambda \in \widehat{B} | \lambda_i = 0 \forall i \notin J \right\}$ . Define the finite dimensional vector space  $U_J := B_J \times \mathbb{C}$ . From the above theorem and the PBW property for the continuous case (see Section 4.3) we can deduce that, for any  $\lambda_0 \in \widehat{B}$ , the family  $\{\mathcal{A}_{n,\nu,\lambda_0+\lambda}\}_{(\nu,\lambda)\in U_J}$  gives a flat formal deformation of  $\mathcal{A}_{n,0,\lambda_0}$  in dim  $U_J = |J| + 1$  parameters, where |J| is the cardinality of the set J.

Let  $\mathfrak{m}$  be the unique maximal ideal in  $\mathbb{C}[[U_J]]$ . For any  $\mathbb{C}[[U_J]]$ -module consider the decreasing filtration  $V \supset \mathfrak{m}V \supset \mathfrak{m}^2V \supset \cdots$  and the associated graded ring

$$\operatorname{gr}_{\mathfrak{m}}(V) := \prod_{h=0}^{\infty} \frac{\mathfrak{m}^h V}{\mathfrak{m}^{h+1} V}.$$

In what follows we will write  $\overline{V} := (\operatorname{gr}_{\mathfrak{m}} V)_0 = V/\mathfrak{m} V$ .

Let  $\hbar'$ ,  $\hbar_i$  for  $i \in J$  be coordinate functions on  $U_J$ , where  $\hbar'$  denotes the projection  $B_J \times \mathbb{C} \longrightarrow \mathbb{C} e_i$ . Now set  $\nu = \hbar'$  and  $\lambda_i = \hbar_i$  for any  $i \in J$ , and regard them as elements in the maximal ideal  $\mathfrak{m} \subset \mathbb{C}[[U_J]]$ . Consider the algebra  $\mathcal{A}_{\mathbb{C}}[[U_J]], n, \nu, \lambda_0 + \lambda$ . Then  $\operatorname{gr}_{\mathfrak{m}}(\mathcal{A}_{\mathbb{C}}[[U_J]], n, \nu, \lambda_0 + \lambda) \cong$  $\mathcal{A}_{n,0,\lambda_0}[[U_J]]$  as  $\mathbb{C}[[U_J]]$ -algebras. In particular  $\overline{\mathcal{A}_{\mathbb{C}}[[U_J]], n, \nu, \lambda_0 + \lambda} \cong \mathcal{A}_{n,0,\lambda_0}$ . This corresponds to the flat formal deformation of  $\mathcal{A}_{n,0,\lambda_0}$  over  $\mathbb{C}[[U_J]]$  given by the family  $\{\mathcal{A}_{n,\nu,\lambda_0+\lambda}\}_{(\nu,\lambda)\in U_J}$ .

More generally, let  $\nu \in \mathbb{C}[[U_J]]$ , and let  $\lambda \in B_J \otimes \mathbb{C}[[U_J]]$ . Consider the  $\mathbb{C}[[U_J]]$ -

algebra  $\mathcal{A}_{\mathbb{C}[[U_J]],n,\nu,\lambda}$ . The following Lemma is the analog of [Gan06] Lemma 5.13 and thanks to the PBW property of  $\mathcal{A}_{n,\nu,\lambda}$  can be proved in exactly the same way.

**Lemma 5.6.2.** Assume that  $\nu \in \mathbb{C}[U_J]$  and  $\lambda \in B_J \otimes \mathbb{C}[U_J] \subset B_J \otimes \mathbb{C}[[U_J]]$ . Then  $\operatorname{gr}_{\mathfrak{m}}(\mathcal{A}_{\mathbb{C}[[U_J]],n,\nu,\lambda}) = \overline{\mathcal{A}_{\mathbb{C}[[U_J]],n,\nu,\lambda}}[[U_J]]$  as  $\mathbb{C}[[U_J]]$ -algebras.

## Chapter 6

## Finite dimensional representations for the continuous case

## 6.1 Plan of the chapter

In this chapter we use the Morita equivalence established in Section 5.6 to study the representation theory of  $\mathcal{A}_{n,\nu,\lambda}$ .

In Section 6.2 we classify finite dimensional representations in rank one, and in Section 6.3 we compare our results for the special case  $\Gamma = SL(2, \mathbb{C})$  with the results of [Kha05] for the deformed symplectic oscillator algebra of rank one.

Finally, in Section 6.5 we consider the higher rank case and we extend the reflection functors and the results of [Gan06] to the continuous case.

## 6.2 Representations in the rank one case

The following easy result holds.

**Proposition 6.2.1.** Any finite dimensional representation of the continuous deformed preprojective algebra  $\hat{\Pi}_{\lambda}(Q)$  is a finite dimensional representation of some ordinary deformed preprojective algebra  $\Pi_{\lambda|J}(Q_J)$ , where  $J \subset I$  is a finite subset of vertices,  $Q_J$  is the corresponding full subquiver of Q, and  $\lambda|_J \in \mathbb{C}^J$  is the restriction of the parameter  $\lambda$  to set of vertices J. Vice versa, any finite dimensional representation of  $\prod_{\lambda|J}(Q_J)$  can be extended to a finite dimensional representation of  $\hat{\Pi}_{\lambda}(Q)$ .

**Proof.** By Definition 5.5.2 we have that a representation Y of  $\hat{\Pi}_{\lambda}(Q)$  is a representation of  $\widehat{CQ}$  such that, for each  $a \in Q$ , the corresponding linear maps  $a : e_{t(a)}Y \to e_{h(a)}Y, a^* : e_{h(a)}Y \to e_{t(a)}Y$  satisfy the relation

$$\sum_{\substack{a \in Q \\ h(a) = i}} aa^* - \sum_{\substack{a \in Q \\ t(a) = i}} a^*a = \lambda_i \mathrm{Id}_{e_i Y}$$

for any *i*. But now, if dim  $Y < \infty$ , we must have dim  $e_i Y = \alpha_i < \infty$  for all *i*, and  $e_i Y = 0$  for all but finitely many *i*. Thus the representation *Y* is supported at a finite number of vertices and the result follows.

Conversely, suppose  $\Pi_{\lambda|_J}(Q_J)$  admits a finite dimensional representation Y, then we can clearly extend it to a representation of  $\hat{\Pi}_{\lambda}(Q)$  by setting  $e_i Y = 0$  for  $i \notin J$ and  $a = a^* = 0$  for  $a \notin Q_J$ .

Proposition 6.2.1 implies the next Corollary.

**Corollary 6.2.2.** For any  $\lambda \in \mathbb{C}^{I}$  there is a bijection between the set of isomorphism classes of finite dimensional simple  $\hat{\Pi}_{\lambda}(Q)$ -modules and the set

$$\widehat{\Sigma}_{\lambda} := \bigcup_{J} \Sigma_{\lambda, J}$$

where J runs over all the finite subsets of indices corresponding to connected subquivers.

**Proof.** The result is implied by Proposition 6.2.1 and by the fact, proved by Crawley-Boevey and Holland, that isomorphism classes of simple  $\Pi_{\lambda|_J}(Q_J)$ -modules are in bijection with  $\Sigma_{\lambda,J}$  ([CBH98], § 7).
### **6.3** The $SL(2,\mathbb{C})$ case

We will now compare Khare's result about representation theory of the deformed symplectic oscillator algebra of rank one with the results of Section 6.2 in the case  $\Gamma = SL(2, \mathbb{C}).$ 

We observe that, in this case, the subalgebra of invariant algebraic distributions supported at the identity can be identified with the algebra of polynomials in the quadratic Casimir element  $\Delta = \frac{1}{4}(EF + FE + \frac{H^2}{2})$  (where E, F, H are the standard generators of  $\mathfrak{sl}_2$ ), that coincides with the center of the enveloping algebra. Now if we let x, y be a symplectic basis of the standard two dimensional complex symplectic vector space L, and we take  $f = f(\Delta)$  to be a polynomial with no constant coefficient, we can see that Khare's deformed symplectic oscillator algebra (cfr [Kha05], § 9)

$$H_f = \frac{TL \sharp \mathcal{U}(\mathfrak{sl}_2)}{\langle [x, y] = 1 + f(\Delta) \rangle}$$

coincides with the infinitesimal Hecke algebra  $H_c(SL(2,\mathbb{C}))$  when we take  $f = f_c$  to be an appropriate polynomial depending on c.

Let  $V_C(i)$ ,  $i \in \mathbb{Z}_{\geq 0}$  be the standard cyclic module of  $\mathfrak{sl}_2$  of highest weight i (i.e. the irreducible finite dimensional representation of dimension i + 1). Denote by  $b_i$  be the scalar by which the Casimir  $\Delta$  acts on  $V_C(i)$  ( $b_i = i(i+2)/8$ ).

Khare's classification of finite dimensional representations of  $H_f$  can be summarized as follows (see [Kha05] § 15, Theorem 11, and § 9, formula (1)).

I) There exists a unique simple  $H_f$ -module of the form

$$(*) \quad V(r,s) := \bigoplus_{i=s}^{r} V_C(i)$$

for any  $s \leq r$  in  $\mathbb{Z}_{\geq 0}$  satisfying the two conditions:

i) 
$$\sum_{i=s}^{r} (i+1)(1+f(b_i)) = 0;$$
  
ii) 
$$\sum_{i=k}^{r} (i+1)(1+f(b_i)) \neq 0 \qquad s < k \le r .$$

#### II) Any finite dimensional irreducible $H_f$ -module is isomorphic to one of the V(r, s).

We observe now that all positive roots for the infinite quiver  $A_{+\infty}$  are of the form  $\alpha = \alpha_{[s,r]} = \sum_{i=s}^{r} \epsilon_i$  for some  $0 \leq s \leq r$ , where  $\epsilon_i$  are coordinate vectors (simple roots) as in Section 5.3 (cfr. [Kac90], § 7.11, where  $\epsilon_i = \alpha_i$  in Kac's notation). Thus, according to Corollary 6.2.2, in the case of  $SL(2, \mathbb{C})$  all possible simple finite dimensional  $\mathcal{H}_c(SL(2,\mathbb{C}))$ -modules must have the form (\*). Moreover, Corollary 6.2.2 tells us that a simple representation of dimension vector  $\alpha$  exists if and only if  $\alpha \in \widehat{\Sigma}_{\lambda}$ . This condition is equivalent to the following two conditions on the root  $\alpha$ :

- a')  $\lambda \cdot \alpha = 0;$
- b') for any nontrivial decomposition  $\alpha = \beta^{(1)} + \cdots + \beta^{(n)}$  into positive roots we must have  $\lambda \cdot \beta^{(k)} \neq 0$  for some k.

Now, any decomposition as in b') looks like:

$$\alpha_{[s,r]} = \alpha_{[s,s+t_1]} + \alpha_{[s+t_1+1,s+t_1+t_2]} + \dots + \alpha_{[s+t_1+\dots+t_n+1,r]}$$

with  $s + t_1 + \cdots + t_n + 1 \leq r$ . In particular, we can consider the decompositions  $\alpha_{[s,r]} = \alpha_{[s,m-1]} + \alpha_{[m,r]}$  for any  $s < m \leq r$ . Since  $0 = \lambda \cdot \alpha_{[s,r]} = \lambda \cdot \alpha_{[s,m-1]} + \lambda \cdot \alpha_{[m,r]}$  our condition implies that in particular  $\lambda \cdot \alpha_{[m,r]} \neq 0$ . On the other hand, any nontrivial decomposition of  $\alpha_{[r,s]}$  contains a root  $\alpha_{[m,r]}$  for  $s < m \leq r$ . Thus, the conditions a'), b') can be rephrased as:

- a)  $\alpha_{[s,r]} \cdot \lambda = 0;$
- b)  $\alpha_{[m,r]} \cdot \lambda \neq 0$   $s < m \le r$ .

We will now translate the conditions a), b) on the dimension vector  $\alpha$  into Khare's conditions i), ii). In order to do this, we have to compare Khare's parameter f with our parameter c. Let's denote by  $\chi_i$  the irreducible character corresponding to  $V_C(i)$ . Then, since the  $\chi_i$ s span the space of invariant functions, we must have

$$\lambda_i = \langle c, \chi_i \rangle = \langle 1 + f(\Delta), \chi_i \rangle.$$
(6.1)

For any  $i \in \mathbb{Z}_{\geq 0}$ . We recall now that constants in  $\mathcal{U}(\mathfrak{sl}_2)$  correspond to multiples of the delta distribution  $\delta_1$ . Moreover for any  $D \in \mathcal{U}(\mathfrak{sl}_2)$  one has  $D(\chi_i)(1) = \chi_i(D) =$  $\operatorname{tr}_{V_C(i)}(D)$ , and, in particular  $\operatorname{tr}_{V_C(i)}(\Delta^l) = \dim(V_C(i))b_i^l = (i+1)b_i^l$ . It is then easy to compute that

$$\langle 1 + f(\Delta), \chi_i \rangle = (i+1)(1+f(b_i)).$$

and the equality (6.1) becomes

$$(i+1)(1+f(b_i)) = \lambda_i.$$
 (6.2)

Thus we can rewrite the conditions i) and ii) as

1) 
$$\sum_{i=s}^{r} \lambda_i = 0$$
  
2) 
$$\sum_{i=k}^{r} \lambda_i \neq 0 \quad s < k \le r$$

which correspond to conditions a), b) above respectively.

### 6.4 Gan's reflection functors

In [Gan06], Wee Liang Gan constructed reflection functors for higher rank. Gan's reflection functors are defined for any loop-free vertex i of any finite quiver Q and, under some conditions on the parameter  $\lambda, \nu$ , they establish an equivalence

$$F_i: \mathcal{A}_{R,n,\nu,\lambda}(Q) - \text{mod} \to \mathcal{A}_{R,n,\nu,r_i\lambda}(Q) - \text{mod}$$

where  $r_i \lambda$  denotes the action of the dual simple reflection  $r_i$  at the vertex i on the parameter  $\lambda \in \mathbb{R}^I$  (which we can obtain by extending the action on  $\mathbb{C}^I$  by R-linearity). Thanks to this property the functors  $F_i$  turned out to be a very powerful tool in the deformation-theoretic approach to the study of finite dimensional representations of higher rank deformed preprojective algebras. Gan's definition of reflection functors can be pushed ahead, without any change, when Q is an infinite quiver with underlying graph an affine Dynkin diagram of type  $A_{\infty}$ ,  $A_{+\infty}$ ,  $D_{\infty}$  (and more generally when Q is an infinite locally finite quiver) and we consider modules that are finitely generated over R. Gan's results about the representations of the wreath product symplectic reflection algebra will then naturally extend to the continuous case. The proofs of all the results in this section are exactly as the ones for the analog statements in [Gan06]. Thus we will mostly refer to such proofs and, when necessary, we will explain why and how they can be adapted to the case of infinite affine Dynkin quivers.

From now on, let us suppose Q is a quiver with underlying graph of type  $A_{\infty}$ ,  $A_{+\infty}$ ,  $D_{\infty}$ , and, to ease notation, let us write  $\mathcal{A}_{n,\nu,\lambda}$  for  $\mathcal{A}_{n,\nu,\lambda}(Q)$ , and  $\hat{\Pi}_{\lambda}$  for  $\hat{\Pi}_{\lambda}(Q)$ .

We observe that Gan's construction works only for loop free vertices ([GG05], 2.3) but, in our case, any vertex is such. Thus let *i* be any vertex of *Q*. Since  $\mathcal{A}_{R,n,\nu,\lambda}$ does not depend on the orientation of *Q*, we can suppose that *i* is a sink (all arrows at *i* point toward *i*). Let *V* be an  $\mathcal{A}_{R,n,\nu,\lambda}$ -module which is finitely generated as an *R*-module. The definition of  $F_i(V)$  is the same as in [Gan06] that we are now going to recall.

Let

$$\mathcal{H} := \left\{ a \in Q | h(a) = i \right\}.$$
(6.3)

**Remark 6.4.1.** Note that, for any infinite affine quiver, the set  $\mathcal{H}$  is finite.

If  $\underline{j} = (j_1, \ldots, j_n) \in I^n$ , where I is the set of vertices of Q, let

 $V_{\underline{j}} := |_{\underline{j}} V$ , and  $\Delta(\underline{j}) := \{ m \in \{1, \dots, n\} | j_m = i \}$ 

where  $|_{\underline{j}}$  is the element  $e_{j_1} \otimes \cdots \otimes e_{j_n}$  as in Section 5.5.2.

**Remark 6.4.2.** Observe that, even if Q is an infinite quiver, if V is finitely generated as a R-module then  $V_{\underline{j}} \neq 0$  only for a finite number of multi-indices  $\underline{j}$ .

For any subset  $D \subset \Delta(j)$ , consider the finite set

$$\mathcal{X}(D) := \{ \text{ all maps } \xi : D \to \mathcal{H} \}.$$
(6.4)

For any  $\xi \in \mathcal{X}(D)$  let

$$t(\underline{j},\xi) := (t_1,\ldots,t_n) \in I^n, \quad \text{where} \quad t_m = \begin{cases} j_m & \text{if } m \notin D \\ t(\xi(m)) & \text{if } m \in D \end{cases}$$

•

 $\operatorname{Set}$ 

$$V(\underline{j}, D) := \bigoplus_{\xi \in \mathcal{X}(D)} V_{t(\underline{j}, \xi)}$$

so that  $V(\underline{j}, \emptyset) = V_{\underline{j}}$ . For any  $\xi$  there are a projection and an inclusion map

$$\pi_{\underline{j},\xi}: V(\underline{j},D) \to V_{t(\underline{j},\xi)}, \qquad \mu_{\underline{j},\xi}: V_{t(\underline{j},\xi)} \hookrightarrow V(\underline{j},D).$$

Moreover, for any  $p \in D$  there is a restriction map  $\rho_p : \mathcal{X}(D) \to \mathcal{X}(D \setminus \{p\})$ . Thus for each  $\xi \in \mathcal{X}(D)$  we can consider the two compositions

$$V(\underline{j}, D) \xrightarrow{\pi_{\underline{j}, \xi}} V_{t(\underline{j}, \xi)} \xrightarrow{\xi(p)_p|_{t(\underline{j}, \xi)}} V_{t(\underline{j}, \rho_p(\xi))} \xrightarrow{\mu_{\underline{j}, \rho_p(\xi)}} V(\underline{j}, D \setminus \{p\})$$

$$V(\underline{j}, D \setminus \{p\}) \xrightarrow{\pi_{\underline{j}, \rho_p(\xi)}} V_{t(\underline{j}, \rho_p(\xi))} \xrightarrow{\xi(p)_p^*|_{t(\underline{j}, \rho_p(\xi))}} V_{t(\underline{j}, \xi)} \xrightarrow{\mu_{\underline{j}, \xi}} V(\underline{j}, D) .$$

where we recall that  $\xi(p)_p|_{t(\underline{j},\xi)} = e_{t_1} \otimes \cdots \otimes \xi(p)e_{t_p} \otimes \cdots \otimes e_{t_n}$  (and similarly for  $\xi(p)_p^*|_{t(\underline{j},\rho_p(\xi))})$ .

Define now

$$\pi_{\underline{j},p}(D): V(\underline{j},D) \to V(\underline{j},D \setminus \{p\}), \qquad \pi_{\underline{j},p}(D):=\sum_{\xi \in \mathcal{X}(D)} \mu_{\underline{j},\rho_p(\xi)}\xi(p)_p|_{t(\underline{j},\xi)}\pi_{\underline{j},\xi} \quad (6.5)$$

$$\mu_{\underline{j},p}(D): V(\underline{j}, D \setminus \{p\}) \to V(\underline{j}, D), \qquad \pi_{\underline{j},p}(D):=\sum_{\xi \in \mathcal{X}(D)} \mu_{\underline{j},\xi}\xi(p)_p^*|_{t(\underline{j},\rho_p(\xi))}\pi_{\underline{j},\rho_p(\xi)}.$$
(6.6)

Let

$$V_{\underline{j}}(D) := \begin{cases} \cap_{p \in D} \operatorname{Ker}(\pi_{\underline{j},p}(D)) & \text{if } D \neq \emptyset \\ V_{\underline{j}} & \text{if } D = \emptyset \end{cases}$$

and let  $V'_{\underline{j}} := V_{\underline{j}}(\Delta(\underline{j})).$ 

Now, for all  $\sigma \in S_n$  we set  $\sigma(\underline{j}) = (j_{\sigma^{-1}(1)}, \ldots, j_{\sigma^{-1}(n)})$ , so that  $\Delta(\sigma(\underline{j})) = \sigma(\Delta(\underline{j}))$ . For any  $D \subset \Delta(\underline{j})$  and any map  $\xi \in \mathcal{X}(D)$ , we define  $\sigma(\xi)$  as the map  $\xi \circ \sigma^{-1} \in \mathcal{X}(\sigma(D))$ . We define  $\sigma|_j$  as the map that is given by the formula

$$\sigma|_{\underline{j}}: V(\underline{j}, D) \to V(\sigma(\underline{j}), \sigma(D)), \qquad \sigma|_{\underline{j}}:=\sum_{\xi \in \mathcal{X}(D)} \mu_{\sigma(\underline{j}), \sigma(\xi)} \sigma \pi_{\underline{j}, \xi}$$

for any  $D \subset \Delta(j)$ .

**Definition 6.4.3.** We define  $F_i(V) := V' = \bigoplus_{\underline{j} \in I^n} V'_{\underline{j}}$  as a  $\widehat{\mathcal{B}}_R \sharp S_n$ -module (the compatibility of the  $S_n$  action can be checked by computation as in [Gan06], Lemma 2.2).

Note that by Remark 6.4.2 the direct sum in Definition 6.4.3 is really a direct sum over a finite subset of indices in I.

For any l = 1, ..., n,  $a \in \overline{Q}$ ,  $\underline{j} \in I^n$  with  $j_l = t(a)$ , we have to define a map  $a'_l|_{\underline{j}} : V'_{\underline{j}} \to V'_{a_l(\underline{j})}$ , where  $a_l(\underline{j})$  is as defined in Section 5.5.2. One has three cases.

**Case I.** If  $h(a), t(a) \neq i$  then  $l \notin \Delta(j) = \Delta(a_l(j))$ . For any  $D \subset \Delta(j)$  we have a map

$$a_l|_{\underline{j},D}: V(\underline{j},D) \to V(a_l(\underline{j}),D), \qquad a_l|_{\underline{j},D}:=\sum_{\xi \in \mathcal{X}(D)} \mu_{a_l(\underline{j}),\xi} a_l|_{t(\underline{j},\xi)} \pi_{\underline{j},\xi}.$$

We define

$$a_l'|_{\underline{j}} := a_l|_{\underline{j},\Delta(\underline{j})}.$$
(6.7)

**Case II.** If t(a) = i, then  $l \in \Delta(\underline{j})$  and  $\Delta(a_l(\underline{j})) = \Delta(\underline{j}) \setminus \{l\}$  since there are no loop edges at i (hence  $h(a) \neq i$ ). If  $l \in D \subset \Delta(\underline{j})$ , for each  $r \in \mathcal{H}$  there is an injective map

$$\tau_{r,l,D}: \mathcal{X}(D \setminus \{l\}) \hookrightarrow \mathcal{X}(D): \eta \to \tau_{r,l,D}(\eta)$$

where

$$\tau_{r,l,D}(\eta)(m) := \begin{cases} \eta(m) & \text{if } m \in D \setminus \{l\} \\ r & \text{if } m = l \end{cases}$$

We observe that  $t(\underline{j}, \tau_{r,l,D}(\eta)) = t(r_l^*(\underline{j}), \eta)$ . Thus there is a projection map

$$\tau^!_{r,l,\underline{j},D}: V(\underline{j},D) \to V(r^*_l(\underline{j}),D \setminus \{l\}), \qquad \tau^!_{r,l,\underline{j},D}:=\sum_{\eta \in \mathcal{X}(D \setminus \{l\})} \mu_{r^*_l(\underline{j}),\eta} \pi_{\underline{j},\tau_{r,l,D}(\eta)},$$

where we used again the fact that there are no loop edges at i and thus, since  $t(r^*) = i$ , we must have  $h(r^*) \neq i$ . Similarly there is an inclusion map

$$\tau_{r,l,\underline{j},D_{l}}: V(r_{l}^{*}(\underline{j}), D \setminus \{l\}) \to V(\underline{j}, D), \qquad \tau_{r,l,\underline{j},D_{l}}:= \sum_{\eta \in \mathcal{X}(D \setminus \{l\})} \mu_{\underline{j},\tau_{r,l,D}(\eta)} \pi_{r_{l}^{*}(\underline{j}),\eta}.$$

We define

$$a'_{l}|_{\underline{j}} := \tau^{!}_{a^{*}, l, \underline{j}, \Delta(\underline{j})}.$$
(6.8)

**Case III.** If h(a) = i, then  $l \notin \Delta(\underline{j})$  and  $\Delta(a_l(\underline{j})) = \Delta(\underline{j}) \cup \{l\}$ . For any  $D \subset \Delta(\underline{j})$  we have the inclusion map

$$\tau_{a,l,a_l(\underline{j}),D\cup\{l\}_1}: V(\underline{j},D) \to V(a_l(\underline{j}),D\cup\{l\})$$

as above. We have a map

$$\theta_{a,l,j,D}: V(\underline{j},D) \to V(a_l(\underline{j}), D \cup \{l\})$$

defined by

$$\theta_{a,l,\underline{j},D} := \left(-\lambda_i + \mu_{a_l(\underline{j}),l}\pi_{a_l(\underline{j}),l} + \nu \sum_{m \in D} s_{ml}|_{a_l(\underline{j})}\right) \tau_{a,l,a_l(\underline{j}),D \cup \{l\}_l}.$$

We define

$$a_l'|_j := \theta_{a,l,j,\Delta(j)}.$$
(6.9)

We have the following proposition

**Proposition 6.4.4.** [[GG05], Proposition 2.7] With the above action  $F_i(V)$  is a  $\mathcal{A}_{R,n,\nu,r_i\lambda}$ -module.

When Q is an infinite affine Dynkin quiver (or in general a locally finite one), the reflection functors satisfy the same properties as in Gan ([Gan06], § 6.2). In particular if i is a loop-free vertex, let  $\Lambda_i$  be the set of all  $(\lambda, \nu) \in \widehat{B}_R \times R = R^I \times R$  such that  $\lambda_i \pm \nu \sum_{m=2}^r s_{1m}$  is invertible in  $R[S_r]$  for all  $r = 1, \ldots, n$ .

**Theorem 6.4.5** ([Gan06], Theorem 5.1). If  $(\lambda, \nu) \in \Lambda_i$ , then the functor

$$F_i: \mathcal{A}_{R,n,\nu,\lambda} - mod \to \mathcal{A}_{R,n,\nu,r_i\lambda} - mod$$

is an equivalence of categories with quasi-inverse functor  $F_i$ , where we are considering the categories of left modules that are finitely generated as R-modules.

**Lemma 6.4.6** ([Gan06], Proposition 5.12). If  $R = \mathbb{C}$  we have:

$$\Lambda_i = \{(\lambda, \nu) \in \widehat{B} \times \mathbb{C} | \lambda_i \pm p\nu \neq 0 \text{ for } p = 0, \dots, n-1\}$$

Let now  $B_J$  and  $U_J$  be the finite dimensional vectors spaces defined at the end of Section 5.6. Let  $R = \mathbb{C}[[U_J]]$ . Let  $\nu \in \mathbb{C}[[U_J]]$ , and  $\lambda \in B_J \otimes \mathbb{C}[[U_J]]$ . With the same notation as in Section 5.6, we say that a  $\mathcal{A}_{\mathbb{C}}[[U_J]], n, \nu, \lambda$ -module  $V_{U_J}$  is a flat formal deformation of a  $\overline{\mathcal{A}_{\mathbb{C}}[[U_J]], n, \nu, \lambda}$ -module V if  $V_{U_J} \cong V[[U_J]]$  as  $\mathbb{C}[[U_J]]$ -modules and  $\overline{V_{U_J}} \cong V$ . Proposition 5.14 of [Gan06] extend to the continuous case with analogous proof.

**Proposition 6.4.7.** Assume that  $\nu \in \mathbb{C}[U_J]$ , and  $\lambda \in B_J \otimes \mathbb{C}[U_J]$  (as in Lemma 5.6.2). Moreover, let  $i \in J$  and assume that  $(\lambda, \nu) \in \Lambda_i$ . If a  $\mathcal{A}_{\mathbb{C}[[U_J]],n,\nu,\lambda}$ -module  $V_{U_J}$  is a flat formal deformation of a finite dimensional  $\overline{\mathcal{A}_{\mathbb{C}[[U_J]],n,\nu,\lambda}}$ -module V, then  $F_i(V_{U_J})$  is a flat formal deformation of  $F_i(V)$ .

#### 6.5 Representations in the higher rank case

For any commutative unital  $\mathbb{C}$ -algebra R and any  $\widehat{\mathcal{B}}_R \sharp S_n$ -module V define the set  $I_V \subset I$  as follows

$$I_V := \left\{ i \in I | \exists \underline{j} := (j_1, \dots, j_n) \in I^n \text{ with } i \in \{j_1, \dots, j_n\} \text{ and } | \underline{j} V \neq 0 \right\}.$$

where we recall that  $|_{\underline{j}} = e_{j_1} \otimes \cdots \otimes e_{j_n}$ .

It is clear that if V is finitely generated over R then  $I_V$  is a finite subset of I (see Remark 6.4.2).

In particular when n = 1,  $R = \mathbb{C}$  and V is finite dimensional, of dimension vector say  $\alpha$ , then  $I_V = \{i \in I | e_i V \neq 0\} = \operatorname{supp}(\alpha)$ , the support of the dimension vector.

Suppose now  $I_V$  is finite, and let  $J \subset I$  be any finite subset of indices such that  $I_V \subset J$ . Consider the finite rank free *R*-module

$$B_{R,J} := \bigoplus_{i \in J} Re_i \hookrightarrow \prod_{i \in I} Re_i = \widehat{B}_R$$

i.e.  $B_{R,J} = \left\{ \lambda \in \widehat{B}_R | \lambda_i = 0 \ \forall i \notin J \right\}$ . Then the module V factors through the homomorphism  $(\widehat{\mathcal{B}}_R \sharp S_n) \twoheadrightarrow B_{R,J}^{\otimes n} \sharp S_n$  defined by

$$\begin{cases} \sigma \longrightarrow \sigma & \text{if } \sigma \in S_n \\ |_{\underline{j}} \longrightarrow |_{\underline{j}} & \text{if } j_s \in J \text{ for all } s \\ |_{\underline{j}} \longrightarrow 0 & \text{otherwise} \end{cases}$$
(6.10)

In accordance with the notation of Section 6.2, in all what follows, for any finite set of indices J and any parameter  $\lambda \in \widehat{B}_R$ , we will denote by  $\lambda|_J \in B_{R,J}$  the restriction of the parameter  $\lambda$  to the set J.

Suppose now V is a  $\mathcal{A}_{R,n,\nu,\lambda}$ -module and  $I_V \subset J$ . Let  $Q_J$  be the full subquiver corresponding to J. Then V factors through the homomorphism  $\mathcal{A}_{R,n,\nu,\lambda} \longrightarrow$   $\mathcal{A}_{R,n,\nu,\lambda|_J}(Q_J)$  defined by formulas (6.10) above and by the assignment

$$\begin{cases} a_l|_{\underline{j}} \longrightarrow a_l|_{\underline{j}} & \text{if } j_s \in J \text{ for all } s \\ a_l|_{\underline{j}} \longrightarrow 0 & \text{otherwise} \end{cases}$$

where we recall that, for any  $a \in \overline{Q}$  and l = 1, ..., n, we denoted by  $a_l|_{\underline{j}}$  the element  $e_{j_1} \otimes \cdots \otimes ae_{j_l} \otimes \cdots \otimes e_{j_n}$ .

Observe that what we said above also implies that the module V factors through an action of the algebra  $\mathcal{A}_{R,n,\nu,\lambda_J}$ , where  $\lambda_J$  is as in Section 5.4 (i.e.  $(\lambda_J)_i = \lambda_i$  if  $i \in J$  and  $(\lambda_J)_i = 0$  otherwise).

Thus, if we want to study modules over  $\mathcal{A}_{R,n,\nu,\lambda}(Q)$  that are finitely generated as *R*-modules, it is actually enough to consider subfamilies of algebras depending only on finitely many of the parameters  $\lambda_i$ .

Let us now go back to the case  $R = \mathbb{C}$ . The easiest examples of finite dimensional  $\mathcal{A}_{n,\nu,\lambda}$ - modules one can look for are the modules which are irreducible as  $\widehat{\mathcal{B}} \sharp S_n$ modules. For any  $i \in I$ , let  $\mathcal{N}_i$  be the complex vector space with dimension vector  $\epsilon_i$ . Consider  $\vec{n} = (n_1, \ldots, n_r)$ , where  $n_i \in \mathbb{Z}_{>0}$ , and  $\sum_{i=1}^r n_i = n$ . Let  $\{i_1, \ldots, i_r\}$  be a set
of r distinct vertices of Q and let  $\mathcal{N} = \mathcal{N}_{i_1}^{\otimes n_1} \otimes \cdots \otimes \mathcal{N}_{i_r}^{\otimes n_r}$ . As in Chapter 3, regard  $S_{n_j}$ as the group permuting the factors in  $\mathcal{N}_{i_j}^{\otimes n_j}$ , and consider  $S_{\vec{n}} := S_{n_1} \times \cdots \times S_{n_r} \subset S_n$ . Let  $X = X_1 \otimes \cdots \otimes X_r$  be a simple module for the group  $S_{\vec{n}}$ . Then  $X \otimes \mathcal{N}$  is a
simple module for  $\widehat{\mathcal{B}} \sharp S_{\vec{n}}$ . One can then form the induced  $\widehat{\mathcal{B}} \sharp S_n$ -module  $X \otimes \mathcal{N} \uparrow :=$   $\mathrm{Ind}_{\widehat{\mathcal{B}} \sharp S_n}^{\widehat{\mathcal{B}} \Im_n}(X \otimes \mathcal{N})$ . Moreover, it is known that any simple finite dimensional  $\widehat{\mathcal{B}} \sharp S_n$ module has this form (this is true by [Mac80], paragraph after (A5), when  $\Gamma$  is
finite, and remains true for  $\Gamma$  reductive when we consider only finite dimensional
representations).

Observe now that we have

$$X \otimes \mathcal{N} \uparrow = \bigoplus_{\sigma} \sigma(X \otimes \mathcal{N}) \tag{6.11}$$

where  $\sigma$  runs over a set of left coset representative of  $S_{\vec{n}}$  in  $S_n$  (in particular we have  $I_{X \otimes N\uparrow} = \{i_1, \ldots, i_r\}$ ).

The following lemma is the analog of [Gan06], Lemma 6.1.

**Lemma 6.5.1.** Suppose  $\nu \in \mathbb{C}[[U_J]]$ , and  $\lambda \in B_J \otimes \mathbb{C}[[U_J]]$ . Let a  $\mathcal{A}_{\mathbb{C}}[[U_J]], n, \nu, \lambda}$ -module  $V_{U_J}$  be a flat formal deformation of the finite dimensional  $\overline{\mathcal{A}_{\mathbb{C}}[[U_J]]}, n, \nu, \lambda}$ -module  $\overline{V_{U_J}}$ . If  $\overline{V_{U_J}}$  is simple as a  $\widehat{\mathcal{B}} \sharp S_n$ -module, then all elements of  $\widehat{\mathcal{E}}_{\mathbb{C}}[[U_J]]$  must act by 0 on  $V_{U_J}$ .

Proof. The proof goes exactly as in [Gan06] when we observe that also in our case  $\widehat{\mathcal{B}} \sharp S_n$  is a semisimple algebra. This implies that  $V_{U_J}$  must be of the form  $(X \otimes \mathcal{N} \uparrow)$  $)[[U_J]]$  as a  $\widehat{\mathcal{B}}_{\mathbb{C}[[U_J]]} \sharp S_n$ -module, with  $i_1, \ldots, i_r \in J$  (trivial deformation of an irreducible module  $X \otimes \mathcal{N} \uparrow$ ). Let us fix  $l \in \{1, \ldots, n\}$ . Any  $a_l|_{\underline{j}} \in \widehat{\mathcal{B}}_{\mathbb{C}[[U_J]]} \sharp S_n$  induces maps between the different summands in formula (6.11). Now, for any  $\underline{j} = (j_1, \ldots, j_n) \in I^n$  and  $\sigma \in S_n$ , if  $j_h = j_{\sigma(h)}$  for any  $h \neq l$  then  $j_l = j_{\sigma(l)}$ . Since Q has no edge loops, this implies that  $a_l|_{\underline{j}}$  must act by 0 on  $\mathcal{N}$ . Since this is true for any a and any l, all elements of  $\widehat{\mathcal{E}}_{\mathbb{C}[[U_J]]}$  must act by 0 on  $V_{U_J}$ .

The following theorem can be proved as Theorem 6.2 in [Gan06] and is equivalent to Theorem 6.5 in [EGG05].

**Theorem 6.5.2.** Assume  $\mathbb{C}[[U_J]] \ni \nu \neq 0$  and  $\lambda \in B_J \otimes \mathbb{C}[[U_J]]$ . The  $\widehat{\mathcal{B}}_{\mathbb{C}[[U_J]]} \sharp S_n$ module  $(X \otimes \mathcal{N} \uparrow)[[U_J]]$  extends to a  $\mathcal{A}_{\mathbb{C}[[U_J]],n,\nu,\lambda_J}$ -module if and only if the following conditions are satisfied:

- (i) For all  $l \in \{1, ..., r\}$ , the simple module  $X_l$  of  $S_{n_l}$  has rectangular Young diagram of some size  $a_l \times b_l$ ;
- (ii) No two vertices in the collection  $\{i_1, \ldots, i_r\}$  are adjacent in Q, i.e.  $(\epsilon_{i_j}, \epsilon_{i_k}) = 0$ for any  $j \neq k \in \{1, \ldots, r\}$ ;
- (iii) For all  $l \in \{1, \ldots, r\}$ , one has  $\lambda_{i_l} = \nu(a_l b_l)$ ;

where we agree that condition (ii) is empty if r = 1 i.e. if  $\vec{n} = (n)$ .

Fix now  $\lambda_0 \in \widehat{B}$ . Let  $Y_1, \ldots, Y_r$  be a collection of simple, pairwise non-isomorphic, finite dimensional representations of  $\widehat{\Pi}_{\lambda_0}$  and denote by  $\alpha_{(i)}$  the dimension vector of  $Y_i$ . Let  $\mathbf{Y} := Y_1^{\otimes n_1} \otimes \cdots \otimes Y_r^{\otimes n_r}$ . Then  $X \otimes \mathbf{Y}$  is an irreducible representation of  $\widehat{\Pi}_{\lambda_0}^{\otimes n} \sharp S_{\vec{n}}$  and, as in the finite case, we can consider the induced  $\mathcal{A}_{n,0,\lambda_0}$ -module  $X \otimes \mathbf{Y} \uparrow := \operatorname{Ind}_{\hat{\Pi}_{\lambda_0}^{\otimes n} \sharp S_n}^{\hat{\Pi}_{\lambda_0}^{\otimes n} \sharp S_n} X \otimes \mathbf{Y}$ . It is known (as before by [Mac80]) that any finite dimensional simple  $\mathcal{A}_{n,0,\lambda_0}$ -module is of this form. For  $i = 1, \ldots, r$ , denote by  $\alpha_{(i)}$  the dimension vector of  $Y_i$ . It is easy to see that:

$$I_{X\otimes\mathbf{Y}\uparrow} = \bigcup_{i=1}^r I_{Y_i} = \bigcup_{i=1}^r \operatorname{supp}(\alpha_{(i)}).$$

Let J be a finite subset of indices corresponding to some connected subquiver and such that  $I_{X\otimes \mathbf{Y}\uparrow} \subset J$ . As in Section 5.4, let  $\lambda^+$  be the unique J-dominant weight  $W_{J}$ conjugate to  $\lambda_0$ , and  $w^+$  the unique element of minimal length such that  $w^+\lambda_0 = \lambda^+$ . Write  $w^+ = s_{j_h} \cdots s_{j_1}$  for some simple reflections corresponding to vertices  $j_1, \ldots, j_h \in$ J. By minimality of the length we know  $(r_{j_g} \cdots r_{j_1}\lambda)_{j_{g+1}} \neq 0$  for  $g = 0, \ldots, h - 1$ . Denote by  $F_{w^+}$  the composition  $F_{j_h} \cdots F_{j_1}$ , and by  $F_{(w^+)^{-1}}$  the composition  $F_{j_1} \cdots F_{j_h}$ .

The following theorem is the analog of Theorem 6.3 in [Gan06].

**Theorem 6.5.3.** Suppose  $\lambda \in B_J \otimes \mathbb{C}[U_J]$  with  $\lambda_i \in U_J$  for any  $i \in J$ , and that  $0 \neq \nu \in U_J$ . The  $\mathcal{A}_{n,0,\lambda_0}$ -module  $X \otimes \mathbf{Y} \uparrow$  has a flat formal deformation to a  $\mathcal{A}_{\mathbb{C}[[U_J]],n,\nu,\lambda_0+\lambda}$ -module if and only if the following conditions are satisfied:

- (i) For all  $l \in \{1, ..., r\}$ , the simple module  $X_l$  of  $S_{n_l}$  has rectangular Young diagram, of size  $a_l \times b_l$ ;
- (*ii*)  $(\alpha_{(l)}, \alpha_{(m)}) = 0$  for all  $l \neq m$ ;
- (iii) For all  $l \in \{1, \ldots, r\}$ , one has  $\lambda \cdot \alpha_{(l)} = (a_l b_l)\nu$ ;

where we agree that condition (ii) is empty if r = 1. When the deformation exists it is unique.

Proof. Take  $J, \lambda^+, w^+, F_{w^+}, F_{(w^+)^{-1}}$  as above. Exactly as in [Gan06], proof of Theorem 6.3, we can prove that  $(r_{j_g} \cdots r_{j_1}(\lambda_0 + \lambda), \nu) \in \Lambda_{j_{g+1}}$  for  $g = 0, \ldots, h -$ 1. Indeed, for any  $C \in \mathbb{C}[[U_J]][S_n]$  the element  $(r_{j_g} \cdots r_{j_1}(\lambda_0 + \lambda))_{j_{g+1}} + \nu C$  has an inverse in  $\mathbb{C}[[U_J]][S_n]$ . Finding this inverse amounts to solving a system of n!equations in n! variables with matrix  $(r_{j_g} \cdots r_{j_1}(\lambda_0 + \lambda))_{j_{g+1}} \mathrm{Id}_{n!} + \nu A$ , with A some matrix. Since  $\lambda, \nu \in \mathfrak{m}$  and  $(r_{j_g} \cdots r_{j_1} \lambda_0)_{j_{g+1}} \neq 0$  (by minimality of the length of  $w^+$ ), the determinant of this matrix is nonzero modulo  $\mathfrak{m}$ . Thus the determinant is invertible in  $\mathbb{C}[[U_J]]$  and so the matrix is invertible. As a consequence we have that the sequence  $F_{w^+}$  establishes an equivalence of categories  $\mathcal{A}_{\mathbb{C}}[[U_J]], n, \nu, \lambda_0 + \lambda - \mathrm{mod} \longrightarrow \mathcal{A}_{\mathbb{C}}[[U_J]], n, \nu, \lambda^+ + w^+(\lambda) - \mathrm{mod}.$ 

For any  $l = 1, \ldots, r$  we have that  $\alpha_{(l)} \in \Sigma_{\lambda_0,J}$ . From Lemma 5.4.1 we know that  $w^+ \Sigma_{\lambda_0,J} = \Sigma_{\lambda^+,J}$ . Since  $\lambda^+$  is *J*-dominant we know that  $\Sigma_{\lambda^+,J} = \{\epsilon_i | i \in J, \lambda_i^+ = 0\}$ . For any  $l = 1, \ldots r$ , define  $i_l \in J$  by  $w^+(\alpha_{(l)}) = \epsilon_{i_l} \in \Sigma_{\lambda^+,J}$ . As we observed at the beginning of this section, since  $I_{X \otimes \mathbf{Y}\uparrow} \subset J$  the  $\mathcal{A}_{n,0,\lambda_0}$ -module  $X \otimes \mathbf{Y}\uparrow$  factors through  $\mathcal{A}_{n,0,(\lambda_0)|_J}(Q_J)$ . Moreover, since  $j_1, \cdots, j_h \in J$ , we can deduce from the definition of the functors  $F_{j_l}$  that the  $\mathcal{A}_{n,0,\lambda^+}$ -module  $F_{w^+}(X \otimes \mathbf{Y}\uparrow)$  factors through  $\mathcal{A}_{n,0,(\lambda^+)|_J}(Q_J)$ . By [CBH98] Theorem 5.1, we must have  $F_{w^+}(X \otimes \mathbf{Y}\uparrow) = X \otimes \mathcal{N}\uparrow$  where  $\mathcal{N} = \mathcal{N}_{i_1}^{\otimes n_1} \otimes \cdots \otimes \mathcal{N}_{i_r}^{\otimes n_r}$ .

From now the proof goes exactly as in [Gan06] and we report it for the reader's convenience.

We first observe that, for any l = 1, ..., r, we have  $\lambda_0 \cdot \alpha_{(l)} = \lambda^+ \cdot \epsilon_{i_l} = 0$ , and that both the products (, ) and  $\cdot$  are *W*-invariant.

Using this, we have that, if the conditions in the theorem are satisfied, then the  $\widehat{\mathcal{B}}_{\mathbb{C}[[U_J]]} \sharp S_n$ -module  $M := X \otimes \mathcal{N} \uparrow [[U_J]]$  satisfies the conditions of Theorem 6.5.2, and thus extends to a  $\mathcal{A}_{\mathbb{C}[[U_J]],n,\nu,\lambda^++w^+(\lambda)}$ -module (with  $\widehat{\mathcal{E}}_{\mathbb{C}[[U_J]]}$  acting by 0). Thus by Proposition 6.4.7 the  $\mathcal{A}_{\mathbb{C}[[U_J]],n,\nu,\lambda_0+\lambda}$ -module  $F_{(w^+)^{-1}}(M)$  is a flat formal deformation of  $X \otimes \mathbf{Y} \uparrow$ .

Vice versa suppose a  $\mathcal{A}_{\mathbb{C}[[U_J]],n,\nu,\lambda_0+\lambda}$ -module V is a flat formal deformation of  $X \otimes \mathbf{Y} \uparrow$ . Then, using again Proposition 6.4.7, we have that  $F_{w^+}(V)$  is a flat formal deformation of  $X \otimes \mathcal{N} \uparrow$ . Since  $F_{w^+}(V) = (X \otimes \mathcal{N} \uparrow)[[U_J]]$  as a  $\widehat{\mathcal{B}}_{\mathbb{C}[[U_J]]} \sharp S_n$ -module, then by Theorem 6.5.2 the conditions (i), (ii), (iii) must be satisfied. Moreover, we have that, by Lemma 6.5.1,  $\widehat{\mathcal{E}}_{\mathbb{C}[[U_J]]}$  must act by 0. Thus  $F_{w^+}(V)$  is the unique flat formal deformation of  $X \otimes \mathcal{N} \uparrow$  and V is the unique flat formal deformation of  $X \otimes \mathcal{N} \uparrow$ .

Assume now that  $\lambda_0 \in B_J$  and that the conditions (i), (ii) of Theorem 6.5.3 hold. Let  $\nu \in \mathbb{C}[U_J]$  and  $\lambda \in B_J \otimes \mathbb{C}[U_J]$  be functions satisfying the condition (iii). Suppose there is a point  $o \in U_J$  such that  $\lambda$  specializes to  $\lambda_0$  and  $\nu$  specializes to 0 at o. Following Gan's notation let  $U'_J$  be the Zariski open set in  $U_J$  defined by  $(r_{j_g} \cdots r_{j_1} \lambda)_{j_{g+1}} \pm p\nu \neq 0$  for  $g = 0, \ldots, h - 1$  and  $p = 0, \ldots, n - 1$ . Since  $o \in U'_J$  this set is nonempty. Let  $\mathbb{C}[U'_J]$  be the ring of regular functions on  $U'_J$  and for any  $u \in U'_J$ let  $\mathfrak{m}_u$  denote the maximal ideal of functions vanishing at u. If V is a  $\mathbb{C}[U'_J]$ -module let  $V^u := V/\mathfrak{m}_u V$ . The following theorem is the analog of [Gan06], Theorem 6.4 and the proof is exactly the same.

**Theorem 6.5.4.** There exists a  $\mathcal{A}_{\mathbb{C}[U'_j],n,\nu,\lambda}$ -module  $V_{U'_j}$  such that:

- (i)  $V_{U'_{I}}^{o} = X \otimes \mathbf{Y} \uparrow as \ a \ \mathcal{A}_{n,0,\lambda_{0}}$ -module, and  $V_{U'_{J}}$  is flat over  $U'_{J}$ ;
- (ii) for any point  $u \in U'_J$ ,  $V^u_{U'_J}$  is a finite dimensional simple  $\mathcal{A}^u_{\mathbb{C}[U'_J],n,\nu,\lambda}$ -module, isomorphic to  $X \otimes \mathbf{Y} \uparrow$  as a  $\widehat{\mathcal{B}}^u_{\mathbb{C}[U'_J]} \sharp S_n$ -module.

# Appendix A

### Proof of Theorem 5.6.1

Before getting started we need to introduce some notation. For any  $i \in I$ , let us choose a basis for the irreducible representation  $N_i$ , and let us denote by  $E_{pq}^{N_i}(\gamma) \in \mathcal{O}(\Gamma)$  the (p,q)-th matrix coefficient for  $N_i$  in such basis, where  $1 \leq p, q \leq d_i = \dim N_i$ . Since  $\Gamma$  is reductive, we know these functions span the algebra  $\mathcal{O}(\Gamma)$ . Moreover, if we take matrix coefficients  $E_{pq}^{N_i}(\gamma)$ ,  $E_{lm}^{N_j^*}(\gamma)$  with respect to dual bases, the following orthogonality relation holds:

$$\int_{\Gamma} E_{pq}^{N_i}(\gamma) E_{lm}^{N_j^*}(\gamma) d\gamma = \frac{1}{d_i} \delta_{ij} \delta_{pl} \delta_{qm} , \qquad (A.1)$$

where  $\int_{\Gamma}$  is the linear form described in Section 4.2. Let  $\check{E}_{pq}^{N_i}$  be the unique distribution such that  $\langle \check{E}_{pq}^{N_i}, E_{lm}^{N_j} \rangle = \delta_{ij} \delta_{pl} \delta_{qm}$ . Using (A.1) we can write  $\langle \check{E}_{pq}^{N_i}, E_{lm}^{N_j} \rangle = d_i \int_{\Gamma} E_{lm}^{N_j} E_{pq}^{N_i^*} d\gamma$ . Using the identifications (4.3) of Section 4.2, we can see how these distributions span topologically  $\mathcal{O}(\Gamma)^*$ .

It is straightforward to compute that, if  $\Delta : \mathcal{O}(\Gamma) \longrightarrow \mathcal{O}(\Gamma) \otimes \mathcal{O}(\Gamma)$  denotes the coproduct for  $\mathcal{O}(\Gamma)$ , then  $\Delta(E_{pq}^{N_i}) = \sum_{r=1}^{d_i} E_{pr}^{N_i} \otimes E_{rq}^{N_i}$ . In all what follows, when there is no ambiguity, we will just omit the sum sign over repeated indices and write  $\Delta(E_{pq}^{N_i}) = E_{pr}^{N_i} \otimes E_{rq}^{N_i}$ . Using just the definition of convolution product (formula (4.1), Section 4.2), it is now easy to see that  $\check{E}_{pq}^{N_i}\check{E}_{lm}^{N_j} = \delta_{ij}\delta_{ql}\check{E}_{pm}^{N_i}$ , so that the identification  $\mathcal{O}(\Gamma)^* = \prod_{i \in I} \operatorname{Mat}(d_i)$  is an algebra isomorphism.

We observe now that  $V \otimes \mathcal{O}(\Gamma)^{*\hat{\otimes}n}$  is a  $\mathcal{O}(\Gamma)^{*\hat{\otimes}n}$ -bimodule with right action only on the second factor and left action defined by

$$\mu(w \otimes \mu') = \sum_{i} \left( x_i \otimes (x_i^*, gw) + y_i \otimes (y_i^*, gw) \right) \mu \mu'$$

for all  $\mu, \mu' \in \mathcal{O}(\Gamma)^{*\hat{\otimes}n}$ ,  $w \in V$ , where  $\{x_i, y_i\}$  is a symplectic basis as in Lemma 4.4.1

and  $\{x_i^*, y_i^*\}$  is its dual basis. Let us now denote by

$$T^{k}_{\mathcal{O}(\Gamma)^{*\hat{\otimes}n}}(V \otimes \mathcal{O}(\Gamma)^{*\hat{\otimes}n}) := (V \otimes \mathcal{O}(\Gamma)^{*\hat{\otimes}n}) \otimes_{\mathcal{O}(\Gamma)^{*\hat{\otimes}n}} \cdots \otimes_{\mathcal{O}(\Gamma)^{*\hat{\otimes}n}} (V \otimes \mathcal{O}(\Gamma)^{*\hat{\otimes}n}).$$

We have

$$TV \sharp \mathcal{O}(\Gamma_n)^* = \left( \bigoplus_{k \ge 0} T^k_{\mathcal{O}(\Gamma)^{*\hat{\otimes}n}}(V \otimes \mathcal{O}(\Gamma)^{*\hat{\otimes}n}) \right) \sharp S_n = T_{\mathcal{O}(\Gamma)^{*\hat{\otimes}n}}(V \otimes \mathcal{O}(\Gamma)^{*\hat{\otimes}n}) \sharp S_n.$$
(A.2)

Following [CBH98] (§ 3) and [GG05] (§ 3.3) we will now define the idempotents  $\varphi_i := \check{E}_{11}^{N_i}$  and  $\varphi = \sum_i \varphi_i$  in the algebra  $\mathcal{O}(\Gamma)^*$ .

For the element element  $\varphi^{\otimes n} \in \mathcal{O}(\Gamma)^{*\hat{\otimes}n}$  we have:

$$\varphi^{\otimes n} = \sum_{i_1,\dots,i_n \in I} \varphi_{i_1} \otimes \dots \otimes \varphi_{i_n}$$

and

$$\sum_{i_{1},p_{1},\dots,i_{n},p_{n}} (\check{E}_{p_{1}1}^{N_{i_{1}}} \otimes \dots \otimes \check{E}_{p_{n}1}^{N_{i_{n}}}) \varphi^{\otimes n} (\check{E}_{1p_{1}}^{N_{i_{1}}} \otimes \dots \otimes \check{E}_{1p_{n}}^{N_{i_{n}}})$$
$$= \sum_{i_{1},p_{1},\dots,i_{n},p_{n}} \check{E}_{p_{1}1}^{N_{i_{1}}} \check{E}_{1p_{1}}^{N_{i_{1}}} \otimes \dots \otimes \check{E}_{p_{n}1}^{N_{i_{n}}} \check{E}_{1p_{n}}^{N_{i_{n}}} = \delta_{1}^{\otimes n} .$$
(A.3)

Since  $\delta_1^{\otimes n}$  is the unit element in  $\mathcal{O}(\Gamma)^{*\hat{\otimes}n}$ , equation (A.3) implies a Morita equivalence  $\varphi^{\otimes n}\left(TV \sharp \mathcal{O}(\Gamma)^{*\hat{\otimes}n}\right) \varphi^{\otimes n} \sim TV \sharp \mathcal{O}(\Gamma)^{*\hat{\otimes}n}$ .

Now clearly we have an isomorphism

$$\widehat{\mathcal{B}} \xrightarrow{\sim} \varphi^{\otimes n} \mathcal{O}(\Gamma)^{*\widehat{\otimes}n} \varphi^{\otimes n} = \prod_{i_1, \dots, i_n} \mathbb{C} \cdot \varphi_{i_1} \otimes \dots \otimes \varphi_{i_n}$$
(A.4)

such that

 $e_{i_1} \otimes \cdots \otimes e_{i_n} \longrightarrow \varphi_{i_1} \otimes \cdots \otimes \varphi_{i_n}$ .

Moreover we have bijections

$$\varphi_i \mathcal{O}(\Gamma)^* \varphi_j \leftrightarrow \operatorname{Hom}_{\Gamma}(N_i, N_j) \qquad \varphi_i \left( L \otimes \mathcal{O}(\Gamma)^* \right) \varphi_j \leftrightarrow \operatorname{Hom}_{\Gamma}(N_i, L \otimes N_j).$$

Indeed we have that  $\mathcal{O}(\Gamma)^* \varphi_j \cong N_j$  and  $(L \otimes \mathcal{O}(\Gamma)^*) \varphi_j \cong L \otimes N_j$ . The first is an irreducible finite dimensional representation and, since  $\Gamma$  is reductive, the second is a semisimple representation. Multiplying on the left by the idempotent  $\varphi_i$  corresponds to projecting on the multiplicity space of the component of type  $N_i$  of such representations.

Thus we have an isomorphism

$$\varphi^{\otimes n} \left( V \otimes \mathcal{O}(\Gamma)^{* \hat{\otimes} n} \right) \varphi^{\otimes n}$$

$$= \prod_{i_1, \dots, j_n} (\varphi_{i_1} \otimes \dots \otimes \varphi_{i_n}) (L^{\oplus n} \otimes \underbrace{\mathcal{O}(\Gamma)^* \hat{\otimes} \dots \hat{\otimes} \mathcal{O}(\Gamma)^*}_n) (\varphi_{j_1} \otimes \dots \otimes \varphi_{j_n})$$

$$= \bigoplus_{l=1}^n \prod_{i_1, \dots, j_n} \operatorname{Hom}_{\Gamma}(N_{i_1}, N_{j_1}) \otimes \dots \otimes \operatorname{Hom}_{\Gamma}(N_{i_l}, L \otimes N_{j_l}) \otimes \dots \otimes \operatorname{Hom}_{\Gamma}(N_{i_n}, N_{j_n})$$

$$\cong \widehat{\mathcal{E}}$$
(A.5)

Now from (A.2) and from identities (A.4) and (A.5) it follows that:

$$\varphi^{\otimes n}\left(T_{\mathcal{O}(\Gamma)^{*\hat{\otimes}n}}\left(V\otimes\mathcal{O}(\Gamma)^{*\hat{\otimes}n}\right)\right)\varphi^{\otimes n}\cong\widehat{T_{\mathcal{B}}\mathcal{E}}$$
(A.6)

and

$$\varphi^{\otimes n} \left( TV \sharp \mathcal{O}(\Gamma)^{* \hat{\otimes} n} \right) \varphi^{\otimes n} = \varphi^{\otimes n} \left( T_{\mathcal{O}(\Gamma)^{* \hat{\otimes} n}} \left( V \otimes \mathcal{O}(\Gamma)^{* \hat{\otimes} n} \right) \sharp S_n \right) \varphi^{\otimes n}$$
$$\cong \widehat{T_{\mathcal{B}} \mathcal{E}} \sharp S_n \tag{A.7}$$

Now by (A.3) we have that  $\mathcal{H}_{k,c}(\Gamma_n)$  is Morita equivalent to  $\varphi^{\otimes n}\mathcal{H}_{k,c}(\Gamma_n)\varphi^{\otimes n}$ . By (A.7) we have that  $\varphi^{\otimes n}\mathcal{H}_{k,c}(\Gamma_n)\varphi^{\otimes n}$  is isomorphic to some quotient of  $\widehat{T}_{\mathcal{B}}\mathcal{E}\sharp S_n$ . We will show in the next theorem that for an appropriate choice of the parameters this quotient is exactly the one described in Definition 5.5.3.

We will need the following auxiliary lemma, which is the analog for infinite affine quivers of Lemma 3.2 of [CBH98]. Let Q be a quiver attached to  $\Gamma$  (with any orientation). Let  $\zeta$  be the linear map  $\zeta : \mathbb{C} \to L \otimes L$  such that  $1 \to y \otimes x - x \otimes y$ .

**Lemma A.0.5.** For any arrow  $a: i \to j$  in Q there exist  $\Gamma$ -module homomorphisms

$$\theta_a: N_i \to L \otimes N_j \quad and \quad \phi_a: N_j \to L \otimes N_i$$

such that for any vertex i

$$\sum_{a \in Q, h(a)=i} (\mathrm{Id}_L \otimes \theta_a) \phi_a - \sum_{a \in Q, t(a)=i} (\mathrm{Id}_L \otimes \phi_a) \theta_a = -d_i (\zeta \otimes \mathrm{Id}_{N_i})$$

as maps from  $N_i$  to  $L \otimes L \otimes N_i$ , and such that

$$(\omega_L \otimes \mathrm{Id}_{N_{t(a)}})(\mathrm{Id}_L \otimes \phi_a)\theta_a = -d_{h(a)}\mathrm{Id}_{N_{t(a)}}$$

and

$$(\omega_L \otimes \mathrm{Id}_{N_{h(a)}})(\mathrm{Id}_L \otimes \theta_a)\phi_a = d_{t(a)}\mathrm{Id}_{N_{h(a)}}$$

Moreover the  $\theta_a$ ,  $\phi_a$  combine to give a basis for each of the spaces  $\operatorname{Hom}_{\Gamma}(N_i, L \otimes N_j)$ .

*Proof.* (of Theorem 5.6.1) In the case  $\Gamma$  is of type  $A_{\infty}$  the same proof as in [CBH98],

Lemma 3.2, works without any change. For  $\Gamma$  of type  $A_{+\infty}$ ,  $D_{\infty}$  the proof goes as the one in [CBH98] for type  $\tilde{D}_n, \tilde{E}_n$  if we observe that also in our case Q is a (semiinfinite) tree, the  $L \otimes N_i$  are all multiplicity free and the vector  $\delta$  is the unique vector in  $\mathbb{C}^I$ , up to scalar multiples, such that  $(\delta, \epsilon_i) = 0$  for all  $i \in I$ .

Proof. We give a proof for  $n \ge 2$  since the proof for n = 1 is similar and easier. Our proof rephrases the proof of Theorem 3.5.2 in [GG05] in the language of algebraic distributions on  $\Gamma_n$ . Using equations (A.4)-(A.7) and Lemma A.0.5 we can define an isomorphism

$$\widehat{T_{\mathcal{B}}\mathcal{E}} \sharp S_n \to \varphi^{\otimes n} \left( TV \sharp \mathcal{O}(\Gamma_n)^* \right) \varphi^{\otimes n}$$

$$e_{i_1} \otimes \cdots \otimes e_{i_n} \cdot \sigma \to \varphi_{i_1} \otimes \cdots \otimes \varphi_{i_n} \cdot \sigma,$$

$$e_{i_1} \otimes \cdots \otimes a \otimes \cdots \otimes e_{i_n} \cdot \sigma \to \varphi_{i_1} \otimes \cdots \otimes \phi_a \otimes \cdots \otimes \varphi_{i_n} \cdot \sigma,$$

$$e_{i_1} \otimes \cdots \otimes a^* \otimes \cdots \otimes e_{i_n} \cdot \sigma \to \varphi_{i_1} \otimes \cdots \otimes \theta_a \otimes \cdots \otimes \varphi_{i_n} \cdot \sigma,$$

for all  $i_1, \ldots, i_n \in I$ ,  $a \in Q$ .

Let us denote by J the subspace of  $TV \sharp \mathcal{O}(\Gamma_n)^*$  topologically spanned by relations (R1), (R2) of Definition 4.4.1. Then the algebra  $\mathcal{H}_{k,c}(\Gamma_n)$  is the quotient of  $TV \sharp \mathcal{O}(\Gamma_n)^*$  by the two-sided ideal generated by J and  $\varphi^{\otimes n} \mathcal{H}_{k,c}(\Gamma_n) \varphi^{\otimes n}$  is the quotient of  $\varphi^{\otimes n} TV \sharp \mathcal{O}(\Gamma_n)^* \varphi^{\otimes n}$  by the ideal

$$\varphi^{\otimes n} \left( TV \sharp \mathcal{O}(\Gamma_n)^* \right) J \left( TV \sharp \mathcal{O}(\Gamma_n)^* \right) \varphi^{\otimes n}$$
  
=  $\varphi^{\otimes n} \left( TV \sharp \mathcal{O}(\Gamma_n)^* \right) \varphi^{\otimes n} \mathcal{O}(\Gamma)^{*\hat{\otimes} n} J \mathcal{O}(\Gamma)^{*\hat{\otimes} n} \varphi^{\otimes n} \left( TV \sharp \mathcal{O}(\Gamma_n)^* \right) \varphi^{\otimes n}$ 

where the identity holds by equation (A.3). Our claim is that the image of the two sided ideal generated by  $\varphi^{\otimes n} \mathcal{O}(\Gamma)^{*\hat{\otimes}n} J \mathcal{O}(\Gamma)^{*\hat{\otimes}n} \varphi^{\otimes n}$  under the above isomorphism is exactly the ideal of the defining relations (I), (II) for  $\mathcal{A}_{n,\nu,\lambda}(Q)$ .

Let us first consider the relations (R1). Notice that for any  $\varphi \in \mathcal{O}(\Gamma)^*$  and  $x, y \in L$ a symplectic basis, we have that in  $TL \sharp \mathcal{O}(\Gamma)^*$ 

$$\varphi(xy - yx) = (xy - yx)\varphi.$$
 (A.8)

In fact

$$\begin{split} \varphi(xy - yx) \\ &= (x(x^*, \gamma x)\varphi + y(y^*, \gamma x)\varphi) y - (y(y^*, \gamma y)\varphi + x(x^*, \gamma y)\varphi) x \\ &= xx(x^*, \gamma y)(x^*, \gamma x)\varphi + xy(y^*, \gamma y)(x^*, \gamma x)\varphi + yx(x^*, \gamma y)(y^*, \gamma x)\varphi + yy(y^*, \gamma y)(y^*, \gamma x)\varphi + yx(x^*, \gamma x)(y^*, \gamma x)\varphi - yy(y^*, \gamma x)(y^*, \gamma y)\varphi - xy(y^*, \gamma x)(x^*, \gamma y)\varphi - xx(x^*, \gamma x)(x^*, \gamma y)\varphi \\ &= (xy - yx) ((x^*, \gamma x)(y^*, \gamma y) - (x^*, \gamma y)(y^*, \gamma x)) \varphi \\ &= (xy - yx)(\det \gamma) \varphi \\ &= (xy - yx)\varphi \end{split}$$

where the last identity holds since  $\det \gamma \equiv 1$  as a function on  $SL(2,\mathbb{C})$ . Also, since c

is a  $\Gamma$ -invariant distribution, for all  $\varphi \in \mathcal{O}(\Gamma)^*$  we have

$$\varphi c = c\varphi \in \mathcal{O}(\Gamma)^*. \tag{A.9}$$

Moreover, if for any  $\varphi \in \mathcal{O}(\Gamma)^*$  we write  $\varphi_i = \delta_1 \otimes \cdots \otimes \varphi \otimes \cdots \otimes \delta_1 \in \mathcal{O}(\Gamma)^{*\hat{\otimes}n}$ , where  $\varphi$  is placed in the *i*th position, we have that

$$\varphi_i \psi_j \left( \delta_{s_{ij}} \Delta_{ij} \right) = \left( \delta_{s_{ij}} \Delta_{ij} \right) \varphi_i \psi_j \tag{A.10}$$

for any  $\varphi, \psi \in \mathcal{O}(\Gamma)^*$  and any  $i, j \in [1, n]$ . To see that (A.10) holds it is enough to test the right and left hand side of the equality on a decomposable function  $\tilde{f}(E_{p_1q_1}^{N_{l_1}} \otimes \cdots \otimes E_{p_nq_n}^{N_{l_n}})$ , where  $\tilde{f}$  is a function on  $S_n$ . Suppose without loss of generality that i = 1, j = 2. For the right hand side we have

$$\langle (\delta_{s_{12}}\Delta_{12}) \varphi_1 \psi_2, \tilde{f}(E_{p_1q_1}^{N_{l_1}} \otimes \dots \otimes E_{p_nq_n}^{N_{l_n}}) \rangle$$

$$= \langle \delta_{s_{12}}, \tilde{f} \rangle \left( \langle \varphi, E_{r_1q_1}^{N_{l_1}} \rangle \langle \psi, E_{r_2q_2}^{N_{l_2}} \rangle \int_{\Gamma} E_{p_1r_1}^{N_{l_1}}(\gamma) E_{p_2r_2}^{N_{l_2}}(\gamma^{-1}) d\gamma \right) \langle \delta_1, E_{p_3q_3}^{N_{l_3}} \rangle \cdots \langle \delta_1, E_{p_nq_n}^{N_{l_n}} \rangle$$

$$= \tilde{f}(s_{12}) \left( \langle \varphi, E_{r_1q_1}^{N_{l_1}} \rangle \langle \psi, E_{r_2q_2}^{N_{l_2}} \rangle \int_{\Gamma} E_{p_1r_1}^{N_{l_1}}(\gamma) E_{r_2p_2}^{N_{l_2}^*}(\gamma) d\gamma \right) E_{p_3q_3}^{N_{l_3}}(1) \cdots E_{p_nq_n}^{N_{l_n}}(1)$$

$$= \delta_{l_1l_2} \tilde{f}(s_{12}) \frac{1}{d_{l_1}} \langle \psi, E_{p_1q_2}^{N_{l_1}} \rangle \langle \varphi, E_{p_2q_1}^{N_{l_1}} \rangle E_{p_3q_3}^{N_{l_3}}(1) \cdots E_{p_nq_n}^{N_{l_n}}(1)$$

while for the left hand side we have

$$\langle \varphi_{1}\psi_{2}\left(\delta_{s_{12}}\Delta_{12}\right), \tilde{f}\left(E_{p_{1}q_{1}}^{N_{l_{1}}}\otimes\cdots\otimes E_{p_{n}q_{n}}^{N_{l_{n}}}\right)$$

$$= \langle \delta_{s_{12}}\varphi_{2}\psi_{1}\Delta_{12}, E_{p_{1}q_{1}}^{N_{l_{1}}}\otimes\cdots\otimes E_{p_{n}q_{n}}^{N_{l_{n}}}\rangle$$

$$= \langle \delta_{s_{12}}, \tilde{f}\rangle\left(\langle\psi, E_{p_{1}r_{1}}^{N_{l_{1}}}\rangle\langle\varphi, E_{p_{2}r_{2}}^{N_{l_{2}}}\rangle\int_{\Gamma}E_{r_{1}q_{1}}^{N_{l_{1}}}(\gamma)E_{r_{2}q_{2}}^{N_{l_{2}}}(\gamma^{-1})\,d\gamma\right)\langle\delta_{1}, E_{p_{3}q_{3}}^{N_{l_{3}}}\rangle\cdots\langle\delta_{1}, E_{p_{n}q_{n}}^{N_{l_{n}}}\rangle$$

$$= \tilde{f}(s_{12})\left(\langle\psi, E_{p_{1}r_{1}}^{N_{l_{1}}}\rangle\langle\varphi, E_{p_{2}r_{2}}^{N_{l_{2}}}\rangle\int_{\Gamma}E_{r_{1}q_{1}}^{N_{l_{1}}}(\gamma)E_{q_{2}r_{2}}^{N_{l_{2}}}(\gamma)\,d\gamma\right)E_{p_{3}q_{3}}^{N_{l_{3}}}(1)\cdots E_{p_{n}q_{n}}^{N_{l_{n}}}(1)$$

$$= \delta_{l_{1}l_{2}}\tilde{f}(s_{12})\frac{1}{d_{l_{1}}}\langle\psi, E_{p_{1}q_{2}}^{N_{l_{1}}}\rangle\langle\varphi, E_{p_{2}q_{1}}^{N_{l_{1}}}\rangle E_{p_{3}q_{3}}^{N_{l_{3}}}(1)\cdots E_{p_{n}q_{n}}^{N_{l_{n}}}(1)$$

where in both cases we used the fact that for any finite dimensional representation  $N_i$ , if we choose dual bases, we have  $E_{pq}^{N_i}(\gamma^{-1}) = E_{qp}^{N_i^*}(\gamma)$  for any  $p, q = 1, \ldots, d_i$ . Now using (A.8), (A.9), (A.10), if we denote by  $J_1$  the vector space spanned by relations (R1) we see that

$$\varphi^{\otimes n}\mathcal{O}(\Gamma)^{*\hat{\otimes}n}J_1\mathcal{O}(\Gamma)^{*\hat{\otimes}n}\varphi^{\otimes n} = \varphi^{\otimes n}J_1\varphi^{\otimes n}\mathcal{O}(\Gamma)^{*\hat{\otimes}n}\varphi^{\otimes n}.$$

Then for any choice of  $i_1, \ldots, i_n \in I$  and  $l \in [1, n]$  we have:

$$\varphi_{i_1} \otimes \cdots \otimes \varphi_{i_n} \cdot [x_l, y_l]$$

$$= [x_l, y_l] \cdot \varphi_{i_1} \otimes \cdots \otimes \varphi_{i_n}$$

$$= \varphi_{i_1} \otimes \cdots \otimes \frac{1}{d_{i_l}} \left( \sum_{a \in Q, h(a)=i_l} \phi_a \theta_a - \sum_{a \in Q, t(a)=i_l} \theta_a \phi_a \right) \otimes \cdots \otimes \varphi_{i_n}$$
(A.11)

and

$$\varphi_{i_1} \otimes \cdots \otimes \varphi_{i_n} c_l = \frac{\lambda_{i_l}}{d_{i_l}} \varphi_{i_1} \otimes \cdots \otimes \varphi_{i_n}.$$
 (A.12)

Indeed we can write  $\varphi_{i_1} \otimes \cdots \otimes \varphi_{i_n} c_l = \varphi_{i_1} \otimes \cdots \otimes \varphi_{i_l} c \otimes \cdots \otimes \varphi_{i_n}$ , and testing on a function  $E_{pq}^{N_j} \in \mathcal{O}(\Gamma)$  we have:

$$\begin{split} &\langle \varphi_{il}c, E_{pq}^{N_j} \rangle \\ &= \langle \varphi_{i_l}, E_{pr}^{N_j} \rangle \langle c, E_{rq}^{N_j} \rangle \\ &= \delta_{i_l j} \delta_{p1} \langle c, E_{1q}^{N_j} \rangle = \delta_{i_l j} \delta_{p1} \delta_{q1} \langle c, E_{11}^{N_{i_l}} \rangle \\ &= \frac{1}{d_{i_l}} \delta_{i_l j} \delta_{p1} \delta_{q1} \langle c, \chi_{i_l} \rangle. \end{split}$$

The last identities follow from the fact that c is a  $\Gamma$ -invariant distribution, thus a sum of duals of characters. More precisely

$$c = \sum_{i} \alpha_{i} \sum_{j=1}^{d_{i}} \check{E}_{jj}^{N_{i}} \qquad \alpha_{i} \in \mathbb{C} \,,$$

and one has

$$\langle c, E_{pq}^{N_j} \rangle = 0 \qquad \text{ if } p \neq q \,,$$

so that

$$\langle c, \chi_{i_l} \rangle = \langle \sum_i \alpha_i \sum_{j=1}^{d_i} \check{E}_{jj}^{N_i}, \sum_{j=1}^{d_{i_l}} E_{jj}^{N_{i_l}} \rangle = d_{i_l} \alpha_{i_l} = d_{i_l} \langle c, E_{11}^{N_{i_l}} \rangle.$$

Now we claim that

$$\left(\varphi_{i_1}\otimes\cdots\otimes\varphi_{i_n}\right)\delta_{s_{lj}}\Delta_{lj}\left(\varphi_{i_1}\otimes\cdots\otimes\varphi_{i_n}\right) = \frac{\delta_{i_ji_l}}{d_{i_l}}\left(\varphi_{i_1}\otimes\cdots\otimes\varphi_{i_n}\right)\delta_{s_{lj}} \qquad (A.13)$$

Indeed, supposing without loss of generality that l = 1 j = 2, and testing the left

hand side on a decomposable function  $\tilde{f}\left(E_{p_1q_1}^{N_{l_1}}\otimes\cdots\otimes E_{p_nq_n}^{N_{l_n}}\right)$  we get:

$$\langle (\varphi_{i_1} \otimes \varphi_{i_2} \otimes \cdots \otimes \varphi_{i_n}) \, \delta_{s_{12}} \Delta_{12}, \tilde{f} \left( E_{p_1 q_1}^{N_{l_1}} \otimes \cdots \otimes E_{p_n q_n}^{N_{l_n}} \right) \rangle$$

$$= \langle \delta_{s_{12}} \left( \varphi_{i_2} \otimes \varphi_{i_1} \otimes \cdots \otimes \varphi_{i_n} \right) \Delta_{12}, \tilde{f} \left( E_{p_1 q_1}^{N_{l_1}} \otimes \cdots \otimes E_{p_n q_n}^{N_{l_n}} \right) \rangle$$

$$= \tilde{f}(s_{12}) \langle \varphi_{i_2}, E_{p_1 r_1}^{N_{l_1}} \rangle \langle \varphi_{i_1}, E_{p_2 r_2}^{N_{l_2}} \rangle \left( \int_{\Gamma} E_{r_1 q_1}^{N_{l_1}} (\gamma) E_{q_2 r_2}^{N_{l_2^*}} (\gamma) \, d\gamma \right) \prod_{j \ge 3} \langle \varphi_{i_j}, E_{p_j q_j}^{N_{l_j}} \rangle$$

$$= \tilde{f}(s_{12}) \delta_{i_1 i_2} \prod_{j \ge 1} \delta_{i_j l_j} \prod_{j \ge 1} \delta_{p_j 1} \prod_{j \ge 3} \delta_{q_j 1} \int_{\Gamma} E_{1 q_1}^{N_{l_1}} (\gamma) E_{q_2 1}^{N_{l_1^*}} (\gamma) \, d\gamma$$

$$= \tilde{f}(s_{12}) \frac{\delta_{i_1 i_2}}{d_{i_1}} \prod_{j \ge 1} \delta_{i_j l_j} \prod_{j \ge 1} \delta_{p_j 1} \prod_{j \ge 1} \delta_{q_j 1}$$

$$= \frac{\delta_{i_1 i_2}}{d_{i_1}} \langle (\varphi_{i_1} \otimes \varphi_{i_2} \otimes \cdots \otimes \varphi_{i_n}) \, \delta_{s_{12}}, \tilde{f} \left( E_{p_1 q_1}^{N_{l_1}} \otimes \cdots \otimes E_{p_n q_n}^{N_{l_n}} \right) \rangle.$$

By (A.11), (A.12), (A.13) we thus have that relations of type (R1) give us exactly the relations (I) in Definition 5.5.3.

We will now find the relations that are given by (R2). We will assume without loss of generality that n = 2. First of all, for any  $u, v \in L$  and any  $\varphi, \psi \in \mathcal{O}(\Gamma)^*$ , if x, y is any basis for L, we can easily see that

$$(\varphi \otimes \psi) \cdot [u_1, v_2] = [x_1, x_2] ((x^*, hu)\varphi \otimes (x^*, gv)\psi) + [x_1, y_2] ((x^*, hu))\varphi \otimes (y^*, gv)\psi) + [y_1, x_2] ((y^*, hu))\varphi \otimes (x^*, gv)\psi) + [y_1, y_2] ((y^*, hu))\varphi \otimes (y, gv)\psi)$$
(A.14)

and similarly

$$(\varphi \otimes \psi) \,\delta_{s_{12}} \omega_L(\gamma u, v) \Delta_{12}$$
  
=  $\delta_{s_{12}} \omega_L(\gamma x, x) \Delta_{12} \left( (x^*, hu) \varphi \otimes (x^*, gv) \psi \right) + \delta_{s_{12}} \omega_L(\gamma x, y) \Delta_{12} \left( (x^*, hu) \varphi \otimes (y^*, gv) \psi \right)$   
+  $\delta_{s_{12}} \omega_L(\gamma y, x) \Delta_{12} \left( (y^*, hu) \varphi \otimes (x^*, gv) \psi \right) + \delta_{s_{12}} \omega_L(\gamma y, y) \Delta_{12} \left( (y^*, hu) \varphi \otimes (y^*, gv) \psi \right)$ .  
(A.15)

To prove this last identity we first have to introduce some more notation. For  $\varphi, \psi \in \mathcal{O}(\Gamma)^*$  we will write  $\varphi_h, \psi_g$  to indicate the variable with respect to which these distributions are considered (so  $\varphi$  is a linear functional on functions in the variable h etc...). Since the value of the distribution  $\Delta_{12}$  on any function  $f_1 \otimes f_2 \in \mathcal{O}(\Gamma)^{\otimes 2}$  can be written as

$$\langle \Delta_{12}, f_1 \otimes f_2 \rangle = \int_{\substack{\Gamma \times \Gamma \\ g = h}} f_1(g) f_2(h^{-1}) \, dg \, dh$$

We will write  $\Delta_{12,(gh)}$  to keep track of the variables. Finally for  $E_{p_1q_1}^{N_{l_1}} \otimes E_{p_2q_2}^{N_{l_2}} \in \mathcal{O}(\Gamma)^{\otimes 2}$ 

we will write

$$\Delta(E_{p_1q_1}^{N_{l_1}} \otimes E_{p_2q_2}^{N_{l_2}})(g,h,g',h') = E_{p_1r_1}^{N_{l_1}}(g) \otimes E_{p_2r_2}^{N_{l_2}}(h) \otimes E_{r_1q_1}^{N_{l_1}}(g') \otimes E_{r_2q_2}^{N_{l_2}}(h')$$

for the coproduct. Let us now consider the decomposable function  $f = \tilde{f}(E_{p_1q_1}^{N_{l_1}} \otimes E_{p_2q_2}^{N_{l_2}})$ , where  $\tilde{f}$  is a function on  $S_n$ . We have

$$\langle (\varphi \otimes \psi) \, \delta_{s_{12}} \omega_L(\gamma u, v) \Delta_{12} , f \rangle = \tilde{f}(s_{12}) \langle (\psi_g \otimes \varphi_h) \left( \omega_L(g'u, v) \Delta_{12,(g'h')} \right), \Delta(E_{p_1q_1}^{N_{l_1}} \otimes E_{p_2q_2}^{N_{l_2}})(g, h, g', h') \rangle = \tilde{f}(s_{12}) \langle \psi_g \varphi_h, E_{p_1r_1}^{N_{l_1}}(g) E_{p_2r_2}^{N_{l_2}}(h) \rangle \langle \Delta_{12,(g'h')}, \omega_L(g'u, v) E_{r_1q_1}^{N_{l_1}}(g') E_{r_2q_2}^{N_{l_2}}(h') \rangle .$$

Now making the change of variable  $(g', h') = (g^{-1}\tilde{g}h, \gamma^{-1}\tilde{h}h)$  and using the fact that the integral is left and right translation invariant we get

$$\tilde{f}(s_{12})\langle\psi_{g}\varphi_{h}, E_{p_{1}r_{1}}^{N_{l_{1}}}(g)E_{p_{2}r_{2}}^{N_{l_{2}}}(h)\rangle\langle\Delta_{12,(g'h')}, \omega_{L}(g^{-1}\tilde{g}hu,v)E_{r_{1}q_{1}}^{N_{l_{1}}}(g^{-1}\tilde{g}h)E_{r_{2}q_{2}}^{N_{l_{2}}}(h^{-1}\tilde{h}g)\rangle.$$
(A.16)

We now observe that:

$$\omega_L(g^{-1}\tilde{g}hu, v) = \omega_L(\tilde{g}hu, gv)$$

$$= \overbrace{(x^*, hu)(x^*, gv)\omega_L(\tilde{g}x, x)}^{1} + \overbrace{(x^*, hu)(y^*, gv)\omega_L(\tilde{g}x, y)}^{2}$$

$$+ \overbrace{(y^*, hu)(x^*, gv)\omega_L(\tilde{g}y, x)}^{3} + \overbrace{(y^*, hu)(y^*, gv)\omega_L(\tilde{g}y, y)}^{4}$$
(A.17)

and that

$$E_{r_1q_1}^{N_{l_1}}(g^{-1}\tilde{g}h)E_{r_2q_2}^{N_{l_2}}(h^{-1}\tilde{h}^{-1}g)$$
(A.18)

$$= E_{r_1s_1}^{N_{l_1}}(g^{-1})E_{s_1t_1}^{N_{l_1}}(\tilde{g})E_{t_1q_1}^{N_{l_1}}(h)E_{r_2s_2}^{N_{l_2}}(h^{-1})E_{s_2t_2}^{N_{l_2}}(\tilde{h}^{-1})E_{t_2q_2}^{N_{l_2}}(g).$$
(A.19)

Using (A.17) we can rewrite (A.16) as a sum of four terms. If we use (A.19) to rewrite the first of these terms, for example, we get

$$\begin{split} \tilde{f}(s_{12}) \langle \psi_{g} \varphi_{h}, (x^{*}, hu)(x^{*}, gv) E_{p_{1}r_{1}}^{N_{l_{1}}}(g) E_{r_{1}s_{1}}^{N_{l_{1}}}(g^{-1}) E_{t_{1}q_{1}}^{N_{l_{1}}}(h) E_{p_{2},r_{2}}(h) E_{r_{2}s_{2}}^{N_{l_{2}}}(h^{-1}) E_{t_{2}q_{2}}^{N_{l_{2}}}(g) \rangle \\ & \cdot \langle \omega_{L}(\tilde{g}x, x) \Delta_{12,(\tilde{g}\tilde{h})}, E_{s_{1}t_{1}}^{N_{l_{1}}}(\tilde{g}) E_{s_{2}t_{2}}^{N_{l_{2}}}(\tilde{h}) \rangle \\ &= \tilde{f}(s_{12}) \langle \psi_{g} \varphi_{h}, (x^{*}, hu)(x^{*}, gv) E_{p_{1}s_{1}}^{N_{l_{1}}}(e) E_{t_{1}q_{1}}^{N_{l_{1}}}(h) E_{p_{2},s_{2}}(e) E_{t_{2}q_{2}}^{N_{l_{2}}}(g) \rangle \cdot \\ & \cdot \langle \omega_{L}(\tilde{g}x, x) \Delta_{12,(\tilde{g}\tilde{h})}, E_{s_{1}t_{1}}^{N_{l_{1}}}(\tilde{g}) E_{s_{2}t_{2}}^{N_{l_{2}}}(\tilde{h}) \rangle \\ &= \tilde{f}(s_{12}) \langle \psi_{g} \varphi_{h}, (x^{*}, hu)(x^{*}, gv) E_{t_{1}q_{1}}^{N_{l_{1}}}(h) E_{t_{2}q_{2}}^{N_{l_{2}}}(g) \rangle \cdot \\ & \cdot \langle \omega_{L}(\tilde{g}x, x) \Delta_{12,(\tilde{g}\tilde{h})}, E_{p_{1}s_{1}}^{N_{l_{1}}}(e) E_{s_{1}t_{1}}^{N_{l_{1}}}(\tilde{g}) E_{p_{2},s_{2}}(e) E_{s_{2}t_{2}}^{N_{l_{2}}}(\tilde{h}) \rangle \end{split}$$

$$\begin{split} &= \tilde{f}(s_{12}) \langle \psi_g \varphi_h \,, \, (x^*, hu)(x^*, gv) E_{t_1 q_1}^{N_{l_1}}(h) E_{t_2 q_2}^{N_{l_2}}(g) \rangle \cdot \langle \omega_L(\tilde{g}x, x) \Delta_{12, (\tilde{g}\,\tilde{h})}, E_{p_1 t_1}^{N_{l_1}}(\tilde{g}) E_{p_2, t_2}(\tilde{h}) \rangle \\ &= \tilde{f}(s_{12}) \langle \left( \omega_L(\tilde{g}u, v) \Delta_{12, (\tilde{g}\,\tilde{h})} \right) ((x^*, hu) \varphi_h) \left( (x^*, gv) \psi_g \right) \,, \, \Delta(E_{p_1 q_1}^{N_{l_1}} \otimes E_{p_2 q_2}^{N_{l_2}})(\tilde{g}, \tilde{h}, h, g) \rangle \\ &= \langle \delta_{s_{12}} \left( (x^*, hu) \varphi \otimes (x^*, gu) \psi \right) \omega_L(\gamma x, v) \Delta_{12} \,, \, f \rangle \end{split}$$

where we just used the properties of the coproduct and counit (evaluation at the identity). It is of course possible to rewrite the remaining three terms in a similar way, so that we get exactly expression (A.15).

Now for any  $i, j, k, l \in I$  an easy computation shows that, via the identification  $TV \sharp \mathcal{O}(\Gamma)^{*\hat{\otimes}^2} \cong T_{\mathcal{O}(\Gamma)^{*\hat{\otimes}^2}}(V \otimes \mathcal{O}(\Gamma)^{*\hat{\otimes}^2})$ , we have

$$(\varphi_{i} \otimes \varphi_{j}) (\varphi \otimes \delta_{1}) [u_{1}, v_{2}] (\delta_{1} \otimes \psi) (\varphi_{k} \otimes \varphi_{l})$$

$$= (\varphi_{i} \varphi \otimes \varphi_{j}) (u_{1} \otimes (\varphi_{k} \otimes \varphi_{j})) \bigotimes (\varphi_{k} \otimes \varphi_{j}) (v_{2} \otimes (\varphi_{k} \otimes \psi \varphi_{l}))$$

$$- (\varphi_{i} \otimes \varphi_{j}) (v_{2} \otimes (\varphi_{i} \otimes \psi \varphi_{l})) \bigotimes (\varphi_{i} \varphi \otimes \varphi_{l}) (u_{1} \otimes (\varphi_{k} \otimes \varphi_{l}))$$
(A.20)

where  $\bigotimes$  denotes the product in  $T_{\mathcal{O}(\Gamma)^*\hat{\otimes}^2}(V \otimes \mathcal{O}(\Gamma)^{*\hat{\otimes}^2})$ , and we can see (A.20) as an identity between algebraic distributions on  $\Gamma^2$  with values in  $T^2V$ . On the other hand we trivially have that

$$(\varphi_i \otimes \varphi_j) (\varphi \otimes \delta_1) (\delta_{s_{12}} \omega_L(\gamma u, v) \Delta_{12}) (\delta_1 \otimes \psi) (\varphi_k \otimes \varphi_l) = \delta_{s_{12}} (\varphi_j \otimes \varphi_i \varphi) \omega_L(\gamma u, v) \Delta_{12} (\varphi_k \otimes \psi \varphi_l) .$$
(A.21)

As in [GG05] we observe now that for any arrow  $a \in \overline{Q}$  we can find distributions  $\varphi_a, \psi_a \in \mathcal{O}(\Gamma)^{*\hat{\otimes}n}$  and vectors  $u_a, v_a \in L$  such that

$$\varphi_{t(a)}\varphi_a\left(u_a\otimes\varphi_{h(a)}\right)\neq 0$$
 and  $\varphi_{h(a)}\left(v_a\otimes\psi_a\varphi_{t(a)}\right)\neq 0.$ 

Also in our case Q has no loop vertices, thus we have that the spaces  $\varphi_i(L \otimes \mathcal{O}(\Gamma)^{*\hat{\otimes}^2})\varphi_j$ are at most one dimensional and for any  $i, j \in I$  we have an identification:

$$\left(\varphi_i \mathcal{O}(\Gamma)^{*\hat{\otimes}2} \otimes L \otimes \mathcal{O}(\Gamma)^{*\hat{\otimes}2} \varphi_j\right)^{\Gamma} \to \varphi_i \left(L \otimes \mathcal{O}(\Gamma)^{*\hat{\otimes}2}\right) \varphi_j$$
$$\alpha \otimes u \otimes \beta \to \alpha \left(u \otimes \beta\right)$$

where  $\varphi_i \mathcal{O}(\Gamma)^{*\hat{\otimes}^2} \otimes L \otimes \varphi_j \mathcal{O}(\Gamma)^{*\hat{\otimes}^2} \cong N_i^* \otimes L \otimes N_j$  as  $\Gamma$ -modules. Moreover again as in [GG05] we have a non degenerate  $\Gamma$ -equivariant pairing

$$\left(\varphi_i \mathcal{O}(\Gamma)^{*\hat{\otimes}^2} \otimes L \otimes \mathcal{O}(\Gamma)^{*\hat{\otimes}^2} \varphi_j\right) \bigotimes \left(\varphi_j \mathcal{O}(\Gamma)^{*\hat{\otimes}^2} \otimes L \otimes \mathcal{O}(\Gamma)^{*\hat{\otimes}^2} \varphi_i\right) \to \mathbb{C}$$
$$(\alpha \otimes u \otimes \beta) \bigotimes (\alpha' \otimes u \otimes \beta') \to (\alpha\beta')(\alpha'\beta)\omega_L(u,u').$$

As a consequence, we can assume that, for any  $a \in \overline{Q}$ , we have  $\omega_L(u_a, v_a) = 1$ .

Moreover  $\varphi_{t(a)}\varphi(v_a\otimes\varphi_{h(a)})=0$  if  $\varphi_{h(a)}(v_a\otimes\psi\varphi_{t(a)})\neq 0$ , and  $\varphi_{h(a)}(u_a\otimes\psi\varphi_{t(a)})=0$  if  $\varphi_{t(a)}\varphi(u_a\otimes\varphi_{h(a)})\neq 0$ .

Note that if  $i \neq l$  or  $j \neq k$  the expression (A) is zero. To see this, let us evaluate the distribution  $(\varphi_j \otimes \varphi_i \varphi) \omega_L(\gamma u, v) \Delta_{12} (\varphi_k \otimes \psi \varphi_l)$  on a function  $E_{p_1q_1}^{N_{l_1}} \otimes E_{p_2q_2}^{N_{l_2}}$ . We have

$$\langle (\varphi_j \otimes \varphi_i \varphi) \, \omega_L(\gamma u, v) \Delta_{12} \left( \varphi_k \otimes \psi \varphi_l \right) , \, E_{p_1 q_1}^{N_{l_1}} \otimes E_{p_2 q_2}^{N_{l_2}} \rangle$$

$$= \langle \varphi_j , \, E_{p_1 r_1}^{N_{l_1}} \rangle \langle \varphi_i \varphi , \, E_{p_2 r_2}^{N_{l_2}} \rangle \langle \omega_L(\gamma u, v) \Delta_{12} , \, E_{r_1 s_1}^{N_{l_1}} \otimes E_{r_2 s_2}^{N_{l_2}} \rangle \langle \varphi_k , \, E_{s_1 q_1}^{N_{l_1}} \rangle \langle \psi \varphi_l , \, E_{s_2 q_2}^{N_{l_2}} \rangle .$$

Since last expression is zero if  $j, k \neq l_1$  and  $i, l \neq l_2$  the above distribution is identically 0 if  $j \neq k$  or  $i \neq l$ .

Thus, if  $a, b \in \overline{Q}$  are two arrows such that  $b \neq a^*$  or  $a \neq b^*$  we get from (A.20), (A) and (R2) that

$$(a \otimes h(b))(t(a) \otimes b) - (h(a) \otimes b)(a \otimes t(b)) = 0.$$

Suppose now j = k and i = l. Consider an edge  $a : i \to j$  in  $\overline{Q}$  and suppose, for simplicity,  $a \in Q$ . We have an injection as an irreducible factor  $\theta_a : N_i \hookrightarrow L \otimes N_j$ . We can choose a basis  $\xi := \{\xi_i\}$  of  $L \otimes N_j = N_i \oplus \ldots$  adapted to this decompositions into irreducibles

$$\xi_1 := \varphi_i = E_{11}^{N_i}, \quad \xi_1 := E_{21}^{N_i}, \quad \xi_3 := E_{31}^{N_i}, \quad \dots, \quad \xi_{d_i} := E_{d_i 1}^{N_i}, \quad \dots$$

On the other hand we can choose a basis  $\mu := \{\mu_i\}$  for  $L \otimes N_j$  adapted to the tensor product

$$\mu_1 := u_a \otimes \varphi_j = u_a \otimes E_{11}^{N_j}, \quad \mu_2 = u_a \otimes E_{21}^{N_j}, \quad \dots, \quad \mu_{2d_j} := v_a \otimes E_{d_j 1}^{N_j}.$$

Let's now define the matrix  $\tau = (\tau_{pq})$  by  $\varphi \mu_q = \sum_p \tau_{pq} \xi_p$  and the matrix  $\rho = (\rho_{pq})$  by  $\psi \xi_q = \sum_p \rho_{pq} \mu_q$ . In other words we have  $\tau = \xi \varphi_\mu$ , where  $\xi \varphi_\mu = (\xi_p \varphi_{\mu_q})$  denotes the matrix representing the linear map induced by  $\varphi$  on  $L \otimes N_j$  if we choose the basis  $\mu$  for the domain and  $\xi$  for the image. Similarly we have  $\rho = \mu \psi_{\xi}$ . Now, recalling that we are using the following identifications

$$N_j \stackrel{\phi_a}{\hookrightarrow} L \otimes N_j \stackrel{\vartheta_a}{\hookrightarrow} L \otimes L \otimes N_j \stackrel{\omega_L \otimes 1}{\to} N_j ,$$

and that by Lemma A.0.5 this composition of morphisms equals  $d_j Id_{N_j}$ , we have that

$$\varphi_i \varphi \left( u_a \otimes \varphi_j \right) = \tau_{11} \varphi_i \quad \text{and} \quad \varphi_j \left( v_a \otimes \psi \varphi_i \right) = -\frac{\rho_{11}}{d_i} \varphi_j.$$
 (A.22)

We now claim that

$$(\varphi_j \otimes \varphi_i \varphi) \left( \omega_L(\gamma u_a, v_a) \Delta_{12} \right) \left( \varphi_j \otimes \psi \varphi_i \right) = \frac{\tau_{11} \rho_{11}}{d_i} \varphi_j \otimes \varphi_i \,. \tag{A.23}$$

First of all it's easy to see that

$$(\varphi_j \otimes \varphi_i \varphi) (\omega_L(\gamma u_a, v_a) \Delta_{12}) (\varphi_j \otimes \psi \varphi_i) = C \varphi_j \otimes \varphi_i$$

where C is some constant. To compute C we will evaluate the left hand side of (A.23) on the function  $E_{11}^{N_j} \otimes E_{11}^{N_i} \in \mathcal{O}(\Gamma)^{\otimes 2}$ . We recall that we can see the functions  $E_{pq}^{N_i}(\gamma)$ as the matrix coefficients for the action of  $\gamma$  on the direct factor  $N_i \subset L \otimes N_j$  in the basis  $\xi$  and the functions  $E_{rs}^L(\gamma)E_{pq}^{N_j}(\gamma)$  as the matrix coefficients for  $\gamma$  on  $L \otimes N_j$  in the basis  $\mu$ . We define the matrix  $\alpha = \{\alpha_{pq}\}$  as the matrix of the change of basis  $\mu_q = \sum_p \alpha_{pq}\xi_p$  and by  $\tilde{\alpha} = (\tilde{\alpha}_{pq})$  its inverse. Accordingly to the previous notation we write  $_{\xi}\varphi_{\xi} = (_{\xi_q}\varphi_{\xi_p})$  (respectively  $_{\xi}\psi_{\xi} = (_{\xi_q}\psi_{\xi_p})$ ) for the matrix of the linear map  $\varphi$ (respectively  $\psi$ ) where we chose the basis  $\xi$  both for the domain and the image.

$$\begin{split} &\langle (\varphi_{j} \otimes \varphi_{i} \varphi) \left( \omega_{L}(\gamma u_{a}, v_{a}) \Delta_{12} \right) \left( \varphi_{j} \otimes \psi \varphi_{i} \right) , E_{11}^{N_{j}} \otimes E_{11}^{N_{i}} \rangle \\ &= \sum_{r,p=1}^{d_{j}} \sum_{r',p'=1}^{d_{i}} \langle \tilde{E}_{11}^{N_{j}}, E_{1r}^{N_{j}} \rangle \langle \varphi, E_{1r'}^{N_{i}} \rangle \left( \int_{\Gamma} \omega_{L}(\gamma u_{a}, v_{a}) E_{rp}^{N_{i}}(\gamma) E_{r'p'}^{N_{i}}(\gamma^{-1}) \, d\gamma \right) \cdot \\ &\cdot \langle \check{E}_{11}^{N_{j}}, E_{p1}^{N_{j}} \rangle \langle \psi, E_{p'1}^{N_{i}} \rangle \\ &= \sum_{r',p'=1}^{d_{i}} \langle \varphi, E_{1r'}^{N_{i}} \rangle \left( \int_{\Gamma} \omega_{L}(\gamma u_{a}, v_{a}) E_{11}^{N_{j}}(\gamma) E_{p'r'}^{N_{i}}(\gamma^{-1}) \, d\gamma \right) \langle \psi, E_{p'1}^{N_{i}} \rangle \\ &= \sum_{r',p'=1}^{d_{i}} \langle \varphi, E_{1r'}^{N_{i}} \rangle \left( \int_{\Gamma} E_{11}^{L}(\gamma) E_{11}^{N_{j}}(\gamma) E_{p'r'}^{N_{i}}(\gamma^{-1}) \, d\gamma \right) \langle \psi, E_{p'1}^{N_{i}} \rangle \\ &= \frac{1}{d_{i}} \sum_{r',p'=1}^{d_{i}} \langle \varphi, E_{1r'}^{N_{i}} \rangle \langle \psi, E_{p'1}^{N_{i}} \rangle \langle \check{E}_{p'r'}^{N_{i}}, E_{11}^{L}(\gamma) E_{11}^{N_{j}}(\gamma) \rangle \\ &= \frac{1}{d_{i}} \sum_{r',p'=1}^{d_{i}} \langle \varphi, E_{1r'}^{N_{i}} \rangle \langle \psi, E_{p'1}^{N_{i}} \rangle \langle \check{E}_{p'r'}^{N_{i}}, \sum_{s,t=1}^{d_{i}} \tilde{\alpha}_{1s} \alpha_{t1} E_{st}^{N_{i}} \rangle \\ &= \frac{1}{d_{i}} \sum_{r',p'=1}^{d_{i}} \langle \varphi, E_{1r'}^{N_{i}} \rangle \langle \psi, E_{p'1}^{N_{i}} \rangle \langle \check{E}_{p'r'}^{N_{i}}, \sum_{s,t=1}^{d_{i}} \tilde{\alpha}_{1s} \alpha_{t1} E_{st}^{N_{i}} \rangle \\ &= \frac{1}{d_{i}} \left( \sum_{p'=1}^{d_{i}} \langle \varphi, E_{1r'}^{N_{i}} \rangle \langle \psi, E_{p'1}^{N_{i}} \rangle \langle \check{\alpha}_{1p'} \alpha_{r'1} \right) \\ &= \frac{1}{d_{i}} \left( \sum_{p'=1}^{d_{i}} \xi_{p'} \psi_{\xi_{1}} \tilde{\alpha}_{1p'} \right) \left( \sum_{r'=1}^{d_{i}} \xi_{i} \varphi_{\xi_{r'}} \alpha_{r'1} \right) \\ &= \frac{1}{d_{i}} \left( \sum_{p'=1}^{d_{i}} \xi_{p'} \psi_{\xi_{1}} \tilde{\alpha}_{1p'} \right) \left( \sum_{r'=1}^{d_{i}} \xi_{i} \varphi_{\xi_{r'}} \alpha_{r'1} \right) \\ &= \frac{1}{d_{i}} \left( \sum_{p'=1}^{d_{i}} \xi_{p'} \psi_{\xi_{1}} \tilde{\alpha}_{1p'} \right) \left( \sum_{r'=1}^{d_{i}} \xi_{i} \varphi_{\xi_{r'}} \alpha_{r'1} \right) \\ &= \frac{1}{d_{i}} \left( \sum_{p'=1}^{d_{i}} \xi_{p'} \psi_{\xi_{1}} \tilde{\alpha}_{1p'} \right) \left( \sum_{r'=1}^{d_{i}} \xi_{i} \varphi_{\xi_{r'}} \alpha_{r'1} \right) \\ &= \frac{1}{d_{i}} \left( \sum_{p'=1}^{d_{i}} \xi_{p'} \psi_{\xi_{1}} \tilde{\alpha}_{1p'} \right) \left( \sum_{r'=1}^{d_{i}} \xi_{i} \varphi_{\xi_{r'}} \alpha_{r'1} \right) \\ &= \frac{1}{d_{i}} \left( \sum_{p'=1}^{d_{i}} \xi_{p'} \psi_{\xi_{1}} \tilde{\alpha}_{1p'} \right) \left( \sum_{p'=1}^{d_{i}} \xi_{p'} \xi_{p'} \langle \varphi_{r'1} \rangle \langle \varphi_{p'} \xi_{p'} \langle \varphi_{p'} \xi_{p'} \langle \varphi_{p'} \langle \varphi_{p'} \xi_{p'} \langle \varphi_{p'} \xi_{p'} \langle \varphi_{p'} \langle \varphi_{p'} \xi_{p'} \langle \varphi_{p'} \langle \varphi_{p'} \langle \varphi$$

where the last identity holds since  $\rho = {}_{\mu}\psi_{\xi} = \alpha^{-1}{}_{\xi}\psi_{\xi} = \tilde{\alpha}{}_{\xi}\psi_{\xi}$  and  $\tau = {}_{\xi}\varphi_{\mu} = {}_{\xi}\psi_{\xi}\alpha$ .

Thus we have that  $C = \frac{\rho_{11}a\tau_{11}}{d_i}$  and the identity (A.23) holds. So now taking i = l,  $j = k \ u = u_a, v = v_a, \varphi = \varphi_a, \psi = \psi_a$  in (A.20) and (A.23), and using (A.22), we have

that relation (R2) gives us exactly

$$(a^* \otimes h(a)) (h(a) \otimes a) - (t(a) \otimes a) (a^* \otimes t(a)) = 2k\delta_{s_{12}} (h(a) \otimes t(a))$$

since  $\tau_{11}, \rho_{11} \neq 0$  in this case as observed above. Also taking  $u = u_a, v = v_a$  in (A.20) and (A.23) we have that, if  $\varphi_i \varphi(u_a \otimes \varphi_j) \neq 0$ , then  $\varphi_j(u_a \otimes \psi \varphi_j) = 0$  and so  $\rho_{d_j+1} = 0$  (see (A.22)) and both sides of (R2) give zero. The same is true if we exchange the roles of  $u_a$  and  $v_a$ . Thus the relations (R2) give exactly the relations (II) of Definition 5.5.3.

## Bibliography

- [AFLS00] J. Alev, M. A. Farinati, T. Lambre, and A.L. Solotar, Homologie des invariants d'une algebre de Weyl sous l'action d'un groupe fini, J. Algebra 232 (2000), no. 2, 564–577.
- [BEG03] Yuri Berest, Pavel Etingof, and Victor Ginzburg, *Finite-dimensional rep*resentations of rational Cherednik algebras, Int. Math. Res. Not. (2003), no. 19, 1053–1088.
- [CBH98] William Crawley-Boevey and Martin P. Holland, Noncommutative deformations of Kleinian singularities, Duke Math. J. 92 (1998), no. 3, 605–635.
- [CE03] Tatyana Chmutova and Pavel Etingof, On some representations of the rational Cherednik algebra, Represent. Theory 7 (2003), 641–650 (electronic).
- [Che95] Ivan Cherednik, Double affine Hecke algebras and Macdonald's conjectures, Ann. of Math. (2) **141** (1995), no. 1, 191–216.
- [Chm06] Tatyana Chmutova, Representations of the rational Cherednik algebras of dihedral type, J. Algebra **297** (2006), no. 2, 542–565.
- [Cox91] H. S. M. Coxeter, *Regular complex polytopes*, second ed., Cambridge University Press, Cambridge, 1991.
- [DG80] Michel Demazure and Peter Gabriel, Introduction to algebraic geometry and algebraic groups, North-Holland Mathematics Studies, vol. 39, North-Holland Publishing Co., Amsterdam, 1980, Translated from the French by J. Bell.
- [Dri86] V. G. Drinfel'd, *Degenerate affine hecke algebras and yangians*, Funktsional. Anal. i Prilozhen. **20** (1986), no. 1, 69–70.
- [EG02] Pavel Etingof and Victor Ginzburg, Symplectic reflection algebras, Calogero-Moser space, and deformed Harish-Chandra homomorphism, Invent. Math. 147 (2002), no. 2, 243–348.
- [EGG05] Pavel Etingof, Wee Liang Gan, and Victor Ginzburg, *Continuous Hecke* algebras, Transform. Groups **10** (2005), no. 3-4, 423–447.

- [Eis95] David Eisenbud, Commutative algebra, Graduate Texts in Mathematics, vol. 150, Springer-Verlag, New York, 1995, With a view toward algebraic geometry.
- [EM05] Pavel Etingof and Silvia Montarani, Finite dimensional representations of symplectic reflection algebras associated to wreath products, Represent. Theory 9 (2005), 457–467 (electronic).
- [EO06] Pavel Etingof and Alexei Oblomkov, Quantization, orbifold cohomology, and Cherednik algebras, Jack, Hall-Littlewood and Macdonald polynomials, Contemp. Math., vol. 417, Amer. Math. Soc., Providence, RI, 2006, pp. 171–182.
- [Gan06] Wee Liang Gan, Reflection functors and symplectic reflection algebras for wreath products, Adv. Math. **205** (2006), no. 2, 599–630.
- [Ger63] Murray Gerstenhaber, *The cohomology structure of an associative ring*, Ann. of Math. (2) **78** (1963), 267–288.
- [Ger64]  $\_$ , On the deformation of rings and algebras, Ann. of Math. (2) **79** (1964), 59–103.
- [GG05] Wee Liang Gan and Victor Ginzburg, Deformed preprojective algebras and symplectic reflection algebras for wreath products, J. Algebra 283 (2005), no. 1, 350–363.
- [GGOR03] Victor Ginzburg, Nicolas Guay, Eric Opdam, and Raphael Rouquier, On the category O for rational Cherednik algebras, Invent. Math. 154 (2003), no. 3, 617–651.
- [GS04] Iain Gordon and S. Paul Smith, Representations of symplectic reflection algebras and resolutions of deformations of symplectic quotient singularities, Math. Ann. **330** (2004), no. 1, 185–200.
- [GS05] I. Gordon and J. T. Stafford, *Rational Cherednik algebras and Hilbert* schemes, Adv. Math. **198** (2005), no. 1, 222–274.
- [GS06] \_\_\_\_\_, Rational Cherednik algebras and Hilbert schemes. II. Representations and sheaves, Duke Math. J. **132** (2006), no. 1, 73–135.
- [JK81] Gordon James and Adalbert Kerber, The representation theory of the symmetric group, Encyclopedia of Mathematics and its Applications, vol. 16, Addison-Wesley Publishing Co., Reading, Mass., 1981, With a foreword by P. M. Cohn, With an introduction by Gilbert de B. Robinson.
- [Kac90] Victor G. Kac, *Infinite-dimensional Lie algebras*, third ed., Cambridge University Press, Cambridge, 1990.

- [Kha05] Apoorva Khare, Category O over a deformation of the symplectic oscillator algebra, J. Pure Appl. Algebra **195** (2005), no. 2, 131–166.
- [KR] Masaki Kashiwara and Raphael Rouquier, *Microlocalization of rational Cherednik algebras*, to appear in Duke Math. Journal, math.RT/0705.1245v2 (2007).
- [Mac80] I. G. Macdonald, *Polynomial functors and wreath products*, J. Pure Appl. Algebra **18** (1980), no. 2, 173–204.
- [McK81] John McKay, Cartan matrices, finite groups of quaternions, and Kleinian singularities, Proc. Amer. Math. Soc. 81 (1981), no. 1, 153–154.
- [Mon07a] Silvia Montarani, Finite dimensional representations of symplectic reflection algebras associated to wreath products. II, Algebr. Represent. Theory 10 (2007), no. 5, 471–479.
- [Mon07b] \_\_\_\_\_, On some finite dimensional representations of symplectic reflection algebras associated to wreath products, Comm. Algebra **35** (2007), no. 5, 1449–1467.
- [vdB98] Michel van den Bergh, A relation between Hochschild homology and cohomology for Gorenstein rings, Proc. Amer. Math. Soc. 126 (1998), no. 5, 1345–1348.
- [vdB02] \_\_\_\_\_, Erratum to: "A relation between Hochschild homology and cohomology for Gorenstein rings" [Proc. Amer. Math. Soc. 126 (1998), no. 5, 1345–1348; MR1443171 (99m:16013)], Proc. Amer. Math. Soc. 130 (2002), no. 9, 2809–2810 (electronic).